# DISCRETE-TIME SLIDING MODE CONTROL FOR NETWORKED CONTROL SYSTEM

A Thesis submitted to Gujarat Technological University

for the Award of

## **Doctor of Philosophy**

in

## **Instrumentation and Control Engineering**

By

## **DIPESH H SHAH**

### 129990917002

under the supervision of

## Dr. Axaykumar J Mehta



GUJARAT TECHNOLOGICAL UNIVERSITY, AHMEDABAD

February-2018

## © DIPESH HARISH SHAH

### DECLARATION

I declare that the thesis entitled "DISCRETE-TIME SLIDING MODE CONTROL FOR NETWORKED CONTROL SYSTEM" submitted by me for the degree of Doctor of Philosophy is the record of research work carried out by me during the period from DECEMBER-2012 to MARCH-2017 under the supervision of "DR. AXAYKUMAR J MEHTA" and this has not formed the basis for the award of any degree, diploma, associateship, fellowship, titles in this or any other University or other institution of higher learning.

I further declare that the material obtained from other sources has been duly acknowledged in the thesis. I shall be solely responsible for any plagiarism or other irregularities, if noticed in the thesis.

Signature of Research Scholar: ..... Date: .....

Name of Research Scholar: **DIPESH HARISH SHAH** 

Place: GTU, AHMEDABAD

#### **CERTIFICATE**

I certify that the work incorporated in the thesis "DISCRETE-TIME SLIDING MODE CONTROL FOR NETWORKED CONTROL SYSTEM" submitted by Shri. DIPESH HARISH SHAH was carried out by the candidate under my supervision/guidance. To the best of my knowledge: (i) the candidate has not submitted the same research work to any other institution for any degree/diploma, Associateship, Fellowship or other similar titles (ii) the thesis submitted is a record of original research work done by the Research Scholar during the period of study under my supervision, and (iii) the thesis represents independent research work on the part of the Research Scholar.

Signature of Supervisor: ..... Da

Date: .....

Name of Supervisor: DR. AXAYKUMAR J MEHTA

Place: IIT-RAM, AHMEDABAD

## **ORIGINALITY REPORT CERTIFICATE**

It is certified that PhD Thesis titled **"DISCRETE-TIME SLIDING MODE CONTROL FOR NETWORKED CONTROL SYSTEM"** by **"DIPESH HARISH SHAH"** has been examined by us. We undertake the following:

- a. Thesis has significant new work/knowledge as compared already published or are under consideration to be published elsewhere. No sentence, equation, diagram, table, paragraph or section has been copied verbatim from previous work unless it is placed under quotation marks and duly referenced.
- b. The work presented is original and own work of the author (i.e. there is no plagiarism). No ideas, processes, results or words of others have been presented as Author own book.
- c. There is no fabrication of data or results which have been complied/analyzed.
- d. There is no falsification by manipulating research materials, equipment or processes, or changing or omitting data or results such that the research is not accurately represented in the research record.
- e. The thesis has been checked using <u>"Turnitin Plagiarism Checker"</u> (copy of originality report attached) and found within limits as per GTU Plagiarism Policy and instructions issued from time to time (i.e. permitted similarity index <=25%).

Date:
Date:

Place: IIT-RAM, AHMEDABAD

PhD	Thesis			
ORIGIN	ALITY REPORT			
%	RIT Y INDEX	% <b>1</b> INTERNET SOURCES	% <b>5</b> PUBLICATIONS	% <b>1</b> STUDENT PAPERS
PRIMAR	Y SOURCES			
1	Control Commu Publication	and Estimation I nication Networl	Methods over <s, 2014.<="" td=""><td>%</td></s,>	%
2	Studies 2015. Publication	in Systems Deci	ision and Con	trol, %1
3	WWW.SPI	ringer.com		% <b>1</b>

EXCLUDE QUOTES	ON	EXCLUDE MATCHES < 1%
EXCLUDE BIBLIOGRAPHY	ON	

### PhD THESIS Non-Exclusive License to GUJARAT TECHNOLOGICAL UNIVERSITY

In consideration of being PhD Research Scholar at GTU and in the interests of the facilitation of research at GTU and elsewhere I, "**DIPESH HARISH SHAH**" having (Enrolment No.) "**129990917002**" hereby grant a non-exclusive, royalty free and perpetual license to GTU on the following terms:

- a) GTU is permitted to archive, reproduce and distribute my thesis, in whole or in a part, and/or my abstract, in whole or in part (referred to collectively as the "Work") anywhere in the world, for non-commercial purposes, in all forms of media;
- b) GTU is permitted to authorize, sub-lease, sub-contract or procure any of the acts mentioned in the paragraph (a);
- c) GTU is authorized to submit the Work at any National/International Library, under the authority of their "Thesis Non-Exclusive License";
- d) The Universal Copyright Notice (<sup>©</sup>) shall appear on all copies made under the authority of this license;
- e) I undertake to submit my thesis, through my University, to any Library and Archives.Any abstract submitted with the thesis will be considered to form part of the thesis.
- f) I represent that my thesis is my original work, does not infringe any rights of others, including privacy rights, and that I have the right to make the grant conferred by this non-exclusive license.
- g) If third part copyrighted material was included in my thesis for which, under the terms of the Copyright Act, written permission from the copyright owners is required, I

have obtained such permission from the copyright owners to do the acts mentioned in paragraph (a) above for the full term of copyright protection.

- h) I retain copyright ownership and moral rights in my thesis, and may deal with the copyright in my thesis, in any way consistent with rights granted by me to my University in this non-exclusive license.
- I further promise to inform any person to whom I mat hereafter assign or license my copyright in my thesis of the rights granted by me to my University in this nonexclusive license.
- j) I am aware of and agree to accept the conditions and regulations of PhD including all policy matters related to authorship and plagiarism.

Signature of the Research Scholar: .....

Name of Research Scholar: DIPESH H SHAH

Date: .....

Place: GTU, AHMEDABAD

Signature of Supervisor: .....

Name of Supervisor: **DR. AXAYKUMAR J MEHTA** 

Date: .....

Place: IIT-RAM, AHMEDABAD

Seal:

### THESIS APPROVAL FORM

The viva-voce of the PhD Thesis submitted by Shri. DIPESH HARISH SHAH (Enrollment No. 129990917002) entitled "DISCRETE-TIME SLIDING MODE CONTROL FOR NETWORKED CONTROL SYSTEM" was conducted on .....

(day and date) at Gujarat Technological University.

#### (Please tick any one of the following option)

- □ The performance of the candidate was satisfactory. We recommend that he/she be awarded the PhD degree.
- □ Any further modifications in research work recommended by the panel after 3 months from the date of first viva-voce upon request of the Supervisor or request of Independent Research Scholar after which viva-voce can be re-conducted by the same panel again.

(briefly specify the modifications suggested by panel)

□ The performance of the candidate was unsatisfactory. We recommend that he/she should not be awarded the PhD degree.

(The panel must give justifications for rejecting the research work)

Name and Signature of Supervisor with Seal

1)(External Examiner 1) Name and Sign

2)(External Examiner 2) Name and Sign

3)(External Examiner 3) Name and Sign

### ABSTRACT

Networked control systems (NCSs) are traditional feedback control loops closed through a real time communication network. That is, in networked control systems, communication networks are employed to exchange information (reference input, plant output, control input, etc...) between control system components (sensors, controllers, actuators, etc...). NCS has become popular in the field of control due to its distinct advantages such as low cost, reduced weight, simple installation and maintenance, resource sharing and high reliability. As a result, NCS have great potential in industrial applications such as manufacturing plants, smart grid, haptic collaboration, vehicles, aircrafts, robotics, spacecrafts and many others. NCS generally possess a dynamic nature which results in various challenges for researchers in terms of random time delay, packet loss, multiple packet loss, packet disordering, resource allocation and bandwidth sharing. It is well known that the performance of NCS is significantly deteriorated due to these communication uncertainties. If these challenges are not handled properly, they may result in degradation of the system's performance. Among all these issues, time delay and packet loss are considered to be crucial issues in NCS that degrades the stability and control performance of closed-loop control systems.

The delays may be constant, time-varying, and in most cases, random. The nature of network delay mainly depends on the configuration of the communication medium. If the communication medium is configured using lease line concept then the delays are always deterministic in nature. And whenever the communication medium is shared by large number of devices then the delays are random in nature. It is worth to mention here that, the amount of time required for the data packets to travel from sensor to controller and controller to actuator is defined as total network delay. The controller mainly suffers from sensor-to-controller delay. When such delay is transformed into discrete-time domain it mostly possesses non-integer type of values. Such network delays in discrete-time domain are defined as fractional delays which may be either deterministic

or random in nature. So, it is important to compensate the effect of deterministic as well random fractional delay in discrete-time domain at each sampling instant.

Further, as mentioned above there are also possibilities of packet loss during the transmission of data packets from sensor to controller as well as controller to actuator. The packet loss takes place due to heavy network load, network congestion and node competition. In discrete-time domain, the network-induced delay greater than one sampling time is also considered as packet loss. The nature of delay and packet loss is mainly dependent on configuration of network medium.

In recent years, many algorithms have been studied for the stability analysis and control design of NCS in discrete-time domain that includes state feedback controller,  $H_{\infty}$ , Model predictive controller and sliding mode controller etc... Among them Sliding Mode Controller is one of the robust control algorithms because of its invariance properties to parameter variation and uncertainties.

This thesis presents novel algorithms for designing Discrete-time Sliding Mode Controller (DSMC) for NCS having both types of fractional delays i.e. deterministic and random alongwith different packet loss conditions i.e. single packet loss and multiple packet loss.

**KEYWORDS:** Networked control system, Discrete-time sliding mode control, Fractional delay, Packet loss, Multi-rate output feedback.

### ACKNOWLEDGEMENTS

I am indebted to various people and organizations during my doctoral research since I received this opportunity in 2012. I would like to express my deepest gratitude to Dr. Axay Mehta, for his great support and guidance throughout my research. I remember the my final M. Tech dissertation when I had first interaction with him. It was then, I was inspired to work for Ph.D. under his guidance. I received this opportunity in 2012. Technical visits with him to Kanpur and Istanbul taught me the greatest lessons of life.

I am obliged to the Trustees and Principal at SVIT, Vasad for allowing me to proceed towards my doctoral research. I have always received great cooperation from my colleagues during this tenure. I am grateful to IC department for establishing advanced control lab and providing me facility to validate my algorithms on real-time systems. I am thankful to my Doctoral Progress Committee members, Dr. Chetan Bhatt and Dr. Nilay Bhuptani for their valuable suggestions during periodic reviews. I thank Dr. B. Bandopadhyay, (IIT Bombay) for giving me an opportunity to attend STTP on Sliding Mode Control (SMC) in January 2013. Concepts cleared during four days of STTP had helped me throughout my work. I am thankful to Science Education and Research Board (SERB), New Delhi and Gujarat Technological University (GTU), Ahmedabad for providing me financial support to attend international conference at Malaysia and Canada. I am also thankful to all my friends and colleagues. Technical and non-technical discussions with Dr. Dipali Thakkar and Prof. Nirali Rathod boosted my work at various instants.

I express my gratitude towards my family for being my biggest supporters. I am thankful to my sister and brother-in-law being very much co-operative during these years. I would express my gratitude towards my loving wife, who constantly supported me in every phase of life. Without her support it would not have been possible for me to attend international conferences at Malaysia, Turkey and Canada. As we both would be submitting our thesis this year (2017), the time spent in sharing crazy as well as

creative thoughts during this phase is the most memorable. I would thank my parentsin-law who have always been greatly supportive.

I want to dedicate this thesis to my late father, who always wished to see me a doctorate. I am blessed and honoured to be able to fulfil his dream. The major credit of my thesis completion belongs to my dearest mother. Without the inspiration, drive and support that she gave me, I could not have achieved all that I have. Lastly, I thank all my family members who have helped me socially and psychologically.

# **TABLE OF CONTENTS**

A	BSTR	ACT		i
A	CKNO	OWLEI	DGEMENTS	iii
Al	BBRE	VIATI	ONS	ix
LI	ST O	F SYM	BOLS	X
LI	ST O	F FIGU	JRES	xix
LI	ST O	F TAB	LES	XX
1	INT	RODU	CTION AND LITERATURE SURVEY	1
	1.1	Introd	uction	1
		1.1.1	Overview of Network Control System	1
		1.1.2	Structure of Network Control System	2
		1.1.3	Concerns in Network Control System	4
		1.1.4	Advantages and Applications of NCS	5
	1.2	Literat	ture Survey	5
		1.2.1	Continuous-time domain	6
		1.2.2	Discrete-time domain	9
		1.2.3	Packet losses	10
		1.2.4	Output feedback	12
	1.3	Thesis	contribution	14
	1.4	Thesis	structure	14
2	PRF ING	ELIMIN MODI	NARIES OF NETWORK CONTROL SYSTEM AND SLID- E CONTROL	17
	2.1	Introd	uction	17
	2.2	Netwo	orked Control System (NCS)	17
		2.2.1	Delays and Packet Loss in NCS	17

		2.2.2	Sliding Mode Control with NCS	20
	2.3	Brief F	Review of Sliding Mode Control (SMC) Technique	21
		2.3.1	Origin of SMC	21
		2.3.2	Continuous-time Sliding Mode Control	24
		2.3.3	Discrete-time Sliding Mode Control	27
		2.3.4	Advantages of discrete-time sliding mode control over continuous time sliding mode control	- 31
	2.4	Challe	nges in NCS with SMC	31
	2.5	Conclu	usion	32
3	DES ING	IGN O TYPE	F DISCRETE-TIME SLIDING MODE CONTROLLER (SWIT) ) FOR FRACTIONAL DELAY	ГСН- 33
	3.1	Introdu	uction	33
	3.2	Netwo	rked Control System with Fractional Delay Compensation	33
	3.3	Proble	m Formulation	34
	3.4	Sliding	g Surface Design for Deterministic Network Delays	36
	3.5	Discre Thiran	te-Time Sliding Mode Control (Switching type) for NCS Using 's Delay Approximation	39
		3.5.1	Stability Analysis	41
	3.6	Result	s and Discussions	43
		3.6.1	System Description	43
		3.6.2	Simulation and Experimental Results of Brushless DC Motor	45
	3.7	Conclu	usion	53
4	DES SWI	IGN OI TCHIN	F DISCRETE-TIME SLIDING MODE CONTROLLER (NON NG TYPE) FOR FRACTIONAL DELAY	- 54
	4.1	Introdu	uction	54
	4.2	Proble	m Formulation	54
	4.3	Discre ran's E	te-Time Sliding Mode Control (Non-Switching type) Using Thi- Delay Approximation	55
		4.3.1	Stability Analysis	57
	4.4	Result	s and Discussions	58
		4.4.1	Simulation Results with Illustrative Example	58
		4.4.2	Simulation and Experimental Results of Brushless DC Motor	64
		4.4.3	Simulation Results With Real-Time Networks	72

	4.5	Conclu	asion	84
5	MU CON	LTIRAT NTROL	TE OUTPUT FEEDBACK DISCRETE-TIME SLIDING MOD LER FOR FRACTIONAL DELAY COMPENSATION	DE 85
	5.1	Introdu	uction	85
	5.2	Proble	m Formulation	85
	5.3	Desigr	of Sliding Surface Using Multirate Output Feedback	88
	5.4	Desigr Using	n of Discrete-Time Networked Sliding Mode Control (DNSMC) Multirate Output Feedback	90
		5.4.1	Stability	91
	5.5	Result	s and Discussion	93
		5.5.1	Simulation Results	93
	5.6	Conclu	usion	99
6	DIS FRA	CRETE ACTION	C-TIME SLIDING MODE CONTROLLER FOR RANDOM NAL DELAYS AND PACKET LOSS	100
	6.1	Introdu	uction	100
	6.2	Proble	m Formulation	100
	6.3	Sliding	g Surface with Random Fractional Delay and Packet Loss	103
	6.4	Discre dom F	te-Time Networked Sliding Mode Control for NCSs With Ran- ractional Delays and Packet Loss	107
		6.4.1	Stability Analysis	109
	6.5	Result	s and Discussions	111
		6.5.1	Simulation Results	111
		6.5.2	Experimental Results	119
	6.6	Conclu	usion	123
7	DIS WIT	CRETE	C-TIME NETWORKED SLIDING MODE CONTROL (DNSM LPTIPLE PACKET TRANSMISSION POLICY	IC) 124
	7.1	Introdu	uction	124
	7.2	Proble	m Formulation	125
	7.3	Desigr	of Sliding Surface With Multiple Packet Loss	129
	7.4	Discre lay and	te-Time Sliding Mode Control Law With Random Fractional De- d Multiple Packet Loss	131
		7.4.1	Stability Analysis	133
	7.5	Result	s and Discussions	134

		7.5.1	Simulation Results	135
		7.5.2	Experimetal Results	141
	7.6	Conclu	usion	146
8	CON	NCLUS	ION, FUTURE SCOPE AND CHALLENGES	147
8	<b>CON</b> 8.1	NCLUS Conclu	ION, FUTURE SCOPE AND CHALLENGES	<b>147</b> 147

## ABBREVIATIONS

- NCS Networked Control System
- CAN Controller Area Network
- LAN Local Area Network
- WAN Wide Area Network
- MAN Metropolitan Area Network
- PAN Personal Area Network
- **FIFO** First In First Out
- TOD Try-one-discard
- GSM Gain Schedule Middleware
- **RTT** Round Trip Time
- WSN Wireless Sensor Networks
- **ZOH** Zero-Order Hold
- **RMPC** Robust Model Predictive Control
- LQG Linear Quadratic Gaussian
- SMC Sliding Mode Control
- VSC Variable Structure Control
- LQR Linear Quadratic Regulator
- **DSMC** Discrete-time Sliding Mode Control
- SISO Single Input Single Output
- LTI Linear Time Invariant
- CSMA/CD Carrier Sense Multiple Acces Collision Detection
- MROF Multi-rate Output Feedback
- **DNSMC** Discrete-time Networked Sliding Mode Control

# LIST OF SYMBOLS

$C_M$	centeralized controller
x(t)	plant state vector in continuous-time domain
u(t)	control input signal in continuous-time domain
d(t)	slow time varying disturbance signal in continuous-time domain
y(t)	output signal in continuous-time domain
x(k)	plant state vector in discrete-time domain
u(k)	control input signal in discrete-time domain
d(k)	disturbance signal in discrete-time domain
y(k)	output signal in discrete-time domain
A	system matrix in continuous-time domain
В	control input matrix in continuous-time domain
C	output matrix
F	system matrix in discrete-time domain
G	control input matrix in discrete-time domain
$a_1, a_2, M_s, c$	constants
$a_1, a_2, M_s, c$ s(t)	constants sliding surface in continuous-time domain
$a_1, a_2, M_s, c$ $s(t)$ $s_d(k)$	constants sliding surface in continuous-time domain priori function
$a_1, a_2, M_s, c$ $s(t)$ $s_d(k)$ $k_s, \iota, \psi$	constants sliding surface in continuous-time domain priori function user defined constant
$a_1, a_2, M_s, c$ $s(t)$ $s_d(k)$ $k_s, \iota, \psi$ $\delta_0$	constants sliding surface in continuous-time domain priori function user defined constant positive offset
$a_1, a_2, M_s, c$ $s(t)$ $s_d(k)$ $k_s, \iota, \psi$ $\delta_0$ $p_s, k'$	constants sliding surface in continuous-time domain priori function user defined constant positive offset positive integer
$a_1, a_2, M_s, c$ $s(t)$ $s_d(k)$ $k_s, \iota, \psi$ $\delta_0$ $p_s, k'$ $k_t$	constants sliding surface in continuous-time domain priori function user defined constant positive offset positive integer user defined gain
$a_1, a_2, M_s, c$ s(t) $s_d(k)$ $k_s, \iota, \psi$ $\delta_0$ $p_s, k'$ $k_t$ $k_t^+, k_t^-$	constants sliding surface in continuous-time domain priori function user defined constant positive offset positive integer user defined gain lower and upper bounds coefficients
$a_{1}, a_{2}, M_{s}, c$ s(t) $s_{d}(k)$ $k_{s}, \iota, \psi$ $\delta_{0}$ $p_{s}, k'$ $k_{t}$ $k_{t}^{+}, k_{t}^{-}$ $\hat{S}$	constants sliding surface in continuous-time domain priori function user defined constant positive offset positive integer user defined gain lower and upper bounds coefficients model uncertainty
$a_1, a_2, M_s, c$ s(t) $s_d(k)$ $k_s, \iota, \psi$ $\delta_0$ $p_s, k'$ $k_t$ $k_t^+, k_t^-$ $\hat{S}$ $S_1, S_2$	constantssliding surface in continuous-time domainpriori functionuser defined constantpositive offsetpositive integeruser defined gainlower and upper bounds coefficientsmodel uncertaintymean and deviated value of $\hat{S}$
$a_1, a_2, M_s, c$ s(t) $s_d(k)$ $k_s, \iota, \psi$ $\delta_0$ $p_s, k'$ $k_t$ $k_t$ $k_t^+, k_t^-$ $\hat{S}$ $S_1, S_2$ $S_u, S_l$	constantssliding surface in continuous-time domainpriori functionuser defined constantpositive offsetpositive integeruser defined gainlower and upper bounds coefficientsmodel uncertaintymean and deviated value of $\hat{S}$ upper and lower bounds of $\hat{S}$
$a_1, a_2, M_s, c$ s(t) $s_d(k)$ $k_s, \iota, \psi$ $\delta_0$ $p_s, k'$ $k_t$ $k_t^+, k_t^-$ $\hat{S}$ $S_1, S_2$ $S_u, S_l$ $\tau_t$	constantssliding surface in continuous-time domainpriori functionuser defined constantpositive offsetpositive integeruser defined gainlower and upper bounds coefficientsmodel uncertaintymean and deviated value of $\hat{S}$ upper and lower bounds of $\hat{S}$ total delay which includes system and network delay in continuous-time domain
$a_1, a_2, M_s, c$ s(t) $s_d(k)$ $k_s, \iota, \psi$ $\delta_0$ $p_s, k'$ $k_t$ $k_t^+, k_t^-$ $\hat{S}$ $S_1, S_2$ $S_u, S_l$ $\tau_t$ $\tau$	constants sliding surface in continuous-time domain priori function user defined constant positive offset positive integer user defined gain lower and upper bounds coefficients model uncertainty mean and deviated value of Ŝ upper and lower bounds of Ŝ total delay which includes system and network delay in continuous-time domain total network delay in continuous-time domain

$ au_{ca}$	controller to actuator in continuous-time domain
h	sampling interval
au'	deterministic total network fractional delay
$ au_{sc}^{\prime}$	deterministic sensor to controller fractional delay
$ au_{ca}^{\prime}$	deterministic controller to actuator fractional delay
$d_l$	lower bound of disturbance
$d_u$	upper bound of disturbance
$ au_p$	system delay in continuous-time domain
$ au_{sp}$	sensor processing delay
$ au_{cp}$	controller computational delay
$ au_{ap}$	actuator processing delay
ν	fractional part of delay
l	order of approximation
δ	signal transmission delay
s(k)	sliding variable in discrete-time domain
$C_s$	sliding gain
$\alpha$	parameter that depends on sensor to controller fractional delay
Q, R	matrices of appropriate dimensions
$\epsilon, q$	user defined constants
sgn	signum function
$d_1$	mean value of disturbance
$d_2$	deviated value of disturbance
$d_s$	compensated disturbance signal
$\eta,\beta,\rho,\gamma$	smallest parameter constant
$V_s$	lyapunov function
$\Phi, \kappa$	stability parameters
$\alpha'$	parameter computed based on actuator to controller fractional delay
$u_a$	compensated control signal at actuator side
$T_s$	settling time
$ au_{CAN}$	amount of network delay in CAN medium
$ au_{ETHER}$	amount network delay in Switched Ethernet medium
heta(s)	output of the system (position)
$V_m$	input to the system

$J_m$	rotor inertia
$R_m$	terminal resistance
$K_m$	motor back emf constant
$ au_r$	total random network delay in continuous-time domain
$\hat{ au}$	total random network fractional delay in discrete-time domain
$ au_{rsc}$	random sensor to controller delay in continuous-time domain
$ au_{rca}$	random controller to actuator delay in continuous-time domain
$\hat{ au}_{sc}$	random sensor to controller fractional delay in discrete-time domain
$\hat{\tau}_{ca}$	random controller to actuator fractional delay in discrete-time domain
$\hat{\tau}_l$	lower bound of random fractional delay
$\hat{\tau}_u$	upper bound of random fractional delay
$\bar{lpha}, \bar{eta}$	probability of state and control data packet loss
$x_c'(k)$	commuicated state variable over the network
$\{d_1, d_2,, d_q\}$	values in a finite set
$\beta_v$	positive scalar quantity
$\alpha(k), d_v$	stochastic variables
$E\{d_v\}$	expectation of stochastic variable $d_v$
w	number of trials
λ	average number of events per interval
e	Euler's constant
$arsigma,\gamma'$	random parameter generated using Thiran Approximation
$E\{\alpha(k)\}$	expectation of stochastic variable $\alpha(k)$
$u_c(k)$	communicated control signal over the network
Γ	stability parameter computed using packet loss and random fractional delay
$\phi$	sampling rate of control input signal
ζ	sampling rate of output signal
$F_{\phi}$	system matrix sampled at $\phi$ sampling interval
$G_{\phi}$	control input matrix sampled at $\phi$ sampling interval
Λ	positive integer
$F_{\zeta}$	system matrix sampled at $\zeta$ sampling interval
$G_{\zeta}$	control input matrix sampled at $\zeta$ sampling interval
$\hat{x}$	estimated state variable computed using multirate output feedback approach
$y_k$	output stack

# LIST OF FIGURES

1.1	Conceptual model of NCS	2
1.2	Shared structure of NCS	3
1.3	Hierarchical structure of NCS	3
1.4	Direct structure of NCS	4
2.1	Block Diagram of Network Control System	18
2.2	State Trajectories of system in Mode-I	22
2.3	State Trajectories of system in Mode-II	22
2.4	Combined system response	22
2.5	Sliding mode for relay system	24
2.6	Phases of Sliding Mode Control	25
3.1	Block diagram of NCS with time delay compensation	34
3.2	Step reponse of Thiran Approximation with $\tau'_{sc}=1$	38
3.3	Step reponse of Thiran Approximation with $\tau_{sc}^{\prime}=0.5$	38
3.4	Block Diagram of Qnet DC Servo Motor Components	43
3.5	Experimental Setup of Quanser DC Servo Motor	44
3.6	Simulation as well as Experimental results for tracking and compen- sated sliding surface for $\tau$ =12.8msec.	46
3.7	Simulation as well as Experimental results of control signal and track- ing response for $\tau$ =12.8msec and $\tau$ =24msec	47
3.8	Simulation as well as Experimental results of compensated sliding surface and control signal for $\tau$ =24msec.	48
3.9	Simulation as well as Experimental results of tracking and compensated sliding surface for $\tau$ =28msec	49
3.10	Simulation as well as Experimental results of control efforts for $\tau$ =28msec along with tracking response under external disturbances	50
3.11	Comparison of proposed algorithm and conventional sliding mode control for $\tau$ =12.8msec	52
4.1	Slow time varying disturbance $d(k)$	59
4.2	Total network delay $\tau$	60

4.3	State variable $x_1$ with initial condition $x_1=1$	60
4.4	State variable $x_2$ with initial condition $x_2=1$	60
4.5	Magnified result of state variables $x_1, x_2 \ldots \ldots \ldots \ldots \ldots$	60
4.6	Compensated sliding surface $s(k)$	61
4.7	Magnified result of compensated sliding surface $s(k)$	61
4.8	Control signal $u(k)$	61
4.9	Compensated control signal $u_a(k)$	61
4.10	Magnified compensated control signal $u_a(k)$	62
4.11	Response of SNR	62
4.12	Nature of state variables for different SNR	62
4.13	Result of Stability	62
4.14	Simulation as well as Experimental results of tracking and compensated sliding surface for $\tau$ =10msec	65
4.15	Simulation as well as Experimental results of control signal and track- ing response for $\tau$ =18msec and $\tau$ =24msec	66
4.16	Simulation as well as Experimental results of compensated sliding surface and control signal for $\tau$ =18msec	67
4.17	Simulation as well as Experimental results of tracking and compensated sliding surface for $\tau$ =28msec	68
4.18	Simulation as well as Experimental results of control signal for $\tau$ =28msec along with tracking response	69
4.19	Comparison of proposed algorithm, switching based sliding mode control and conventional sliding mode control for $\tau$ =12.8msec	71
4.20	Control signal response of coventional SMC algorithm for $\tau$ =12.8msec	72
4.21	Scheduling policies of sensor to actuator with CAN network under ideal condition	74
4.22	Scheduling policies of sensor to actuator with CAN network under traf- fic conditions	74
4.23	Tracking response of system with CAN network under ideal condition	74
4.24	Tracking response of system with CAN network under traffic conditions	74
4.25	Magnified tracking response with CAN network under ideal conditions	75
4.26	Magnified tracking response with CAN network under traffic conditions	75
4.27	Compensated sliding variable $s(k)$	75
4.28	Magnified compensated sliding variable $s(k)$	75
4.29	Control signal $u(k)$	75

4.30	Magnified control signal $u(k)$	76
4.31	Tracking response of system when packet loss is is $10\%$	76
4.32	Tracking response of system when packet loss is $30\%$	76
4.33	Tracking response of system when packet loss is $50\%$	76
4.34	Scheduling policy of sensor to controller when packet loss is $10\%$ $$ .	76
4.35	Scheduling policy of sensor to controller when packet loss is $30\%$ $$ .	77
4.36	Scheduling policy of sensor to controller when packet loss is $50\%$ $$ .	77
4.37	Scheduling policies of sensor to actuator of Switched Ethernet network under ideal condition	78
4.38	Scheduling policies of sensor to actuator of Switched Ethernet network under traffic condition	78
4.39	Tracking response of the system with Switched Ethernet as networked medium under idle condition.	78
4.40	Tracking response of the system with Switched Ethernet as networked medium with traffic condition	79
4.41	Magnified tracking response of the system with network under ideal condition	79
4.42	Magnified tracking response of the system with network under traffic load	79
4.43	Compensated sliding variable $s(k)$	79
4.44	Magnified compensated sliding variable $s(k)$	80
4.45	Control signal $u(k)$	80
4.46	Magnified control signal $u(k)$	80
4.47	Tracking response of system when packet loss is $10\%$	80
4.48	Tracking response of system when packet loss is $30\%$	80
4.49	Tracking response of system when packet loss is $50\%$	81
4.50	Scheduling policy of sensor to controller when packet loss is $10\%$ $$ .	81
4.51	Scheduling policy of sensor to controller when packet loss is $30\%$ $$ .	81
4.52	Scheduling policy of sensor to controller when packet loss is $50\%$ $$ .	81
4.53	Time delay compensation scheme with CAN as a networked medium.	82
4.54	Tracking response of conventional SMC with CAN as a communication medium.	83
4.55	Time delay compensation using Switched Ethernet as a communication medium.	83

4.56	Tracking response of conventional SMC with Switched Ethernet as a networked medium.	83
5.1	Block diagram of NCS with multirate output feedback approach	86
5.2	Actual state $x_1$ and estimated state $\hat{x}_1$ with initial condition $x_1=0.5$ for $\tau = 12.8msec$	94
5.3	Actual state $x_2$ and estimated state $\hat{x}_2$ with initial condition $x_2=0.5$ for $\tau = 12.8msec$	94
5.4	Magnified result of actual state $x_1$ and estimated state $\hat{x}_1$ with initial condition $x_1=0.5$ for $\tau = 12.8msec$	94
5.5	Magnified result of actual state $x_2$ and estimated state $\hat{x}_2$ with initial condition $x_2=0.5$ for $\tau = 12.8msec$	95
5.6	Sliding surface $s(k)$ for $\tau = 12.8msec$	95
5.7	Magnified sliding surface $s(k)$ for $\tau = 12.8msec$	95
5.8	Control signal $u(k)$ for $\tau = 12.8msec$	95
5.9	Magnified control signal $u(k)$ for $\tau = 12.8msec \dots \dots$	95
5.10	Actual state $x_1$ and estimated state $\hat{x}_1$ with initial condition $x_1=0.5$ for $\tau = 25.6msec$	96
5.11	Actual state $x_2$ and estimated state $\hat{x}_2$ with initial condition $x_2=0.5$ for $\tau = 25.6msec$	96
5.12	Magnified result of actual state $x_1$ and estimated state $\hat{x}_1$ with initial condition $x_1=0.5$ for $\tau = 25.6msec$	96
5.13	Magnified result of actual state $x_2$ and estimated state $\hat{x}_2$ with initial condition $x_2$ =0.5 for $\tau = 25.6msec$	96
5.14	Sliding surface $s(k)$ for $\tau = 25.6msec$	97
5.15	Magnified sliding surface $s(k)$ for $\tau = 25.6msec$	97
5.16	Control Signal $u(k)$ for $\tau = 25.6msec$	97
5.17	Magnified control signal $u(k)$ for $\tau = 25.6msec \dots \dots$	97
5.18	Output stack $y_k$	97
6.1	Block diagram of NCS with fractional delay compensation and packet loss	101
6.2	Total networked fractional delay $\tau$	112
6.3	Magnified plot for sensor to controller fractional delay	112
6.4	Magnified plot for controller to actuator fractional delay	112
6.5	State variable $x_1(k)$	113
6.6	Magnified state variable $x_1(k)$	113

6.7	State variable $x_2(k)$	113
6.8	Magnified state variable $x_2(k)$	113
6.9	Compensated sliding variable $s(k)$	114
6.10	Magnified compensated sliding variable $s(k)$	114
6.11	Control Signal $u(k)$	114
6.12	Magnified control signal $u(k)$	114
6.13	Compensated control signal $u_a(k)$	115
6.14	Magnified compensated control signal $u_a(k)$	115
6.15	Compensated control signal $u_a(k)$ with 10% packet loss	115
6.16	Magnified compensated control signal $u_a(k)$ with $10\%$ packet loss $$ .	115
6.17	Compensated control signal $u_a(k)$ with 30% packet loss	116
6.18	Magnified compensated control signal $u_a(k)$ with $30\%$ packet loss $$ .	116
6.19	Compensated control signal $u_a(k)$ with 50% packet loss	116
6.20	Magnified compensated control signal $u_a(k)$ with $50\%$ packet loss .	116
6.21	Compensated control signal $u_a(k)$ with fractional delay greater than sampling interval	117
6.22	Magnified compensated control signal $u_a(k)$ with fractional delay greater than sampling interval	117
6.23	Results of position control of DC Motor with $10\%$ packet loss $~$	119
6.24	Result of magnified position control of DC Motor with $10\%$ packet loss	120
6.25	Result of compensated sliding variable with $10\%$ packet loss $\ldots$ .	120
6.26	Magnified compensated sliding variable with $10\%$ packet loss $~$	120
6.27	Control signal with $10\%$ packet loss $\ldots \ldots \ldots \ldots \ldots \ldots$	120
6.28	Magnified control signal with $10\%$ packet loss	120
6.29	Compensated control signal $u_a(k)$ with 10% packet loss	121
6.30	Magnified compensated control signal $u_a(k)$ with $10\%$ packet loss .	121
6.31	Compensated control signal $u_a(k)$ with 30% packet loss	121
6.32	Magnified compensated control signal $u_a(k)$ with 30% packet loss $$ .	121
6.33	Tracking Response with network delays greater than sampling interval	121
6.34	Magnified tracking response with network delays greater than sampling interval	122
7.1	Block diagram of NCS with multiple packets transmission	125
7.2	Schematic Diagram of NCSs with state data frame splitted in two parts	126

7.3	Total networked fractional delay	135
7.4	Magnified sensor to controller random fractional delay	136
7.5	Magnified controller to actuator random fractional delay	136
7.6	State variable $x_1(k)$	136
7.7	Magnified state variable $x_1(k)$	136
7.8	State variable $x_2(k)$	137
7.9	Magnified state variable $x_2(k)$	137
7.10	Compensated sliding variable $s(k)$	137
7.11	Magnified compensated sliding variable $s(k)$	137
7.12	Control signal $u(k)$	138
7.13	Magnified control signal $u(k)$	138
7.14	Compensated control signal $u_a(k)$	138
7.15	Magnified compensated control signal $u_a(k)$	138
7.16	Compensated control signal $u_a(k)$ with 10% packet loss	139
7.17	Magnified compensated control signal $u_a(k)$ with 10% packet loss $\ .$	139
7.18	Compensated control signal $u_a(k)$ with 20% packet loss	139
7.19	Magnified compensated control signal $u_a(k)$ with 20% packet loss $$ .	139
7.20	Compensated control signal $u_a(k)$ with 30% packet loss	140
7.21	Magnified compensated control signal $u_a(k)$ with 30% packet loss $% \left( \frac{1}{2} \right) = 0.0000000000000000000000000000000000$	140
7.22	Position Control of DC Motor under multiple packet transmission .	142
7.23	Magnified position control	142
7.24	Compensated sliding variable $s(k)$	142
7.25	Magnified compensated sliding variable $s(k)$	142
7.26	Control signal $u(k)$	142
7.27	Magnified control signal $u(k)$	143
7.28	Compensated control signal $u_a(k)$	143
7.29	Magnified compensated control signal $u_a(k)$	143
7.30	Compensated control signal $u_a(k)$ with 10% multiple packet losses .	143
7.31	Magnified compensated control signal $u_a(k)$ with 10% multiple packet loss	143
7.32	Compensated control signal $u_a(k)$ with 20% multiple packet losses .	144
7.33	Magnified compensated control signal $u_a(k)$ with 20% multiple packet losses	144

7.34	Compensated control signal $u_a(k)$ with 30% multiple packet losses .	144
7.35	Magnified compensated control signal $u_a(k)$ with 30% multiple packet	
	losses	144

# LIST OF TABLES

3.1	Simulations and Experimental Results Under different Networked De- lays	53
3.2	Comparison of Proposed Algorithm with Conventional SMC	53
4.1	Comparison of Proposed Algorithm, Switching based SMC and Conventional SMC	72
4.2	Comparison of Proposed Algorithm with Conventional SMC in True Time	82

## **CHAPTER 1**

### **INTRODUCTION AND LITERATURE SURVEY**

### **1.1 Introduction**

#### **1.1.1** Overview of Network Control System

The advent of communication networks, introduced the concept of remotely controlling a system, which gave birth to Networked Control Systems (NCSs). The classical definition of NCSs can be as follows: When a traditional feedback control system is closed via a communication channel, which may be shared with other nodes outside the control system, then the control system is called a Network Control System (NCS) (Ling et al., 2007). An NCS can also be defined as a feedback control system wherein the control loops are closed through a real-time network (Gupta and Chow, 2010; Asif and Webb, 2015). The defining feature of an NCS is that information (reference input, plant output, control input, etc...) is exchanged using a network among control system components (sensors, controllers, actuators, etc...). Figure (1.1) shows the conceptual model of the NCS. The networked medium can be wired or wireless depending on the type of the applications. In NCS, when any form of data that is transmitted through wires then such medium is called as wired network while when any form of data that is transmitted without use of electrical conductor then such medium is called as wireless network medium. The main advantage of using wired medium is data security, while the main advantage of using the wireless network medium is to get rid of wires when NCS architecture is complex. In NCS the wired communication is carried out through CAN, Switched Ethernet, Ethernet, Profibus and Profinet networked medium while wireless communication is done through Wirelss LAN, Wireless PAN, Wireless MAN or Wireless WAN (Asif and Webb, 2015).



Figure 1.1: Conceptual model of NCS (Gupta and Chow, 2010)

#### **1.1.2** Structure of Network Control System

In general there are two major types of control systems that utilize communication networks: (1) shared-network control system and (2) remote control system. Figure (1.2) shows the architecture of shared network control system. It can be noticed that using shared-network control system the transfer of information from sensors to controllers and control signals from controllers to actuators can greatly reduce the complexity of connections and provides more flexibility in installation, ease of maintenance and troubleshooting. Moreover, it also provides the communication among control loops [Gupta and Chow (2010), Zhang et al. (2013b), Zhao et al. (2015)]. This feature is extremely useful when a control loop exchanges information with other control loops to perform more sophisticated controls, such as fault accommodation and control. Similar structures for network-based control have been applied to automobiles and industrial plants. The other control system that utilizes the network medium is remote control system. In remote control system, the place where central controller is installed is called a local site and the place where plant is installed is called a remote site. The data transfer between local site and remote site is done through communication network. Sometimes the remote control system is also defined as tele-operation control system. There are two general approaches to design an NCS using remote control system: (i) hierarchical structure and (ii) direct structure. In hierarchical structure there are several subsystems that are connected to central controller through communication network. Each subsys-



Figure 1.2: Shared structure of NCS (Gupta and Chow, 2010)

tem contains sensor, actuator, and controller by itself as depicted in Figure (1.3). In this case, a subsystem controller receives a set point from the central controller  $C_M$ . The subsystem then tries to satisfy this set point by itself. The sensor data or status signal is transmitted back via network to the central controller. Figure (1.4) shows the



Figure 1.3: Hierarchical structure of NCS (Gupta and Chow, 2010)

block diagram of direct structure. In this case, the sensor and actuator are attached to a plant, while a controller is separated from the plant by a network connection. The sensor transmits the signal to the controller through the network and controller sends back the processed control signal to the plant via actuator through the network. This type of configuration is used mainly in the process industries and haptic surgery. Many complex network control system use the combination of both the structures defined as hybrid structure.



Figure 1.4: Direct structure of NCS (Gupta and Chow, 2010)

#### 1.1.3 Concerns in Network Control System

The presence of communication medium in network control system leads to several natural issues such as:

- **Time delay**: In network control system time delay is one of the cruicial issue. The time required for the data to travel within the network is defined as time delay. The nature of time delay depends on the various factors such as network configuration, distance of communication between plant and controller, baud rate, network characteristics and network topology. The time delay can affect the performance of the system in all the structure of NCSs (shared, hierarchical and direct).
- **Packet Loss**: The next serious issue that affect the performance of NCS is packet loss. The packet loss is mainly caused due to congestion, network traffic and jitter problems. Whenever the data transmitted from sensor or controller through the network and fails to reach the destination then such condition is defined as packet loss condition. In NCS there are two types of packet loss: (i) single packet loss and (ii) multiplepacket loss. In some of the models of NCS the packet loss situation are interlinked with time delay parameter. The packet loss situation occurs in all the structure of NCSs (shared, hierarchical and direct).
- **Packet disorder**: The packet disorder issue is generally caused in wireless NCS due to heavy traffic, congestion or jitter. In wireless NCS, the communications takes place in the form of small packets. So, in order to have secure communication each packet is provided with a unique identification number in the header. While transmission if any packet is loss and fails to reach at the destination packet disorder situation takes place. If these disorder is not corrected then it severely affects the performance of the system. This situation takes place in all the structure of NCSs connected wirelessly.
- **Bandwidth Sharing**: This issue mainly occurs in the shared structure or hierarchical structure of NCSs. Both these structures provides the flexiblity of connecting large number of devices (such as plant, controller, sensor and actuator) through a common network medium. So, as the number of devices increases

the bandwidth sharing is also increased which in turn causes lesser transmission speed, congestion problem, jittering problem or networked traffic problem. This may cause further instability in the system.

• Security: The security issue is one of the major concern in NCS when the communication is done without wires. In wireless communication there are chances of hacking due to which the false data is generated at the controller side and may cause the instability in the system. This issue needs to be handled very appropriately when the communication is done through shared structure or hierarchical structure of NCS.

#### **1.1.4** Advantages and Applications of NCS

For many years now, data networking technologies have been widely applied in industrial and military control applications. These applications include manufacturing plants, automobiles, and aircraft. Connecting the control system components in these applications, such as sensors, controllers, and actuators, via a network can effectively reduce the complexity of systems, with nominal economical investments. Furthermore, network controllers allow data to be shared efficiently. It is easy to fuse the global information to take intelligent decisions over a large physical space. They eliminate unnecessary wiring. It is easy to add more sensors, actuators and controllers with very little cost and without heavy structural changes to the whole system. Most importantly, they connect cyber space to physical space making task execution from a distance easily accessible. These systems are becoming more realizable today and have a lot of potential applications, including space explorations, terrestrial exploration, factory automation, remote diagnostics and troubleshooting, hazardous environments, experimental facilities, domestic robots, automobiles, aircraft, manufacturing plant monitoring, nursing homes or hospitals, tele-robotics, smart grid etc....

#### **1.2 Literature Survey**

Due to its distinct advantages NCS has become popular in the field of control for industrial applications. Also NCS has become an active research topic among international researchers fraternity due to its wide applications. NCS generally possess a dynamic nature which results in various challenges for researchers in terms of random time delay, packet loss, multiple packet loss, packet disordering, resource allocation and bandwidth sharing. If these challenges are not handled properly, they may result in degradation of the system's performance. Among these challenges, time delay and packet loss are considered to be crucial issues in NCS that causes potential instability.

The next section represents the concise literature survey regarding the compensating the effects of network delay and packet loss in continuous-time domain and discrete-time domain.

#### **1.2.1** Continuous-time domain

Various researchers (Luck and Ray (1990a,b), Luck and Ray (1990c, 1994), Chan (1995), Yu et al. (2000), Kim et al. (2002), Montestruque and Antsaklis (2003), Yue et al. (2004), Yang (2006), Godoy and Porto (2010), Li et al. (2010), Vardhan and Kumar (2011), Urban et al. (2011)) have laid their sincere efforts for designing different control algorithms that compensates the effect of network delay. In the early stages of NCS, when modelling of random time delay was difficult to obtain, the most appropriate approach was to treat the random time delay as constant which are called as deterministic delays. Luck and Ray (1990*a*,*b*) introduced the concept of compensating the time delay in continuous-time domain. They compensated the effect of time delay by introducing the receiver buffer at the controller and actuator side. The size of the buffer was equal to sensor to controller delay and controller to actuator delay. The proposed methodology was tested under IEEE 802.4 network test bed considering the deterministic types of delays. Later on (Luck and Ray, 1990c, 1994) designed predictor-based compensator in which observer was designed to estimate the plant states and predictor was used to predict the control sequences based on the past input signals. The FIFO buffer was set at the controller side and actuator side that stores the past output measurements as well as control measurements. The size of the buffer was set according to the upper bounds of sensor to controller delay and controller to actuator delay. They also tested the efficacy of the proposed algorithm on IEEE 802.4 networked medium. Chan (1995) designed conventional form of memory feedback controller based on delay compensation method. Yu et al. (2000) designed multiple step delay compensator for NCS in the presence of dynamic noise and measurement noises. Kim et al. (2002) modelled an NCS as a switched system with constant input delays and derived the sufficient conditions for the system stability using piecewise continuous Lyapunov methods.

Montestruque and Antsaklis (2003) designed state feedback controller that compensates the effect of deterministic network delays in continuous-time domain. Yang (2006) proposed the state feedback controller in the presence of network delays in continuous-time domain. They proposed ZOH model at controller and actuator side to compensate the effect of network delay. They also assumed that the sensor is time driven device while actuator is event driven device. Godoy and Porto (2010) designed PID controller to compensate the network delay and validate the feasibility of controller through DC motor as plant and CAN (Control Area Network) bus as networked medium. Li *et al.* (2010) designed a method for internet-based network control system in a dual rate configuration to achieve load minimization and dynamic performance specifications. The remote PID controller was design which regulates the output according to desirable reference and adopts the lower sampling rate to reduce the load on the network. The performance of the system was validated for fixed network delays.

In view of this, an increasing number of researchers began to investigate different control methodologies for NCSs with random or time varying delays. Zhang *et al.* (2001) proposed the stability criteria for NCS having network delays shorter as well as longer than sampling interval. They also proposed state feedback controller using conventional estimator technique that compensates the effect of network delays having time varying nature. Similarly, Walsh et al. (2002) proposed the mathematical model of NCS considering time varying network induced delay. They derived the stability criteria for general NCS in continuous-time domain based on TOD (try-once-discard) algorithm. Yue et al. (2004) designed state feedback controller in the presence of time varying network delays. They assumed that the network delays are lesser than sampling interval. Tipsuwan and Chow (2004) proposed the concept of external gain scheduling via GSM. The GSM was used to adjust the controller gains externally at the controller output with respect to the current network traffic conditions without interrupting the internal design of controller. The network delays in the forward channel and feedback channel were modelled using RTT approach. Ji and Kim (2005) proposed state feedback control with estimator to compensates the effect of time varying network delay in the presence of matched uncertainty. They tested the efficacy of the proposed controller using Ethernet as a network medium. Ma and Zhao (2006) derived the stability criteria for closed loop NCS using the average dwell time approach and piecewise Lyapunov function method. They designed state feedback controller with estimator that takes care of sen-
sor to controller delay. Peng and Yue (2006) designed the state feedback controller for NCS considering time varying network delay in the states and matched uncertainty. Gao et al. (2007) proposed a new time delay system approach which contains multiple successive delay components in the plant states and based on that they designed the  $H_{\infty}$  controller to overcome the effect of these state delays. Liu *et al.* (2007) designed network predictive controller to overcome the effects of random network delay in continuous-time domain. The effects of random delays were compensated through network delay compensator placed on the actuator side. The network delay chooses the control input values from the control latest prediction sequence. Cuellar et al. (2007) proposed an observer based predictor using the Pade approximation technique for time lag processes. Sun and Xu (2009) modelled the random time delays using stochastic approach in continuous-time domain. They used Markov jump linear systems approach to model sensor to controller random delay while controller to actuator delay was assumed to be constant. Yuhong and Yeguo (2010) designed state feedback controller considering time varying network delay in the states and proved the closed loop NCS stability using LMI approach. Ono et al. (2010) designed a state feedback controller based on a modified Smith predictor which stabilized the plant in the presence of dead time. Vardhan and Kumar (2011) used smith predictor algorithm to compensate the effects of time varying network delays in continuous-time domain. Urban et al. (2011) studied the effect of network delays in wired and wireless networked medium using PID controller. They used CAN protocol for the wired communication and Zigbee protocol for wireless networked medium. Similarly, Ridwan and Trilaksono (2011) designed the  $H_{\infty}$  state feedback controller assuming all states variable are measurable in the presence of time-varying network delays. Vallabhan et al. (2012) have used the analytical framework approach for compensation of random time delay and packet loss. Hikichi et al. (2013) worked on continuous-time delay compensation using predictors and disturbance observer for designing a PID controller. Hu et al. (2013) designed a sliding mode intermittent controller for bidirectional associative memory (BAM) using neural networks with delays. Cac et al. (2014) used a pole placement method for compensating the time delay in the continuous-time domain. The algorithm was designed for the CAN type deterministic networked medium. Recently, Yi et al. (2014) solved the time delay problem by using the Smith predictor algorithm. The method was verified over wireless sensor networks (WSN) connected between the controller output and

plant input. Recently, Khanesar *et al.* (2015) modelled the random time delays using a uniform probability distribution function in continuous-time domain. Saravanakumar *et al.* (2016) proved the stability using a Markovian Jump approach for neural networks having varying time interval delays.

#### **1.2.2** Discrete-time domain

Like continuous-time domain, many researchers (Jacovitti and Scarano (1993), Nilsson et al. (1998), Shousong and Qixin (2003), Zhivoglyadov and Middleton (2003), Yue et al. (2005), Zhao et al. (2008), Gou (2009), Xiong and Lam (2009), Li et al. (2014), Guo et al. (2014), Yao et al. (2014), Argha et al. (2015)) have also tried to focus their work in discrete-time domain. Jacovitti and Scarano (1993) proposed various time delay estimation techniques for discrete-time systems. Nilsson et al. (1998) used stochastic approach to design state feedback controller for time varying network delays in discretetime domain. Similarly, Shousong and Qixin (2003) also used stochastic approach for designing optimal controllers for NCS. They assumed that the random network delays are greater than sampling interval. Zhivoglyadov and Middleton (2003) proposed state feedback observer technique for linear network control system to compensate the effect of random delays. Yue et al. (2005) provided the model of NCSs with random network induced delay in discrete-time domain. They designed  $H_{\infty}$  controller to compensate the effect of random delays in the presence of matched uncertainty. Zhao et al. (2008) designed integrated predictive controller for networked control system. The predictive controller is applied to generate the control predictions for each delayed sensing data and previous control information. They also designed the time delay compensator at actuator side that actively compensates the forward channel delay when control action is taken. Gou (2009) designed the state feedback controller in discrete-time domain based on variable-period sampling approach for random network delays in NCS. Xiong and Lam (2009) introduced the concept of ZOH model at controller and actuator side that compensates the effect random network delays in discrete-time domain. The proposed ZOH model has an capability of choosing the newest control input. Yang et al. (2010) proposed discrete-time sliding mode observer that estimates the random delay and compensates its effect in the presence of matched uncertainty. Li et al. (2014) designed a sliding mode predictive control for compensation of delay in a networked control system using a Kalman Predictor. They considered networked delays are random in nature with an integral multiple of sampling interval. Guo *et al.* (2014) considered the state estimation problem for wireless NCS. The sliding mode observer was designed to solve the state estimation problem considering stochastic uncertainty and time delay. Yao *et al.* (2014) designed a robust model predictive control (RMPC) and state observer for a class of time varying systems under input constraints such as matched uncertainty. Argha *et al.* (2015) designed stochastic type sliding mode controller that compensates the effect of random networked delay with values lesser than sampling interval.

Various researchers (Nilsson et al. (1998), Zhivoglyadov and Middleton (2003), Zhang et al. (2005), Yang et al. (2006), Gou (2009), Shi and Yu (2009), Dong and Kim (2012), Ge et al. (2013), Argha et al. (2015)) have also laid their sincere efforts to model random time delays in last decade. Among them Nilsson et al. (1998) introduced the time stamp technique to model the random time varying networked delay. Shousong and Qixin (2003) used stochastic approach to model the random network delays. Zhang et al. (2005), Gou (2009) as well as Dong and Kim (2012) modelled random time delay using the concept of Markov Chain process in discrete-time domain. They have used two state Markov chain model to describe sensor to controller delay and controller to actuator delay. While, Yang et al. (2006) modelled random networked delay using Bernoulli's distributed white sequence approach. Shi and Yu (2009) modelled random delays using Markov chain process and designed output feedback controller to handle the effects of random delay. Ge et al. (2013) used an independent and identically distributed approach to model the time varying networked delay and proposed state feedback controller. Recently, Argha et al. (2015) proposed Bernoulli's white sequence approach for modelling the random time delay and proposed sliding mode controller in the presence of random time delay and matched uncertainty.

#### 1.2.3 Packet losses

As mentioned above, there are also possibilities of packet loss during the transmission of data packets from sensor to controller as well as controller to actuator. The packet loss takes place due to heavy network load, network congestion and node competition. In NCS there are two types of packet losses (i) single packet loss and (ii) multiple packet loss. The single packet loss situation generally occurs when the communication of data transfer is done over a shorter distance and multiple packet loss situations generally occurs when communication of data transfer is done over a longer distance. In the research works (Zhang et al. (2001), Yue et al. (2005), Zhang et al. (2005), Yang et al. (2006), Hespanha et al. (2007), Gupta et al. (2007), Wu and Chen (2007), Shi and Yu (2009), Zhu and Yang (2009), Niu and Ho (2010), Dong and Kim (2012), Ge et al. (2013), Wen and Gao (2013), Li et al. (2014), Khanesar et al. (2015), Argha et al. (2015), Song et al. (2016)), mathematical model is proposed assuming that the packet loss within the communication medium occurs when network delay is greater than sampling interval. Zhang et al. (2001) consider the deterministic single packet loss model with packet dropouts occurring at an asymptotic rate. Yue et al. (2005) designed single and multiple packet loss model in context with random network delays. They assumed that whenever the controller and actuator are not updated the data packet loss takes place for that sampling interval. Zhang et al. (2005) considered the packet loss in correspondence with random time delay model. They also made a generalized assumption that when the delays are greater than sampling interval the data packets will be lost at the controller side. Similarly, Yang et al. (2006) also considered the same packet loss approach while modelling the random time delays. Hespanha et al. (2007) used Bernoulli's probability distribution function to derive the mathematical model of single packet loss as well as multiple packet loss. In both the cases the situation of packet loss was considered when the network delay is greater than sampling interval. Gupta et al. (2007) designed optimal LQG controller that compensates the effect of packet loss occurring within the network. Similarly, Wu and Chen (2007) used the concept of the state estimation to compensate the effect of packet loss in discrete-time domain in NCS. Shi and Yu (2009) assumed the packet loss situation while modelling the random time delays using Markov's chain process. Zhu and Yang (2009) designed state feedback controller with multiple-packet transmission. They proposed the model of NCS with multiple packet transmission and given mathematical model of packet dropout in sensor to controller channel and controller to actuator channel. Niu and Ho (2010) designed the compensator using probability function that compensates the effect of packet loss within the network. Dong and Kim (2012) used Dirac delta probability function to derive the mathematical model of packet loss assuming the single packet loss situation. Wang et al. (2013) used the concept of polytopic-uncertainty-based data drift to model the packet loss occurring in sensor to controller and controller to actuator.

They designed the  $H_{\infty}$  controller to compensate the effect of random time delay and packet loss. Wen and Gao (2013) proposed  $H_{\infty}$  controller for NCS in multiple packet transmission with random delays. They modelled multiple packet using markov's chain process. Li *et al.* (2014) designed sliding mode predictive controller under multiple packet transmission policy. Khanesar *et al.* (2015) derived the packet loss model using the concept of uniform distributed probability function with single packet loss assumption. Argha *et al.* (2015) included the random packet loss situation while modelling the random time delays using Bernoulli's distribution. Recently, Song *et al.* (2016) have proposed the packet loss model using Markov chain process. The model was validated for a single packet drop as well as successive packet drops. They proposed discrete-time integral sliding mode controller using the proposed model to compensate the effects of packet loss.

#### **1.2.4 Output feedback**

The above all literatures, discusses about design of controllers based on the state information method. The major disadvantage of these controllers is that, its performance depends upon the availability of state information. In various applications of NCS such as missile guidance control, aircraft control, chemical industries and automobile sectors it is not mandatory that all the states information is available. In such cases, it is better to design the controller based on output feedback method. The main advantage of this method is that, the performance of controller depends on the availability of output information which is always available. Recently, many researchers have paid much attention on designing the controllers based on output feedback approach in NCS. Mu et al. (2004) proposed Luenberger output feedback based controller for discrete-time networked systems. The controller consists of two parts: a state observer that estimates plant states from the output when it is available via network, and a model of the plant that is used to generate the control signal when plant output is not available. Similarly, Hespanha and Naghshtabrizi (2005) designed observer based Luenberger output feedback to deal with these problems for anticipative and non anticipative control unit in continuous-time domain. Shi and Yu (2009) proved the stability of NCS with random time delays using output feedback method. Zhang and Xia (2011) also designed predictive controller that compensates the effect random delays in the presence of matched

uncertainty using output feedback approach. Yu and Antsaklis (2011) introduced the concept of event triggered for designing output feedback controller in NCS in the presence of time varying network delays. Zhang et al. (2013a) designed output feedback sliding mode controller to study the effect of variable time delay in the presence of matched uncertainty. Jungers et al. (2013) proved stability of NCSs including global time varying networked delay. They designed controller based on dynamic output feedback approach dependent on estimation of time varying delay. Similarly, Wang et al. (2013) designed output feedback  $H_{\infty}$  controller for NCSs with packet dropouts, network induced delays and data drift. They introduced polytopic uncertainty based data drift to model closed loop NCSs which include random time delay and packet loss. Qiu et al. (2013) designed robust output feedback controller for T-S fuzzy based affine systems with unreilable communication links with multiple packets-dropout. Later, Hong et al. (2014) designed conventional observer using output feedback approach for wireless NCSs with time varying network delays and packet dropouts. They modelled wireless NCS using asynchronous dynamic system with assumption that time varying network delays can be more or less than sampling interval. Yu et al. (2014) designed multiple dynamic output feedback controllers for networked control systems in the presence of random time delays and packet loss.

It is worth to mention here that, the time required for the data packets to travel from sensor to controller and controller to actuator is defined as total network delay. When such delay is transformed into discrete-time domain it mostly possesses non-integer type of values. Such network delays in discrete-time domain are defined as fractional delays. The networked control system has sensor to controller fractional delay present in the feedback channel and controller to actuator fractional delay present in the forward channel. The nature of both these fractional delays depends on the type of the communication medium. In NCS, when the data packets are exchanged through real time communication medium the network delay always have the fractional delay. So, it is important to compensate the effect of deterministic and random type of packet loss and matched uncertainty.

## **1.3** Thesis contribution

This thesis contributes mainly following:

- Firstly, a novel discrete-time sliding surface is proposed using the compensated state information and proposed a design of discrete-time sliding mode controller that encompasses deterministic type fractional delay and single packet loss. The Thiran's approximation technique is used for compensating the deterministic type of fractional delays. Two types of Discrete-time Sliding Mode Controllers are proposed that is Switching type and Non-switching type DSMC. The conditions for stability of the closed loop system using proposed controller are derived using the Lyapunov approach. The algorithms are checked with simulation also validated by the experimental results on servo system for various performance parameters. The proposed algorithms are also compared with conventional sliding mode controller using CAN and Switched Ethernet as network medium. The robustness properties of the algorithm are also checked with slowly varying matched uncertainties.
- The above algorithms are using the state information for the controller design but in most of the control scenario only the output information is available. The thesis incorporates the multi-rate output feedback approach for the state estimation in the closed loop. The main advantage of using multi-rate output feedback is that the error between estimated state variables and actual state variables goes to zero once the output sample is available. The proposed multi-rate output feedback discrete-time sliding mode controller also performs well in the networked environment.
- Next, the thesis proposes the discrete-time sliding surface design for the random fractional delay and single packet loss that occur within the sampling period. The random delay is compensated using Thiran's approximation technique in the presence of packet loss situation. The random fractional delay is modelled by Poisson's distribution function and Packet loss are modelled by Bernoulli's function. The closed loop stability is proved using the Lyapunov function. The efficacy of proposed novel non-switching type of DSMC is endowed by simulation results and also experimentally validated by servo system.
- Further, the proposed algorithms extended for the random fractional delay with multiple packet loss situation. The simulation as well experimental results with various fractional delay situation and matched uncertainties show the efficacy of the proposed algorithms.

## **1.4** Thesis structure

The thesis is structured as follows:

• Chapter – 1 briefs about introduction and literature survey for NCS. The chapter discusses a brief introduction of NCS with conceptual model and different stuctures of NCS. Various issues related to NCS are also discussed.

- Chapter 2 discusses the Preliminaries of Networked Control System and Sliding Mode Control. In this chapter a basic block diagram of NCS with different types of time delays that affect the performance of the system are discussed. The origin of sliding mode controller in continuous-time domain and discrete-time domain are also briefly discussed. Lastly, various challenges of NCS with SMC are also highlighted.
- The main contribution of thesis design of discrete-time sliding mode control for deterministic type fractional delay is mentioned in Chapter -3. In this chapter, the compensation of determinitic fractional delay is studied through Thiran's Approximation in the sliding surface. The discrete-time sliding mode control law is derived using proposed sliding surface with switching type reaching law. Further, the stability of the closed loop NCS is proved through Lyapunov approach. The efficacy of the proposed algorithm is tested under simulation environment and experimental environment.
- Chapter 4 briefs about the designing of non-switching type discrete-time sliding mode controller in the presence of deterministic fractional delay and matched uncertainty. In this chapter, the design of control law is based on sliding surface derived using Thiran Approximation. Further, the stability of the closed loop NCS is proved through Lyapunov approach that ensures the finite time convergence of system states within the specified band. The efficacy of the proposed algorithm is tested under simulation environment, experimental environment and real-time networks.
- Chapter 5 describes the design of discrete-time sliding mode control using multirate output feedback approach with fractional delay compensation. In this chapter, the control signal is computed based on the output measurements available at the controller side and the fractional delay is compensated using Thiran approximation. The stability of the closed loop NCS with derived control law is proved using Lyapunov approach. The simulation results are carried out in the presence of network delay and matched uncertainty in order to prove the effectiveness of proposed algorithm.
- Chapter 6 describes the design discrete-time sliding mode controller for random communication delay and packet loss. In this chapter, the compensation of random fractional delay is studied using Thiran's Approximation with packet loss condition. The mathematical models of random fractional delay and single packet loss are derived using stochastic approach. The derived discrete-time sliding mode control law is verified through simulation and implementation results in the presence of random fractional delay, packet loss and matched uncertainty.
- Chapter 7 discusses the mathematical model of multiple packet loss and design of discrete-time sliding mode control for multiple packet transmission. In this chapter, the discrete-time sliding mode control law is designed in the presence of multiple packets transmission. The multiple packet loss is modelled using probability function. The efficiency of the proposed algorithm is verified through simulaton and experimental results.
- The concluding remarks along with future scope and challenges are mentioned in Chapter 8. In this chapter, final comments and future scope of discrete-time

SMC algorithms are discussed. Lastly, various challenges are also listed that are still remain unsloved in network control system domain.

## **CHAPTER 2**

# PRELIMINARIES OF NETWORK CONTROL SYSTEM AND SLIDING MODE CONTROL

### 2.1 Introduction

In this chapter, we introduce the concept of Networked Control System (NCS), the irregularities such as time delay and packet loss that occurs in the NCS. We also discussed the various control methods available for NCS. This chapter also presents the concept of sliding mode control alongwith literature survey on SMC for NCS.

### 2.2 Networked Control System (NCS)

NCS is mainly classified in three structures: (i) shared network structure, (ii) hierarchical network structure and (iii) direct network structure. Figure (2.1) shows the block diagram of typical network control system with direct network structure. In this structure, the sensor, plant and actuator are connected to controller through communication network. The communication network in NCS transfers the data in the form of packets. The thick lines indicates the continuous-time data signal while the dotted lines indicates discrete-time data signals. As shown in Figure (2.1) the data signal transmitted from sensor to controller through the network is called as feedback channel while the control signal transmitted from controller to actuator through the same network is called as forward channel. In NCS most of the applications are based on time-sensitive parameter. So, in such cases, if the network delay increases beyond its tolerance limit the plant or the device can either be damaged or gives inferior performance.

#### 2.2.1 Delays and Packet Loss in NCS

In NCS there are four types of time delay (i) sensor to controller delay, (ii) controller to actuator delay, (iii) computational delay and (iv) processing delay. The sensor to con-



Figure 2.1: Block Diagram of Network Control System

troller delay is present in the forward channel and controller to actuator delay present in the feedback channel. The time required for the data to travel from sensor to controller is called a sensor to controller network delay. And similarly the time required for control signal to travel from controller to actuator is called a controller to actuator delay. The combination of both these delays are defined as network delays. The computational delay and processing delay are caused due to the presence of sensor, controller and actuator. So they are also defined as the system delays.

The combination of system delays and network delays is defined as total delay and mathematically it is represented as:

$$\tau_t = \tau + \tau_p, \tag{2.1}$$

where,  $\tau$  is the total network delay and  $\tau_p$  is the system delays.

The mathematical representation of total network delay and system delay is given as:

$$\tau = \tau_{sc} + \tau_{ca},\tag{2.2}$$

$$\tau_p = \tau_{sp} + \tau_{cp} + \tau_{ap}, \tag{2.3}$$

where,  $\tau_{sc}$  is sensor to controller delay,  $\tau_{ca}$  is controller to actuator delay,  $\tau_{ap}$  is actuator processing delay,  $\tau_{cp}$  is controller computational delay and  $\tau_{sp}$  is sensor processing delay. In NCS it is always assumed that the effect of system delays ( $\tau_p$ ) are negligible compared to network delays ( $\tau$ ). So, Eqn. (2.1) can be written as,

$$\tau_t = \tau. \tag{2.4}$$

Thus, from above Eqn. (2.4), it can be noticed that when real time networks are not considered in NCS the total network delays are always equal to total delay. These network delays can either be deterministic or random in nature.

The nature of the network delays  $(\tau)$  depends upon the configuration of networks while the system delays  $(\tau_p)$  are always deterministic in nature. When the communication takes place using the concept of lease line then network delays  $(\tau)$  are deterministic in nature. And when the communication medium is shared by large number of devices then network delays  $(\tau)$  are random in nature.

A natural assumption on network delays  $(\tau)$  can be made as,

$$\tau \prec h,$$
 (2.5)

Or

$$\tau_l \le \tau \le \tau_u. \tag{2.6}$$

where, *h* is sampling interval,  $\tau_l$  is lower bound of delay and  $\tau_u$  is upper bound of delay. Observing condition (2.5) and (2.6), it can be concluded that the network delay should always be bounded. During transmission if the packet gets delayed or fails to arrive at the destination within the specified condition (2.5) and (2.6), then such packets are considered as lost packets within the network.

In NCS, the packet loss occurs either in the forward channel or feedback channel. The packet loss are broadly classify in two different categories based on the distance of communication: (i) If the distance of communication is shorter, the data transfer in NCS takes place in the form of frames. Such frame when lost during transmission, is treated as single packet loss. (ii) If the distance of communication is longer, the same frame is breakdown in the form of small packets. Such packets when lost during transmission are defined as multiple packet loss.

#### 2.2.2 Sliding Mode Control with NCS

Many researchers have proposed different controller design methodologies in discretetime as well continuous-time domain for NCSs such as Gain Scheduler Middleware [Tipsuwan and Chow (2004)],  $H_{\infty}$  [Yue *et al.* (2005)], State Feedback [[Peng and Yue (2006)] [Gao *et al.* (2007)], Sliding Mode Controller (SMC) [Mehta and Bandyopadhyay (2009)], Adaptive Controller [Vallabhan *et al.* (2012)], Smith Predictive [Hikichi *et al.* (2013)], Back-stepping Controller [Yi *et al.* (2014)], fuzzy-based sliding mode control [Khanesar *et al.* (2015)] etc... Among all these controllers SMC has been an active research area in the field of NCS because it has ability to reject the disturbances which makes it more robust.

In past few decades many researchers have tried to implement the SMC algorithm in various ways to compensate the effects of network delay and packet loss in NCSs. Wang *et al.* (2011) proposed variable structure control algorithm for time delay system. They proposed the controller based on the concept of smith predictor. Gao (2013) designed integral type of sliding mode control in continuous-time domain in the presence of variable time delay and matched uncertainty. They proposed sliding mode compensator in reaching law that compensates the effect of variable network delay. Goyal *et al.* (2015) designed fuzzy-based sliding mode control in continuous-time domain. They considered the state-based delay rather than control input delay. Recently, Khanesar *et al.* (2015) proposed indirect fuzzy based sliding mode control in continuous-time domain for NCS. The effect of random time delay was compensated using Pade approximation and packet loss was compensated using probability distribution function.

With the rapid development in digital controllers, various researchers have contributed their work in discrete-time domain. The main advantage of designing the controllers in discrete-time domain is that the effects of control signal can be observed very clearly at each sampling instant. Moreover, in case of NCS since the communication is carried out in digital signal form so it better to design SMC in discrete-time rather than continuous-time domain. Xiaojuan and Jinglin (2010) designed fuzzy based sliding mode control for NCS and studied the effect of time delay using Ethernet as a network medium in discrete-time domain. Niu and Ho (2010) designed sliding mode control that compensates the effect of single packet loss using probability distribution function. Yin *et al.* (2011) designed adaptive based sliding mode control in the presence of variable time

delay and uncertainties. They considered sensor to controller delay and transformed the control signal into non-delayed form using fuzzy fusion system in discrete-time domain. Zhang *et al.* (2013*a*) designed output feedback based sliding mode controller to study the effect of variable time delay in the presence of matched uncertainty. Li *et al.* (2014) designed sliding mode controller using Kalman predictor in the presence of multiple packet transmission. The Kalman predictor was used to estimate the integer type of network delays. Goyal *et al.* (2015) proposed robust sliding mode control for nonlinear discrete-time delayed system based on neural network. They assumed network delay as deterministic in nature. Argha *et al.* (2015) proposed discrete-time sliding mode control for andom time delay and packet loss using Bernoulli's distribution. They considered integer type of network delay in discrete-time domain. Recently, Song *et al.* (2016) designed integral sliding mode controller that compensates the effect of single as well as multiple packet loss using markov's chain process.

# 2.3 Brief Review of Sliding Mode Control (SMC) Technique

#### 2.3.1 Origin of SMC

The concept of Variable Structure Control (VSC) was introduced by Emelyanov group in late 1950's (Emelyanov and Korovin, 1981). The main idea of VSC was to switch between the various control structures according to the evaluation of the system states. The concept of VSC can be understand through the following example. Consider two constituent systems given as:

$$\ddot{x} = -a_1 x, \tag{2.7}$$

$$\ddot{x} = -a_2 x, \tag{2.8}$$

where,  $0 \prec a_2 \prec a_1$ . The phase potraits of the systems are shown in Figure (2.2) and (2.3) repectively. It can be observed that both the systems are oscillartory in nature and are unstable. But, when both structures are switched at appropriate time, then,

combined system is asymptotically stable. The combined response of both the system is shown in Figure (2.4). Thus it can be noticed, that the property not present in any of the system is obtained by VSC.



Figure 2.2: State Trajectories of system in Mode-I



Figure 2.3: State Trajectories of system in Mode-II



Figure 2.4: Combined system response

While carrying out research on VSC some of the researchers [Utkin (1993), Edwards and Spurgeon (1996), Drakunov *et al.* (1990), Furuta (1990)] made an unusual observation that switching between two or more unstable control structures may result in stable control system. They introduced the notions of variable structure control, variable structure system and a new control idea called sliding mode control came into existence. The sliding mode control, or SMC, is a nonlinear control method that alters the dynamics of a nonlinear system by application of a discontinuous control signal that forces the system to "slide" along a cross-section of the system's normal behaviour. Hence, sliding mode control is a variable structure control method. The multiple control structures are designed so that trajectories always move towards an adjacent region with a different control structure, and so the ultimate trajectory will not exist entirely within one control structure. Instead, it will slide along the boundaries of the control structures. The motion of the system as it slides along these boundaries is called a sliding mode and the geometrical locus consisting of the boundaries is called the sliding surface. The conventional example of sliding mode is second order relay system which is given by,

$$\ddot{x} + a_2 \dot{x} + a_1 x = u, \tag{2.9}$$

$$u = -M_s sign(s), (2.10)$$

$$s = cx + \dot{x},\tag{2.11}$$

where,  $a_1$ ,  $a_2$ ,  $M_s$ , c are contants. From Eqn. (2.10) it can be noticed that the control input in the second order system might take only two values  $M_s$  and  $-M_s$  and causes discontinuities on the straight line s = 0 in the state plane  $(x, \dot{x})$ . Figure (2.5) shows the response of the sliding mode for the specified example with  $a_1 = a_2 = 0$ . From the result it can be observed that in the neighbourhood segment mn on the switching line s = 0, the trajectories of the system (2.9) runs in the opposite directions which leads to the appearance in sliding mode along this line [Sabanovic et al. (2004)]. Thus the Eqn. (2.11) can be interpreted as sliding mode equation. Further, it can be noticed that the order of sliding mode equation is less than the original system (2.9). Thus it can be said that the sliding mode does not depend on plant dynamics but it is determined by parameter c only. It is worth to point out that for desired performance of the closed loop system not only the sliding mode controllers have to be properly designed but the switching rule should also be chosen properly. Based on the appropriate selection of switching rule and designed controller the system states will drive onto the predefined sliding surface in finite time. This ensures stability of sliding motion on the surface and desired dynamic characteristics of the systems are achieved.



Figure 2.5: Sliding mode for relay system

Due to its lower sensitivity towards plant parameters variations and external disturbances it eliminates the necessity of exact modelling. It allows the decoupling of overall system motion into partial components of lower dimensions. This reduces the complexity of feedback design. In sliding mode control the control actions are function of discontinuous state which can be easily implemented using conventional power converters. Due to such properties SMC has been proved applicable to a wide range of applications such as robotics, electrical drives, electrical generators, motion control, process control and networked control system.

#### 2.3.2 Continuous-time Sliding Mode Control

In sliding mode control the control law consists of two important phases: (i) reaching phase (ii) sliding mode phase. When the system state is driven from any initial state to reach the switching manifold in finite time then such phase is called reaching phase and when the system is induced into the sliding motion on the switching manifolds, then such phase is defined as sliding mode phase. Figure (2.6) shows the phases of sliding mode control with s as continuous-time sliding function given by:

$$s = \{x \in X | s(x,t) = 0\}.$$
(2.12)

In order to induce the sliding mode following properties should exist: (i) the system



Figure 2.6: Phases of Sliding Mode Control

stability should be restricted to the sliding surface and (ii) the sliding mode should start within a finite time. The sufficient condition for the sliding motion to slide on the given surface is given by:

$$s\dot{s} \prec 0,$$
 (2.13)

where, s is the sliding surface and  $\dot{s}$  is rate of change of distance from the sliding surface. The condition specified in Eqn. (2.13) is called a reachability condition. The reachability condition is not sufficient for the sliding mode. The main drawback of condition mentioned in (2.13) is that, s(t) takes infinite time to reach on the sliding surface. Thus, to over come this drawback another condition is defined as:

$$s\dot{s} \prec -\eta |s|, \eta \succ 0.$$
 (2.14)

This condition is known as ' $\eta$ -reachability' condition that ensures the finite time convergence to s = 0.

As discussed earlier, the designing of the sliding mode controller includes reaching law design, sliding surface design and control law design. Let us consider the continuous-time system given by:

$$\dot{x}(t) = Ax(t) + Bu(t),$$
 (2.15)

$$y(t) = Cx(t). \tag{2.16}$$

The dynamics of the sliding function can be expressed in the form of constant-rate reaching law as:

$$\dot{s} = -k_s sgn(s), k_s \succ 0. \tag{2.17}$$

Let, the sliding surface be given by:

$$s(t) = C_s x(t), \tag{2.18}$$

where,  $C_s$  is the sliding gain that can be computed using pole-placement method based on the proper slection of poles or LQR method based on the proper slection of Q and Rmatrices.

Thus, the control law for the system (2.15, 2.16) is derived by using the condition  $\dot{s} = 0$  as:

$$u(t) = -(C_s B)^{-1} [A C_s x(t) + k_s sgn(s)].$$
(2.19)

The reaching laws proposed in the literatures [Fallaha *et al.* (2011), Mehta and Bandyopadhyay (2015)] are,

• Constant-proportional rate

$$\dot{s} = -qs - k_s sgn(s), q \succ 0 \tag{2.20}$$

• Power-rate reaching law

$$\dot{s} = -k_s |s|^{\iota} sgn(s), 0 \prec \iota \prec 1 \tag{2.21}$$

• Exponential reaching law

$$\dot{s} = -\frac{k_s}{N(s)} sgn(s), \tag{2.22}$$

where,  $N(s) = \delta_0 + (1 - \delta_0)e^{-\iota |s|^{p_0}}$ ,  $\delta_0$  is strictly positive offset less than 1,  $p_0$  is a strictly positive integer.

The main limitation of continuous-time SMC is that once the closed loop system states reach on the sliding surface, a discontinuous control term switches with high frequency which results in chattering phenomenon. The chattering is caused due to various reasons such as switching time delay, controller computation delay, dynamics of plant elements such as actuator and sensor etc... In practical applications this phenomenon is not desirable as it affects the performance of plant. While in electrical and mechanical applications it causes high heat losses and generates wear and tear of the moving parts of machines. In discrete-time sliding mode control generally low switching frequencies are required because of limited sampling frequency due to which it becomes more useful for practical implementation. Moreover, in discrete-time SMC the computation of control signal is done at each sampling interval and remains constant for that period. So due to this the system state trajectory is unable to move along the sliding surface but follows the zigzag motion about the sliding surface defined as quasi-sliding mode motion.

#### 2.3.3 Discrete-time Sliding Mode Control

The concept of discrete-time sliding mode control is first introduced by Milosavljevic (1985). In his work he proved that the sliding motion in discrete-time is more accurate than continuous-time sliding mode control. Later on, Sarpturk *et al.* (1987), Utkin (1993), Bartolini *et al.* (1995), Gao *et al.* (1995), Bartoszewicz (1996, 1998), Su *et al.* (2000) and many others extended their work in discrete-time sliding mode control.

The designed procedure of DSMC includes: (i) design of sliding surface, (ii) reaching law and (iii) control law that steers the system states to slide along predefined sliding surface over a finite interval of time. In discrete-time SMC Gao's introduced the concept of switching function in the reaching law that causes the system states to move towards the vicinity of the origin but cannot get arbitrarily closed to the origin. While Bartolini and Bartoszewicz designed the reaching law without considering the switching function. They considered that discrete-time control is naturally discontinuous in nature and thus may not require an explicit discontinuity in the control law. The reaching law proposed by them causes the system states to get arbitrarily closed to the origin. Various state-based discrete-time control algorithms are designed using different reaching laws available in Milosavljevic (1985), Sarpturk *et al.* (1987), Gao *et al.* (1995), Bartolini *et al.* (1995) and Bartoszewicz (1996, 1998) respectively.

In order to derive the discrete-time control algorithms, let us consider the continuoustime SISO system as:

$$\dot{x}(t) = Ax(t) + Bu(t),$$
 (2.23)

$$y(t) = Cx(t), \tag{2.24}$$

where,  $x \in \mathbb{R}^n$  respresents system state vector,  $u \in \mathbb{R}^m$  represents control input,  $y \in \mathbb{R}^p$ represents system output,  $A \in \mathbb{R}^{n \times n}$ ,  $B \in \mathbb{R}^{n \times m}$  and  $C \in \mathbb{R}^{p \times n}$  are the matrices of appropriate dimensions.

Let the system (2.23) and (2.24) be discretized at *h* sampling interval given by,

$$x(k+1) = Fx(k) + Gu(k),$$
(2.25)

$$y(k) = C(k). \tag{2.26}$$

As discussed earlier, the designed of sliding mode control algorithms involves the design of sliding surface and reaching law. Let the discrete-time sliding surface be given by:

$$s(k) = C_s x(k). \tag{2.27}$$

Various researchers have proposed famous reaching laws in discrete-time domain as listed below.

• Sarpturk's Reaching Law: The reaching law proposed by Sarpturk *et al.* (1987) is the direct discretization of continuous-time sliding mode given by:

$$|s(k+1)| \prec |s(k)|.$$
 (2.28)

Here, the sliding surface is always directed towards the surface and also the norm of |s(k)| monotonically decreases. The reaching law can be written in other way as,

$$(s(k+1) - s(k))sgn(s(k)) \prec 0, \tag{2.29}$$

$$(s(k+1) - s(k))sgn(s(k)) \succ 0.$$
 (2.30)

The first condition (2.29) indicates that the closed loop system state trajectories should move towards the direction of sliding surface and the second condition (2.30) indicates that the closed loop system state trajectories are not allowed to go too far in that direction. Thus observing condition (2.29) and (2.30), will lead to lower and upper bounds for control actions. The control law proposed in [Sarpturk *et al.* (1987)] is given as,

$$u(k) = -k_t(x, s)x(k),$$
 (2.31)

where,  $k_t$  is the gain given by:

$$k_t(x,s) = \begin{cases} k_t^+; & \text{when } x(k)s(k) \succ 0\\ k_t^-; & \text{when } x(k)s(k) \prec 0 \end{cases}$$

where,  $k_t^+$  and  $k_t^-$  represents the coefficients of each upper bound and lower

bound of control action that can be determined by evaluating the condition (2.29) and (2.30) respectively.

• Gao's Reaching Law: The switching based reaching law proposed by Gao *et al.* (1995) is given as,

$$s(k+1) = (1 - qh)s(k) - \epsilon h sgn(s(k)),$$
(2.32)

where, h is the sampling interval satisfying  $h \succ 0$ ,  $q, \epsilon \succ 0$  and  $1 - qh \succ 0$ . Using reaching law (2.32), the switching based control law for system (2.25, 2.26) is computed as,

$$u(k) = -(C_s G)^{-1} [F C_s x(k) - (1 - qh)s(k) + \epsilon h sgn(s(k))].$$
(2.33)

From above Eqn. (2.33), it can be noticed that there are two parameters q and  $\epsilon$  in control law for tuning the response. The discrete-time sliding mode control law proposed in (2.33) should achieve the following performances [Gao *et al.* (1995)]. (i) Starting from any initial state, the trajectory will move monotonically toward the switching plane and cross it in finite time. (ii) Once the trajectory has crossed the switching plane, it will cross the plane again in every successive sampling period, resulting in a zigzag motion about the switching plane. and (iii) The size of each successive zigzag step is nonincreasing and the trajectory stays within a specified band.

The reaching law in Eqn. (2.32) states that the state vector always move towards the quasi-sliding mode band given as:

$$s(k) \le \frac{\epsilon h}{1 - qh}.\tag{2.34}$$

From Eqn. (2.34), it can be observed that  $\epsilon$  is directly proportional to quasisliding mode band. Thus if the value of  $\epsilon$  is too large than the system will have high overshoots and could also increase the transient response. While on the other hand the value of qh should be less than unity otherwise it will speed up the transient response.

• Bartoszewicz's Reaching Law: Bartoszewicz (1996, 1998) proposed non-switching reaching law as,

$$s(k+1) = d(k) - d_0 + s_d(k+1),$$
(2.35)

where, d(k) is the unknown disturbance,  $d_0$  is the mean value of disturbance d(k)and  $\mu_d$  is minimum deviated disturbance with  $d_u$  as upper bound and  $d_l$  as lower bound. Also,  $d_0$  and  $d_2$  is given by

$$d_0 = \frac{d_l + d_u}{2}$$
 and  $d_2 = \frac{d_u - d_l}{2}$ 

 $s_d(k)$  is an priori known function such that the following applies:

• If  $s(0) \succ 2d_2$  then,

$$s_d(0) = s(0)$$

$$s_d(k)s_d \ge 0 \text{ for any } k \ge 0$$

$$s_d(k) \ge 0 \text{ for any } k \ge k'$$

$$s_d(k+1)| \prec |s_d(k)| - 2d_2 \text{ for any } k \le k'$$

These relations state that  $s_d(k)$  converges monotonically from its initial position to the origin of the state space in a finite time. Moreover, in each control step the hyperplane moves by the distance greater than  $2d_2$ . This, together with Eqn. (2.35), states that even in the case of worst combination of disturbance the reaching condition is satisfied.

If s(0) ≺ 2d<sub>2</sub> then s<sub>d</sub>(k) = 0 for any k ≥ 0. The k' in the above relations is a positive integer user defined constant which provides the faster convergence rate of the system and magnitude of the control signal u(k). The control law computed using the reaching law (2.35) for the system (2.25, 2.26) is given as,

$$u(k) = (C_s G)^{-1} [C_s F x(k) + d_0 - s_d(k+1)].$$
(2.36)

The reaching law in Eqn. (2.35) states that the system state vector always move towards the QSMB for any  $k \ge k'$  such that:

$$|s(k)| = |d(k-1) - d_0| \le d_2.$$
(2.37)

• Bartoszewicz's Reaching Law: Bartoszewicz and Lesniewski (2014) proposed the other reaching law that provides the faster convergence of sliding variable without increasing the amplitude of the control signal. The reaching law is given as,

$$s[(k+1)h] = \{1 - q[s(kh)]\}s(kh) - \hat{S}(kh) - d(kh) + d_1 + S_1, \quad (2.38)$$

where,  $\hat{S}(kh)$  represents the model uncertainty on sliding variable evolution and d(kh) represents the effect of disturbance on this variable. Further,  $S_1$  and  $d_1$  represents the mean values of  $\hat{S}(kh)$  and d(kh) repectively given as,

$$S_1 = \frac{S_u + S_l}{2},$$
 (2.39)

$$d_1 = \frac{d_u + d_l}{2},$$
 (2.40)

where,  $S_u$ ,  $S_l$  are upper and lower bounds of  $S_1$  and  $d_u$ ,  $d_l$  are upper and lower bounds of  $d_1$ .

The convergence rate factor of q[s(kh)] in Eqn. (2.38) is given as,

$$q[s(kh)] = \frac{\psi}{\psi + |s(kh)|},\tag{2.41}$$

where,  $\psi$  is designer's constant satisfying  $\psi \succ S_2 + d_2$ , where  $S_2$  and  $d_2$  the greatest possible deviation of  $\hat{S}$  and d. They are represented as,

$$S_2 = \frac{S_u - S_l}{2},$$
 (2.42)

$$d_2 = \frac{d_u - d_l}{2}.$$
 (2.43)

• The control law computed the reaching law (2.38) for the system (2.25, 2.26) is

given as,

$$u(k) = (C_s G)^{-1} [C_s F x(k) + \{1 - q[s(kh)]\} s(kh) + S_1 + d_1].$$
(2.44)

The reaching law in Eqn. (2.38) states that the system states always move towards the QSMB for any  $k \succ k_0$  such that:

$$|s(kh)| \le \frac{\psi(S_2 + d_2)}{\psi - (S_2 + d_2)}.$$
(2.45)

## 2.3.4 Advantages of discrete-time sliding mode control over continuous-

#### time sliding mode control

- With the increase in use of digital computers and microcontrollers for the implementation of control algorithms, a discrete-time model of the system is justified.
- To validate the better performance of the system it is better to implement the discrete-time algorithms rather continuous-time algorithms.
- In continuous-time sliding mode control due to high-frequency switching chattering takes place which may cause damage to the system. So, it is not used for all practical applications. While, in discrete-time sliding mode control relatively low switching frequencies are required so DSMC algorithm is more practical to implement.
- When continuous-time algorithms are implemented using digital controllers for implementation, the chattering generated around the sliding mode and stability of the sliding mode are compromised.
- A large class of discrete-time systems are computer controlled and information about the system measurements are available only at specific time instances and control inputs can only be changed at these time instances. Eg. biological systems, thyristor, radar system, economic systems, etc.

### 2.4 Challenges in NCS with SMC

There are various challenges in network control system with sliding mode control that has not been explored. Some of them are listed below:

• In literature, the researchers have proposed various SMC algorithms with time delay compensation [1, 8, 12, 20, 30, 31, 34, 41, 44, 63]. But in all the papers, it is assumed that the delays are mutiples of sampling interval. However, when the communication is carried through real-time network the delays always have fractional behaviour. Thus, there is need to design SMC for fractional delay in discrete-time domain.

- Futhermore, it can also be noticed that none of the literatures on SMC discusses about the designing of sliding surface that compensates the effect of fractional delay. So there is a need of designing the sliding surface such that it compensates the effect of fractional delay.
- The compensation of network delays and packet loss are done at the controller side. None of the researcher has tried to compensate their effects in the sliding surface.
- Till now the sliding mode controllers are designed based on the presence of multiple packet transmission but, the compensation of multiple packets loss with sliding mode control in discrete-time domain is still an open research problem in NCSs.
- Various researchers have tried to explore their work on designing the sliding mode controllers based on conventional output feedback method in the presence of network delay and packet loss. But, the compensation algorithm based on output feedback method in the field of NCSs have not been much explored.
- The designing of the higher order sliding modes in the presence of network nonidealities (such as random time delay, packet loss and matched uncertainty) is still an open research area in NCSs.

## 2.5 Conclusion

In this chapter, we introduced the basic concept of NCS and its configurations. The various irregularities of NCS are also discussed. The Sliding Mode Control (SMC) technique for NCS are also explained and at last challenges for designing the SMC for NCS are also discussed.

## **CHAPTER 3**

# DESIGN OF DISCRETE-TIME SLIDING MODE CONTROLLER (SWITCHING TYPE) FOR FRACTIONAL DELAY

## 3.1 Introduction

In this chapter, a novel approach for designing a discrete-time sliding mode controller using Thiran's delay approximation is presented, to compensate the effect of sensor to controller fractional delay and controller to actuator fractional delay in discrete-time domain. The forward channel delay is compensated at the actuator side while feedback channel delay is compensated at the sliding surface. A novel sliding surface is proposed using delay compensation and a SMC law is derived. The stability condition for the closed loop system with proposed controller is derived using Lyapunov function. The efficacy of the proposed algorithm is shown by simulation results and also validated by the experimental results considering DC Servo motor as plant in the presence of deterministic network delays and matched uncertainty.

# 3.2 Networked Control System with Fractional Delay Compensation

Figure (3.1) potrays the block diagram of NCS with time delay compensation scheme. It can be noticed that the state information as well as control information are transmitted to the controller and actuator through the network. During transmission, the state information will experience sensor to controller delay while the control information will suffer from controller to actuator delay. These delays are broadly defined as the amount of time required for the data packets to travel within the network. Thus, in order to avoid the degradation it is necessary to compensate these network delays at the controller end

as well as at actuator end. Moreover apart from these network delays it is necessary to consider the system delays.



Figure 3.1: Block diagram of NCS with time delay compensation

## **3.3** Problem Formulation

Consider the linear time invariant SISO system with network delay as:

$$\dot{x}(t) = Ax(t) + Bu(t - \tau) + Dd(t),$$
(3.1)

$$y(t) = Cx(t), \tag{3.2}$$

where  $x \in \mathbb{R}^n$  is system state vector,  $u \in \mathbb{R}^m$  is control input,  $y \in \mathbb{R}^p$  is system output,  $A \in \mathbb{R}^{n \times n}$ ,  $B \in \mathbb{R}^{n \times m}$ ,  $C \in \mathbb{R}^{p \times n}$ ,  $D \in \mathbb{R}^{n \times m}$  are the matrices of appropriate dimensions, d(t) is the matched bounded disturbance with  $|d(t)| \leq d_{max}$  and  $\tau$  is the total networked induced delay in continuous-time domain.

The discrete form of system (3.1) and (3.2) is:

$$x(k+1) = Fx(k) + Gu(k - \tau') + d(k),$$
(3.3)

$$y(k) = Cx(k), \tag{3.4}$$

where  $F = e^{Ah}$ ,  $G = \int_0^h e^{At} B dt$ ,  $d(k) = \int_0^h e^{At} D d((k+1)h - t) dt \in O(h)$ . Since  $|d(t)| \leq d_{max}$ , it can be inferred that d(k) is also bounded and O(h) [Mehta and Bandy-

opadhyay (2016)]. For simplicity, it is assumed that d(k) is slowly varying and remain constant over the interval  $kh \le t \le (k+1)h$  [Mehta and Bandyopadhyay (2016)]. The network induced fractional delay ( $\tau'$ ) occurring within the network denoted as,

$$\tau' = \frac{\tau}{h},$$

where h is the sampling interval and  $\tau$  is the total networked induced delay.

**Remark** – 1: In this work, it is considered that network induced fractional delay ( $\tau'$ ) in discrete-time domain have non-interger values so that the precise effect of network delays are compensated at each sampling instants.

Assumption -1: The total network induced fractional delay is deterministic in nature satisfying,

$$\tau \prec h,$$
 (3.5)

where  $\tau$  indicates the total network delay in continuous-time domain.

**Remark** – 2: The above condition (3.5) indicates that the values of total network induced fractional delay ( $\tau'$ ) in discrete-time domain will be less than unity.

The total network induced fractional delay is the combination of sensor to controller fractional delay ( $\tau'_{sc}$ ) and controller to actuator fractional delay ( $\tau'_{ca}$ ) which is given as,

$$\tau' = \tau'_{sc} + \tau'_{ca},\tag{3.6}$$

where,  $\tau'_{sc} = \frac{\tau_{sc}}{h}$  and  $\tau'_{ca} = \frac{\tau_{ca}}{h}$ . Assumption – 2: The disturbance d(k) is bounded by upper and lower bound as:

$$d_l \le d(k) \le d_u,\tag{3.7}$$

where,  $d_l$  and  $d_u$  denotes lower and upper bound of disturbance.

**Remark** – 3: In this work, the sensor processing delay  $\tau_{sp}$ , controller computational delay  $\tau_{cp}$  and actuator processing delay  $\tau_{ap}$  are neglected as their values are  $(\frac{1}{4})^{th}$ times lesser than network induced delay. Problem Statement: To design robust discrete-time sliding mode controller for the system (3.3,3.4) in the presence of deterministic fractional network delays  $\tau'_{sc}$  and  $\tau'_{ca}$  under the assumptions (1) and (2).

The sliding mode controller design involves the sliding surface design and the control law that computes the control sequences and steer the states towards the surface. The next segment proposes the design of sliding surface that compensates the effect of fractional delay occuring from sensor to controller.

## 3.4 Sliding Surface Design for Deterministic Network Delays

Tustin approximation and Bilinear transformation are two widely used approaches for approximation of time delay in discrete-time domain. The main limitation of these two approaches are that fractional delay cannot be approximated. While, Thiran approximation [Thiran (1971)] technique approximates the non-integer types of delays in discrete-time domain. Thiran has proposed the time delay approximation algorithm for maximally flat group of fractional delays occuring in signal processing applications. Hence it is proper candidate for fractional delay compensation for discrete-time SMC design.

The fractional delay in discrete-time can be approximated by Thiran's approximation as:

$$z^{-\nu} = \sum_{k=0}^{l} (-1)^{k} {l \choose k} \prod_{i=0}^{l} \frac{2\tau'_{sc} + i}{2\tau'_{sc} + k + i} z^{-k},$$
(3.8)

where, l indicates the order of approximation and  $\nu = \frac{\delta}{h}$  indicates the fractional part of delay,  $\delta$  is the delay occuring during signal transmission and h is the sampling interval. The order of approximation is given by:

$$l = ceil(\nu), \tag{3.9}$$

where, *ceil* operator rounds the nearest positive integer greater than or equal to  $\nu$ .

Next, the sliding surface using above approximation is proposed as Lemma -1

below.

Lemma – 1: The compensated sliding variable s(k) for the given system (3.3,3.4) with sensor to controller network fractional delay ( $\tau'_{sc}$ ) satisfying condition (3.5) under the assumptions (1) and (2) is given as:

$$s(k) = C_s x(k) - \alpha C_s (x(k-1)), \qquad (3.10)$$

where,

 $\alpha = \frac{\tau_{sc}'}{\tau_{sc}'+1}$  and  $C_s$  is the sliding gain.

**Proof**: The sliding variable with the delayed state vector at the receiving end of controller is given by:

$$s(k) = C_s x(k - \tau'_{sc}),$$
 (3.11)

where,  $\tau'_{sc}$  is the sensor to controller fractional delay. The sliding gain  $C_s$  is calculated using discrete LQR method through proper selection of Q and R matrices.

Applying z-Transform to Eqn. (3.11) we get,

$$s(z) = C_s x(z) z^{-\tau'_{sc}},$$
 (3.12)

where,  $\tau'_{sc} = \nu = \frac{\tau_{sc}}{h}$ .

It is assumed that total fractional delay ( $\tau'$ ) is less than unity. Considering  $\tau'_{sc} \prec 1$  and using Eqn. (3.9),  $z^{-\tau'_{sc}}$  can be approximated as,

$$z^{-\tau'_{sc}} = \sum_{k=0}^{1} (-1)^k \binom{l}{k} \Pi_{i=0}^1 \frac{2\tau'_{sc} + i}{2\tau'_{sc} + k + i} z^{-k}.$$
(3.13)

The above Eqn. (3.13) can be further expanded as,

$$z^{-\tau'_{sc}} = \left[ (-1)^0 \begin{pmatrix} 1\\0 \end{pmatrix} \left\{ \frac{2\tau'_{sc}}{2\tau'_{sc}} * \frac{2\tau'_{sc} + 1}{2\tau'_{sc} + 1} \right\} z^0 + (-1)^1 \begin{pmatrix} 1\\1 \end{pmatrix}$$

$$\left\{ \frac{2\tau'_{sc}}{2\tau'_{sc} + 1} * \frac{2\tau'_{sc} + 1}{2\tau'_{sc} + 2} \right\} z^{-1} \right].$$
(3.14)

On simplifying we get,

$$z^{-\tau'_{sc}} = 1 - \alpha z^{-1}, \tag{3.15}$$

where,  $\alpha = \frac{\tau'_{sc}}{\tau'_{sc}+1}$ .

Figures (3.2) and (3.3) shows the step response of delayed and non-delayed system for  $\tau'_{sc}=1$  and  $\tau'_{sc}=0.5$  respectively. It can be noticed that Thiran's Approximation approximates the fractional delay accurately and at each sampling instants the effect of fractional delay is nullified. It can be further extended that the step response of delay system is same as that of the non-delayed system that is computed through Thiran Approximation.



Figure 3.2: Step reponse of Thiran Approximation with  $\tau_{sc}^{\prime}{=}1$ 



Figure 3.3: Step reponse of Thiran Approximation with  $\tau'_{sc}$ =0.5

Thus, substituting Eqn. (3.15) into (3.12),

$$s(z) = C_s x(z) [1 - \alpha z^{-1}].$$
(3.16)

Further expanding it, we may get

$$s(z) = C_s x(z) - \alpha C_s z^{-1} x(z).$$
(3.17)

Applying inverse *z*-Transform, we may have

$$s(k) = C_s x(k) - \alpha C_s s x(k-1).$$
(3.18)

This completes the proof.

From Eqn. (3.18), it is inferred that the network induced fractional delay from sensor to controller can be compensated in the sliding surface s(k) at each sampling instant h using the information of immediate past state, the current state and parameter  $\alpha$ .

Now, we are ready to design a sliding mode control law using the proposed sliding surface (3.18).

# 3.5 Discrete-Time Sliding Mode Control (Switching type) for NCS Using Thiran's Delay Approximation

In this section, switching type control law alongwith its stability is proposed based on Gao's reaching law [Gao *et al.* (1995)] using compensated sliding surface (3.18). The Gao's reaching law provides the faster convergence within the specified quasi-sliding mode band.

Theorem – 1: The discrete-time sliding mode controller for system (3.3, 3.4) in the presence of deterministic fractional delays satisfying (3.5) and matched uncertainty d(k) is given as,

$$u(k) = -(C_s G)^{-1}[Mx(k) - Nx(k) - (1 - qh)(s(k)) + \epsilon h sgn(s(k))] - d(k).$$
(3.19)

where,

 $M = (C_s F)$  and  $N = \alpha C_s$ .

Proof: Let us consider the reaching law in [Gao et al. (1995)] with sensor to con-

troller fractional delay as:

$$s[(k+1)] = (1-qh)s(k) - \epsilon h sgn(s(k)), \qquad (3.20)$$

where,

 $q, \epsilon \succ 0, 0 \prec (1 - qh) \prec 1, sgn$  is the signum function, h represents the sampling interval and s(k) is the compensated sliding surface (3.18).

The reaching law in Eqn. (3.20) states that the state vector always move towards the quasi sliding mode band given as:

$$s(k) \le \frac{\epsilon h}{2 - qh}.\tag{3.21}$$

Substituting Eqn. (3.18) in Eqn. (3.20), we may get:

$$C_s x(k+1) - \alpha C_s x(k) = (1 - qh)s(k) - \epsilon h sgn(s(k)).$$

Substituting the value of x(k+1), we get

$$C_s[Fx(k) + G(u(k) + d(k))] - \alpha C_s x(k) = (1 - qh)s(k) - \epsilon h sgn(s(k)).$$
(3.22)

Further, simplifying we may write above Eqn. (3.22) as,

$$C_s[Fx(k) + G(u(k) + d(k))] - \alpha C_s x(k) = (1 - qh)s(k) - \epsilon h sgn(s(k)).$$
(3.23)

Further, above Eqn. (3.23) can be expressed as a control law

$$u(k) = -(C_s G)^{-1}[Mx(k) - Nx(k) + (1 - qh)(s(k)) - \epsilon h sgn(s(k))] - d(k).$$
(3.24)

where,

 $M = (C_s F)$  and  $N = \alpha C_s$ .

This completes the proof.

The stability condition is derived further using compensated sliding surface (3.18) and control law proposed in Eqn. (3.24) such that the system states remain within

specified band (3.21) for a finite interval of time.

#### 3.5.1 Stability Analysis

The state trajectories of the closed loop system (3.3,3.4) with network delay  $(\tau')$  and matched uncertainty d(k) with the controller (3.24) drive towards the sliding surface (3.18) and maintains on it for any  $q, \epsilon, \beta \succ 0, 0 \prec 1 - qh \prec 1$  and  $1 - qh \prec \epsilon$  provided the following condition hold true:

$$0 \leq \Phi \prec s^{T}(k)s(k). \tag{3.25}$$

**Proof**: The compensated sliding surface (3.18) is given by:

$$s(k) = C_s x(k) - \alpha C_s x(k-1).$$
(3.26)

Selecting the Lyapunov function as:

$$V_s(k) = s^T(k)s(k).$$
 (3.27)

Writing forward difference of the above Eqn. (3.27),

$$\Delta V_s(k) = s^T(k+1)s(k+1) - s^T(k)s(k).$$
(3.28)

Substituting the value of s(k + 1) using Eqn. (3.26) we get,

$$\Delta V_s(k) = [C_s x(k+1) - \alpha C_s x(k)]^T [C_s x(k+1) - \alpha C_s x(k)] - s^T(k) s(k).$$
(3.29)

Substituting the value of x(k+1),

$$\Delta V_s(k) = [C_s[Fx(k) + G(u(k) + d(k))] - \alpha C_s x(k)]^T$$

$$[C_sFx(k) + G(u(k) + d(k))] - \alpha C_s x(k)] - s^T(k)s(k).$$
(3.30)

Substituting the value of u(k) from Eqn. (3.24) and further solving it we have,

$$\Delta V_s(k) = [(1 - qh)s(k) - \epsilon h sgn(s(k))]^T * [(1 - qh)$$

$$s(k) - \epsilon h sgn(s(k))] - s^T(k)s(k).$$
(3.31)

Denoting,

$$\Phi = [(1-qh)s(k) - \epsilon hsgn(s(k))]^T * [(1-qh)s(k) - \epsilon hsgn(s(k))]$$

Then we have,

$$\Delta V_s(k) = \Phi - s^T(k)s(k). \tag{3.32}$$

The term  $\Phi$  can be tuned close to zero by appropriately selecting the parameter qand  $\epsilon$ . If  $\Phi$  is closed to zero, then  $s^T(k)s(k)$  will be larger than  $\Phi$ . Thus, for any small parameter  $\beta$ , we have  $\Phi - s^T(k)s(k) \prec \beta s^T(k)s(k)$ .

Thus, by tuning the parameter q and  $\epsilon$ , we have,  $\Delta V_s(k) \prec \beta s^T(k)s(k)$  which guarantees the convergence of  $\Delta V_s(k)$  and implies that any trajectory of the system (3.3,3.4) will be driven onto the sliding surface and maintain on it. This completes the **proof**.

The control signal u(k) computed in (3.24) using compensated sliding surface (3.10), will also experience controller to actuator fractional delay ( $\tau'_{ca}$ ) which results in the delayed control signal  $u(k - \tau'_{ca})$ . So, in order to avoid the degradation of the plant response again the time delay is compensated from controller to actuator. The compensated control signal at the actuator end can be represented as:

$$u_a(k) = u(k) - \alpha' u(k-1), \tag{3.33}$$

where,

$$\alpha' = \frac{\tau'_{ca}}{1 + \tau'_{ca}}.$$

It can be noticed from above Eqn. (3.33) that the compensated control signal  $u_a(k)$  de-

pends on difference of the present control signal that is available from network as well as past control signal which is multiplied over the parameter  $\alpha'$  approximated through Thiran Approximation. Thus the effect of controller to actuator fractional delay is compensated at actuator side which is further applied to the plant.

### **3.6 Results and Discussions**

This section briefly discusses about the simulation results as well as experimental results of the proposed control algorithm in the presence of deterministic network delays and matched uncertainty. The efficiency and robustness of the proposed control algorithm is tested considering DC Servo Motor as real time plant. On real time implementation the effects of control algorithm are examined in the presence of network non-idealities (fractional time delay and matched uncertainty).

#### **3.6.1** System Description

In this section, Quanser Qnet 2.0 brushed DC motor setup is explained in detailed on which the simulation as well as experimental results are carried out using proposed control law. The performance of the Brushless DC Motor is tested under different networked delays as well external disturbances to prove the robustness of the proposed algorithm. The results obtained with proposed algorithm is compared with the conventional SMC algorithm.



Figure 3.4: Block Diagram of Qnet DC Servo Motor Components

Figure (3.4) shows the block diagram of Quanser made Qnet 2.0 brushed DC motor setup used for the simulation as well as experimental purpose. The QNET DC Motor provides an integrated amplifier and a communication interface with the NI ELVIS II (+) for the amplifier command and encoder port. The NI ELVIS II (+) is interfaced to the PC via USB link to the QNET DC Motor setup as shown in Figure (3.5). The


Figure 3.5: Experimental Setup of Quanser DC Servo Motor

NI ELVIS II (+) block reads the angular encoder as an input and commands the power amplifier which acts as driver for the motor. The various network delays are generated through software blocks.

The detailed mathematical model along with the system parameters of the DC motor are given as [Astrom *et al.* (2015)]:

$$\frac{\theta(s)}{V_m(s)} = \frac{K_m}{J_m R_m s^2 + K_m^2 s},$$
(3.34)

where,

 $\theta(s)$ =output from the system (position),

 $V_m$ =input to the system,

 $J_m$ =Rotor inertia= $4 * 10^{-6} kgm^2$ ,

 $R_m$ =Terminal Resistance=8.4 ohms,

 $K_m$ =Motor back emf constant=0.042 V/(rad/s).

Substituting these parameters, the state space model of the above system (3.34) is given as,

$$\dot{x}(t) = Ax(t) + Bu(t - \tau) + Dd(t),$$
(3.35)

$$y(t) = Cx(t), \tag{3.36}$$

where,

$$A = \begin{bmatrix} -201 & 0\\ 1 & 0 \end{bmatrix}, B = \begin{bmatrix} 1\\ 0 \end{bmatrix},$$

$$C = \begin{bmatrix} 0 & 1 \end{bmatrix}, D = \begin{bmatrix} 1 \\ 1 \end{bmatrix}.$$
  
Discretizing the system at sampling interval  $h = 30msec$ , we get,

$$x(k+1) = Fx(k) + Gu(k - \tau') + d(k), \qquad (3.37)$$

$$y(k) = Cx(k), (3.38)$$

where,

$$F = \begin{bmatrix} 0.001836 & 0 \\ 0.004753 & 1 \end{bmatrix}, G = \begin{bmatrix} 0.004753 \\ -0.0001242 \end{bmatrix}$$
$$C = \begin{bmatrix} 0 & 1 \end{bmatrix}.$$

#### 3.6.2 Simulation and Experimental Results of Brushless DC Motor

In this section, the simulation and experimental results of Position Control DC motor are thoroughly discussed in presence of deterministic network delays. Figures (3.6) to (3.10) shows the simulation and experimental responses of system for trajectory tracking, compensated sliding variable and control efforts under different network delays. To show the robustness properties slow time varying disturbance signal is applied through the input channel. The total networked induced delay with a range of 12.8*msec* to 28msec is generated through network block for which the effect of delay is compensated satisfying condition (3.5). The position of DC motor is considered as the reference input. The sliding gain is computed using discrete LQR method which comes out  $C_s = [2.5156 \quad 31.6288]$  with Q = diag(1000, 1000) and R = 1. While the quasisliding band (3.21) comes out to be  $|s(k)| \leq -5$  to 5 with tunning parameters q = 30and  $\epsilon = 2000$ .

Figures (3.6(a)) to (3.7(d)) shows the simulation and experimental results of position control DC motor plant for total network delay of  $\tau = 12.8msec$  with  $\tau_{sc} = 6.4msec$ and  $\tau_{ca} = 6.4msec$ . The fractional part of total network delay is obtained as  $\tau' = 0.426$ ,  $\tau'_{sc} = 0.213$  and  $\tau'_{ca} = 0.213$  for h = 30msec. The trajectory response of the system in case of simulation and experimental are shown in Figures (3.6(a)) and (3.6(b)) respectively. In both cases, the output tracks the reference trajectory in the presence of specified network delay. In order to show the exact effect of time delay compensation



(a) Simulated response of Position control for  $\tau$ =12.8msec



(c) Magnified simulated response of Position control for  $\tau$ =12.8msec



(e) Simulated compensated sliding variable for  $\tau$ =12.8msec



(g) Magnified simulated response of s(k) for  $\tau$ =12.8msec



(b) Experimental response of Position control for  $\tau$ =12.8msec



(d) Magnified experimental response of Position control for  $\tau$ =12.8msec



(f) Experimental compensated sliding variable for  $\tau$ =12.8msec



(h) Magnified experimental response of s(k) for  $\tau$ =12.8msec



10





(a) Simulated control efforts for  $\tau$ =12.8msec



(c) Magnified simulated control efforts  $\tau$ =12.8msec



(e) Simulated response of Position control for  $\tau=24$ msec



(g) Magnified simulated response of Position control for  $\tau$ =24msec



(b) Experimetal control efforts for  $\tau$ =12.8msec



(d) Magnified experimental control efforts for  $\tau$ =12.8msec



(f) Experimental response of Position control for  $\tau$ =24msec



(h) Magnified experimental response of Position control for  $\tau$ =24msec

Figure 3.7: Simulation as well as Experimental results of control signal and tracking response for  $\tau$ =12.8msec and  $\tau$ =24msec.

for



(a) Simulated compensated sliding variable for  $\tau$ =24msec



(c) Magnified simulated response of s(k) for  $\tau=24$ msec



(e) Simulated control efforts for  $\tau$ =24msec



 $\tau$ =24msec



(b) Experimental compensated sliding variable for  $\tau$ =24msec



(d) Magnified experimental response of s(k) for  $\tau=24\mathrm{msec}$ 



(f) Experimental control efforts for  $\tau$ =24msec



control efforts for (h) Magnified experimental control efforts for  $\tau$ =24msec

Figure 3.8: Simulation as well as Experimental results of compensated sliding surface and control signal for  $\tau$ =24msec.



Position Control (degrees) 400 200 0 -200 -400 50 90 100 60 70 0 10 20 30 40 80 Time (sec)

(a) Simulated response of Position control for  $\tau$ =28msec



(b) Experimental response of Position control for  $\tau$ =28msec



(c) Magnified simulated response of Position control for  $\tau$ =28msec



(e) Simulated compensated sliding variable for  $\tau$ =28msec



(g) Magnified simulated response of s(k) for  $\tau$ =28msec

(d) Magnified experimental response of Position control for  $\tau$ =28msec



(f) Experimental compensated sliding variable for  $\tau$ =28msec



(h) Magnified experimental response of s(k) for  $\tau$ =28msec

Figure 3.9: Simulation as well as Experimental results of tracking and compensated sliding surface for  $\tau$ =28msec.



Figure 3.10: Simulation as well as Experimental results of control efforts for  $\tau$ =28msec along with tracking response under external disturbances.

at the output, results are magnified as shown in Figures (3.6(c)) and (3.6(d)). It can be noticed that the effect of fractional time delay from sensor to controller is compensated as the output tracks the trajectory at 6.4msec. The same effect of time delay compensation from sensor to controller can be observed in sliding surface Figures (3.6(e)) and (3.6(f)) as well as control signal Figures (3.7(a)) and (3.7(b)). Observing the magnified results of Figures (3.6(g)), (3.6(h)), (3.7(c)) and (3.7(d)) it can be noticed that the sliding surface and control signal both are computed at first sampling instant even in the presence of sensor to controller delay. Thus, the effects of fractional delay from sensor to controller at sliding surface and control signal are compensated and remains within the specified sliding band (3.21). The proposed algorithm was further extended for higher values of  $\tau$ . Figures (3.7(e)) to (3.8(h)) shows the simulation and experimental results of position control DC motor for total networked delay of  $\tau = 24msec$  with  $\tau_{sc} = 12msec$ and  $\tau_{ca} = 12msec$ . The fractional part of total network delay is computed as  $\tau' = 0.8$ ,  $au_{sc}^{\prime}=0.4$  and  $au_{ca}^{\prime}=0.4$  for h=30msec. The simulated and experimental trajectory response of the system are shown in Figures (3.7(e)) and (3.7(f)) respectively. Observing the results it can be noticed the output tracks the reference signal in the specified networked delay. In order to show the effect of delay compensation the output results are magnified shown in Figures (3.7(g)) and (3.7(h)) respectively. It can be noticed that the effect of fractional delay from sensor to controller is nullified as the output tracks the reference trajectory at t = 12msec. The similar effect of time delay compensation can be observed in sliding surface as well as control efforts signal shown in Figures (3.8(a))

to (3.8(h)). Observing the simulated and experimental magnified results of sliding surface [(3.8(c)) and (3.8(d))] as well as control signal [(3.8(g)) and (3.8(h))], it can be noticed that, in both the cases the sliding surface and control signal are computed at first sampling instant. Thus the fractional delay from sensor to controller is compensated and remains within the specified sliding band (3.21). Figures (3.9(a)) to (3.10(c))shows the simulation and experimental results of position control DC motor for total networked delay of  $\tau = 28msec$  with  $\tau_{sc} = 14msec$  and  $\tau_{ca} = 14msec$ . The fractional part of total networked delay for h = 30 msec is obtained as  $\tau' = 0.933$ ,  $\tau'_{sc} = 0.466$ and  $\tau_{ca}^{\prime}=0.466$  respectively. The simulation and experimental results with magnified response of reference trajectory are shown in Figures (3.9(a)) to (3.9(d)) respectively. Observing the results it can be concluded that the output tracks the reference trajectory at t = 14msec for the specified networked delay. Thus the effect of fractional delay from sensor to controller is nullified at the output as shown in Figures (3.9(c))and (3.9(d)). The similar effect of time delay compensation will be observed in sliding surface and control signal results shown in Figures (3.9(e)) to (3.10(c)). Observing the results it can be noticed that in simulation as well as experimental case the sliding surface and the control signal are computed from first sampling instant. Thus the effect of fractional delay from sensor to controller is compensated at sliding surface as well as at control signal. Apart from delay compensation, the position of motor was also controlled by applying the external disturbances through rotating the wheel in forward and reverse direction. The situation of motor under external disturbances is shown in Figure (3.10(d)).

Thus, from all the results it can be concluded that the proposed algorithm works efficiently with network delay range of  $12.8msec \le \tau \le 28msec$  in experimental as well as in simulated environment. The proposed controller compensates the network time delay for q = 30 and  $\epsilon = 2000$  satisfying (3.5) and shows the stable response satisfying condition (3.25) in the presence of matched uncertainty.

#### Comparison of proposed algorithm with conventional sliding mode control

In this section, the experimental results of proposed algorithm are compared with conventional sliding mode control. The sliding surface and control algorithm for conven-



(a) Position control of DC motor using proposed algorithm for  $\tau$ =12.8msec



(c) Compensated sliding variable of proposed algorithm for  $\tau$ =12.8msec



(b) Position control of DC motor using conventional SMC algorithm for  $\tau$ =12.8msec



(d) Compensated sliding variable of conventional SMC algorithm for  $\tau$ =12.8msec



(e) Control efforts response of proposed algorithm for  $\tau$ =12.8msec

(f) Control efforts response of conventional SMC algorithm for  $\tau$ =12.8msec

Figure 3.11: Comparison of proposed algorithm and conventional sliding mode control for  $\tau$ =12.8msec

tional sliding mode control is defined as:

$$s_c(k) = C_s x(k - \tau'_{sc}),$$

$$u(k) = -(C_s G)^{-1} [C_s x(k - \tau'_{sc}) - (1 - qh)(s_c(k)) + \epsilon h sgn(s_c(k))] - d(k - \tau'_{sc}).$$

The results of tracking response, control signal and sliding variable for total networked delay of  $\tau = 12.8msec$  are shown in Figures (3.11(a)) to (3.11(f)). From the comparative results it can be noticed that the conventional sliding mode control becomes unstable for a small delay of  $\tau_{sc} = 6.4msec$ . Thus, the Thiran's approximation proved to be efficient method in discrete-time sliding mode control. Simulation and experimental results for different networked delays range are summarized in Table-3.1. The

Network Delays $(\tau)$	Simulations or Experimental			
	Chattering	Design performance		
12.8msec	within QSMB	satisfactory		
24msec	within QSMB	satisfactory		
28msec	within QSMB	satisfactory		

Table 3.1: Simulations and Experimental Results Under different Networked Delays

Algorithm	Comparison Results					
	au	$T_s$	Chattering	Response		
Coventional SMC	12.8msec	Undefined	high	unstable		
Proposed method	12.8msec	1sec	within QSMB	stable		

Table 3.2: Comparison of Proposed Algorithm with Conventional SMC

comparison of discrete-time sliding mode control with time delay approximation and conventional sliding mode control are shown in Table- 3.2.

# 3.7 Conclusion

In this chapter, we explored Thiran's approximation technique for fractional delay compensation in discrete-time domain. The effect of fractional delay generated due to the communication medium is compensated in sliding surface. The sliding surface is designed in such a manner that the system states slides along the predetermined surface according to network delay. Using this approach, switching type discrete-time sliding mode controller is designed which computes the control actions in the presence of network delay and matched uncertainty. The stability of the closed loop NCS is assured by using Lyapunov approach. The effectiveness of the derived algorithm is tested on brushless DC motor setup with deterministic networked delay and matched uncertainty. The experimental results are compared with conventional SMC algorithm. The comparative results proved that the fractional delay approximated using Thiran Approximation is most efficient technique as it compensates the fractional network delays in discretetime domain in the presence of matched as well as unmatched uncertainties.

# **CHAPTER 4**

# DESIGN OF DISCRETE-TIME SLIDING MODE CONTROLLER (NON-SWITCHING TYPE) FOR FRACTIONAL DELAY

## 4.1 Introduction

In this chapter, a unique approach is presented for designing non-switching discretetime sliding mode controller using Thiran's delay approximation. The effect of sensor to controller delay is compensated using Thiran's delay approximation technique in sliding surface. Further, Lyapunov approach is used to determine the stability of closed loop NCSs with proposed controller. The feasibility and effectiveness of the control methodology are outlined through simulation and experimental results showing the siginificant response in the presence of networked delay. The efficacy of the proposed control algorithm is further validated in the presence of real time networks such as CAN and Switched Ethernet using True Time Simulator.

# 4.2 **Problem Formulation**

**Remark** – 4: In this chapter nature of the system, total fractional network delays ( $\tau'$ ), sensor to controller fractional delay ( $\tau'_{sc}$ ) and controller to actuator fractional delay ( $\tau'_{ca}$ ) would remain same as mentioned in section (3.3).

Problem Statement: The main objective, is to design non-switching based discretetime sliding mode control law for system (3.3,3.4) in the presence of fractional delay  $\tau'_{sc}$  and  $\tau'_{ca}$  satisfying (3.5) and matched uncertainty satisfying (3.7). The next section describes the design of non-switching type sliding mode control law using the proposed sliding surface (3.18).

# 4.3 Discrete-Time Sliding Mode Control (Non-Switching type) Using Thiran's Delay Approximation

In this section, non-switching type control law alongwith its stability is proposed based on reaching law in [Bartoszewicz and Lesniewski (2014)] using compensated sliding surface (3.18). The reaching law proposed by Bartoszewicz causes less chattering compare to Gao's law and offers faster covergence with limited magnitude of the control signal.

**Theorem** – 2: The non-switching discrete-time sliding mode controller for system (3.3, 3.4) in the presence of deterministic sensor to controller fractional delay satisfying (3.5) and matched uncertainty d(k) is given as,

$$u(k) = -(C_s G)^{-1} [Hx(k) - Ix(k) - J(s(k)) + d_s(k) - d_1] - d(k).$$
(4.1)

where,

$$H = (C_s F), I = \alpha C_s, J = \{1 - q[s(k)]\}.$$

**Proof**: Let us consider the reaching law in [Bartoszewicz and Lesniewski (2014)] in the presence of sensor to controller fractional delay given as:

$$s[(k+1)h] = \{1 - q[s(k)]\} - d_s(k) + d_1,$$
(4.2)

where,

where  $d_s(k)$  is disturbance at the controller end,  $q[s(k)] = \frac{\psi}{\psi + |s(k)|}$  with  $\psi$  as user defined constant satisfying  $\psi \ge d_2$ ,  $d_1$  and  $d_2$  are mean and deviation of d(k).

**Remark** – 5: The disturbace d(k) appearing in the reaching law is applied through the

network. So, the compensated disturbance using Thiran's approximation is given as:

$$d_s(k) = d(k) - \alpha d(k-1).$$
(4.3)

The reaching law in Eqn. (4.2) indicates that the system states always move towards the specified sliding band given as:

$$|s(k)| \le \frac{\psi d_2}{\psi - d_2}.\tag{4.4}$$

Substituting Eqn. (3.18) into Eqn. (4.2) we may get:

$$C_s x(k+1) - \alpha C_s x(k) = \{1 - q[s(k)]\} - d_s(k) + d_1,$$

and substituting the value of x(k+1) we get,

$$C_s[Fx(k) + G(u(k) + d(k))] - \alpha C_s x(k) = \{1 - q[s(k)]\} - d_s(k) + d_1.$$

Further simplifying we may write,

$$C_s Fx(k) + C_s G(u(k) + d(k)) - \alpha C_s x(k) = \{1 - q[s(k)]\} - d_s(k) + d_1.$$
(4.5)

Further, solving the above Eqn. (4.5), control law can be expressed as:

$$u(k) = -(C_s G)^{-1} [Hx(k) - Ix(k) - J(s(k)) + d_s(k) - d_1] - d(k).$$
(4.6)

where,

$$H = (C_s F), I = \alpha C_s \text{ and } J = \{1 - q[s(k)]\}$$
  
This completes the proof.

The stability condition is derived further using compensated sliding surface (3.18) and control law proposed in Eqn. (4.6) such that the system states remain within specified band (4.4) over a finite interval of time.

#### 4.3.1 Stability Analysis

For given positive scalars  $\tau'_{sc}$  and  $\tau'_{ca}$  with total networked delay  $\tau'$  satisfying (3.5), the trajectories of the closed loop system (3.3,3.4) with controller (4.6) and d(k) satisfying (3.7) drive towards the sliding surface (3.18) such that the following condition (4.7) is feasible:

$$0 \leq \kappa \prec s^T(k)s(k). \tag{4.7}$$

Proof: The compensated sliding surface is given by,

$$s(k) = C_s x(k) - \alpha C_s x(k-1).$$
(4.8)

Selecting the Lyapunov function as,

$$V_s(k) = s^T(k)s(k).$$
 (4.9)

Writing forward difference of the above equation,

$$\Delta V_s(k) = s^T(k+1)s(k+1) - s^T(k)s(k).$$
(4.10)

Substituting the value of s(k + 1) using Eqn. (4.8) we get,

$$\Delta V_s(k) = [C_s x(k+1) - \alpha C_s x(k)]^T [2C_s x(k+1) - \alpha C_s x(k)] - s^T(k)s(k).$$
(4.11)

Substituting the value of x(k+1),

$$\Delta V_s(k) = [C_s[Fx(k) + G(u(k) + d(k))] - \alpha C_s x(k)]^T$$

$$[C_sFx(k) + G(u(k) + d(k))] - \alpha C_s x(k)] - s^T(k)s(k).$$
(4.12)

Substituting the value of u(k) from Eqn. (4.6) and further solving it we have,

$$\Delta V_s(k) = [(1 - q[s(k)])s(k) - d_s(k) + d_1]^T * [(1 - q[s(k)])$$

$$s(k) - d_s(k) + d_1] - s^T(k)s(k).$$
(4.13)

Denoting,

$$\kappa = [(1 - q[s(k)])s(k) - d_s(k) + d_1]^T * [(1 - q[s(k)])s(k) - d_s(k) + d_1]$$

Then we have,

$$\Delta V_s(k) = \kappa - s^T(k)s(k). \tag{4.14}$$

The term  $\kappa$  can be tuned close to zero by appropriately selecting the parameter  $\psi$ . If  $\kappa$  is closed to zero, then  $s^T(k)s(k)$  will be larger than  $\kappa$ . Thus, for any small parameter  $\eta$ , we have  $\kappa - s^T(k)s(k) \prec \eta s^T(k)s(k)$ .

Thus, by tuning the parameter  $\psi$ , we have,  $\Delta V_s(k) \prec \eta s^T(k)s(k)$  which guarantees the convergence of  $\Delta V_s(k)$  and implies that any trajectory of the system (3.3,3.4) will be driven onto the sliding surface and maintain on it.

This completes the proof.

# 4.4 Results and Discussions

This section briefly discusses about the simulation results as well as experimental results of the proposed control algorithm in the presence of deterministic network delays and matched uncertainty. The efficiency and robustness of the proposed control algorithms are tested under three different situations: (i) illustrative example (ii) real time plant as DC servo motor and (iii) real time networks.

#### 4.4.1 Simulation Results with Illustrative Example

In this segment an illustrative example from [Wu and Chen (2007)] is simulated in MATLAB environment.

Consider the continuous-time LTI system as,

$$\dot{x}(t) = Ax(t) + Bu(t - \tau) + Dd(t),$$
(4.15)

$$y(t) = Cx(t), (4.16)$$

where,

$$A = \begin{bmatrix} -0.7 & 2\\ 0 & -1.5 \end{bmatrix}, B = \begin{bmatrix} -0.03\\ -1 \end{bmatrix},$$
$$C = \begin{bmatrix} 1 & 0 \end{bmatrix}, D = \begin{bmatrix} 1\\ 1 \end{bmatrix}.$$

Discretizing the above system parameters at sampling interval of h = 30 msec,

$$x(k+1) = Fx(k) + Gu(k - \tau') + d(k), \qquad (4.17)$$

$$y(k) = Cx(k), \tag{4.18}$$

where,  

$$F = \begin{bmatrix} 0.9792 & 0.05805 \\ 0 & 0.956 \end{bmatrix}, G = \begin{bmatrix} -0.001771 \\ -0.02934 \end{bmatrix},$$

$$C = \begin{bmatrix} 1 & 0 \end{bmatrix}.$$



Figure 4.1: Slow time varying disturbance d(k)

Figures (4.2) to (4.13) shows the nature of the system under networked environment. In order to check the robustness of the derived control law a slow time varying distur-



Figure 4.2: Total network delay  $\tau$ 



Figure 4.3: State variable  $x_1$  with initial condition  $x_1=1$ 



Figure 4.4: State variable  $x_2$  with initial condition  $x_2=1$ 



Figure 4.5: Magnified result of state variables  $x_1, x_2$ 

bance is applied at the input of the system as shown in Figure (4.1). Figure (4.2) shows the deterministic network induced fractional delay with range of  $3msec \le \tau \le 20msec$ 



Figure 4.6: Compensated sliding surface s(k)



Figure 4.7: Magnified result of compensated sliding surface s(k)





Figure 4.9: Compensated control signal  $u_a(k)$ 

under which the system shows the stable response satisfying (3.5). In this work, network delay is considered as the time required for the data packets to travel from sensor



Figure 4.10: Magnified compensated control signal  $u_a(k)$ 



Figure 4.11: Response of SNR



Figure 4.12: Nature of state variables for different SNR



Figure 4.13: Result of Stability

to controller and controller to actuator. The time required for data packets to travel from sensor to controller is  $1.5msec \le \tau_{sc} \le 10msec$  and for controller to actuator is  $1.5msec \leq \tau_{ca} \leq 10msec$  respectively. The sliding gain  $C_s$  is calculated using discrete LQR method with Q = diag(1000, 1000) and R = 1. The computed values of sliding gain are  $C_s = [-1.77 \quad -2.766]$ . The quasi-sliding mode band computed to be  $|s(k)| \leq +0.2$  to -0.2 with proper selection of user defined constant  $\psi = 100$ .

Figures (4.3) and (4.4) shows the plant state variables with initial condition  $x(k) = \begin{bmatrix} 1 & 1 \end{bmatrix}$ . Both the states converges to zero from given initial condition in the presence of network fractional delay. Figure (4.5) shows the magnified result of the plant state variables. It can be noticed that the effect of network fractional delay at the ouptut of the system is compensated as it is computed from first sampling instant. Figure (4.6) shows the compensated sliding surface calculated using Thiran approximation rule. It can be observed that the compensated sliding variable is computed from first sampling instant even in the presence of sensor to controller fractional delay. The magnified response of the same is shown in Figure (4.7). Figure (4.8) shows the control signal u(k) which is computed using proposed compensated sliding surface s(k). This control signal is further applied to the plant through the network. The same approach of time delay compensation is used to compute the compensated control signal  $u_a(k)$ . The result of the same is shown in Figure (4.9). From the magnified result in Figure (4.10), it can be justified that the effect of controller to actuator fractional delay is also compensated as the control signal is computed from first sampling interval.

The algorithm is also examined for different SNR as shown in Figure (4.11). It can be observed from Figure (4.12) that the system states coverge to zero for different SNR. Figure (4.13) shows the results of stability. It can be observed from Figure (4.13) that for given  $\psi = 100$  and  $d_2 = 0.2$  guarantees the covergence of  $\Delta V_s(k)$  and implies that the trajectories of system (3.3,3.4) will be driven on the compensated sliding surface and maintain on it under the specified network fractional delay and matched uncertainty. Thus, from above results it is justified that Thiran approximation provides better compensation in discrete-time domain in the presence of deterministic network delays and matched uncertainty.

#### 4.4.2 Simulation and Experimental Results of Brushless DC Motor

The state space model of the system (3.34) is given as,

$$\dot{x}(t) = Ax(t) + Bu(t - \tau) + Dd(t), \tag{4.19}$$

$$y(t) = Cx(t), (4.20)$$

where,

$$A = \begin{bmatrix} -201 & 0 \\ 1 & 0 \end{bmatrix}, B = \begin{bmatrix} 1 \\ 0 \end{bmatrix},$$
$$C = \begin{bmatrix} 0 & 1 \end{bmatrix}, D = \begin{bmatrix} 1 \\ 1 \end{bmatrix}.$$

Discretizing the system at sampling interval h = 30 msec, we get,

$$x(k+1) = Fx(k) + Gu(k-\tau') + d(k), \qquad (4.21)$$

$$y(k) = Cx(k), (4.22)$$

where,  

$$F = \begin{bmatrix} 0.001836 & 0 \\ 0.004753 & 1 \end{bmatrix}, G = \begin{bmatrix} 0.004753 \\ -0.0001242 \end{bmatrix},$$

$$C = \begin{bmatrix} 0 & 1 \end{bmatrix}.$$

This section briefs about the simulation and experimental results of Position control Brushless DC motor in the presence of various deterministic delays using nonswitching control law. The effect of time delay compensation are deeply analyzed through tracking response, compensated sliding variable and control signal for different network delays as shown in Figures (4.14) to (4.18). The robustness of the proposed algorithm is determined by applying time varying disturbance signal at the input side of the channel. The total networked induced delay with a range of 10msec to 28msecwas generated for which the effect of time delay is compensated satisfying condition (3.5). The sliding gain is computed through discrete LQR method which comes out to be  $C_s = [24.5156 \ 31.6288]$  with Q = diag(1000, 1000) and R = 1. The quasisliding mode band computed to be  $|s(k)| \le +0.2$  to -0.2 with proper selection of user defined constant  $\psi = 1500$ . Figures (4.14(a)) to (4.15(d)) shows the simulated and experimental results of position control brushless DC motor for total networked induced



(a) Simulated result of Position control for  $\tau$ =10msec



(c) Magnified simulated result of Position control for  $\tau$ =10msec



(e) Simulated result of compensated sliding variable for  $\tau$ =10msec



(g) Magnified simulated result of s(k) for  $\tau{=}10\mathrm{msec}$ 



(b) Experimental result of Position control for  $\tau$ =10msec



(d) Magnified experimental result of Position control for  $\tau$ =10msec



(f) Experimental result of compensated sliding variable for  $\tau$ =10msec



(h) Magnified experimental result of s(k) for  $\tau$ =10msec



ding Variab



(c) Magnified simulated control signal for  $\tau$ =10msec



(e) Simulated result of Position control for  $\tau$ =18msec



(g) Magnified simulated result of Position control for  $\tau$ =18msec



(b) Experimetal control signal for  $\tau$ =10msec



(d) Magnified experimental control signal for  $\tau$ =10msec



(f) Experimental result of Position control for  $\tau$ =18msec



(h) Magnified experimental result of Position control for  $\tau$ =18msec

Figure 4.15: Simulation as well as Experimental results of control signal and tracking response for  $\tau$ =18msec and  $\tau$ =24msec



Sliding Variable 0.02 С -0.02 0 10 20 30 40 50 60 70 80 90 100 Time (sec)

(a) Simulated result of compensated sliding variable for  $\tau$ =18msec



(c) Magnified simulated result of s(k) for  $\tau$ =18msec



(g) Magnified simulated control signal for  $\tau$ =18msec

(b) Experimental result of compensated sliding variable for  $\tau$ =18msec



(d) Magnified experimental result of s(k) for  $\tau$ =18msec



(f) Experimental control signal for  $\tau$ =18msec



(h) Magnified experimental control signal for  $\tau$ =18msec

Figure 4.16: Simulation as well as Experimental results of compensated sliding surface and control signal for  $\tau$ =18msec



(a) Simulated result of Position control for  $\tau=28$ msec



(c) Magnified simulated result of Position control for  $\tau$ =28msec



(e) Simulated result of compensated sliding variable for  $\tau$ =28msec



(g) Magnified simulated result of s(k) for  $\tau{=}28\mathrm{msec}$ 



(b) Experimental result of Position control for  $\tau$ =28msec



(d) Magnified experimental result of Position control for  $\tau$ =28msec



(f) Experimental result of compensated sliding variable for  $\tau$ =28msec



(h) Magnified experimental result of s(k) for  $\tau$ =28msec

Figure 4.17: Simulation as well as Experimental results of tracking and compensated sliding surface for  $\tau$ =28msec

sliding Vari

68



Figure 4.18: Simulation as well as Experimental results of control signal for  $\tau$ =28msec along with tracking response

delay of  $\tau = 10msec$  with  $\tau_{sc} = 5msec$  and  $\tau_{ca} = 5msec$ . The fractional part of total networked delay for sampling interval of h = 30 msec is computed to be  $\tau' = 0.33$ ,  $\tau'_{sc} = 0.166$  and  $\tau'_{ca} = 0.166$  respectively. Figures (4.14(a)) and (4.14(b)) shows the simulated as well as experimental trajectory results of the plant. It can be observed that the position of DC motor is controlled according to variations in the reference inputs without chattering even in the presence of specified network delay. The tracking results are magnified as shown in Figures (4.14(c)) and (4.14(d)) in order to examine the effect of time delay compensation. It can be noticed that in both the cases the fractional part of the delay from sensor to controller is compensated as the position of the motor commence the reference input signal at 5msec. The same consequence of time delay compensation is observed in sliding variable (4.14(e)) and (4.14(f)) as well as control signal (4.15(a)) and (4.15(b)) respectively. Observing the magnified results (4.14(g)), (4.14(h)), (4.15(c)) and (4.15(d)) it can be noticed that the compensated sliding variable and control signal both are computed from first sampling instants. Thus, the effect of fractional delay from sensor to controller is compensated at the sliding surface and control signal. The proposed algorithm was further examined for  $\tau = 18msec$  and 28msec respectively. Figures (4.15(e)) to (4.15(h)) shows the results of position control brushless DC motor. The results are carried out under the total networked delay of  $\tau = 18msec$  with  $\tau_{sc} = 9msec$  and  $\tau_{ca} = 9msec$ . The fractional part of delay for h=30msec is computed as  $\tau'=0.6,\,\tau'_{sc}=0.3$  and  $\tau'_{ca}=0.3$  respectively. The simulated and experimental tracking results of the plant for the specified networked delay are shown in Figures (4.15(e)) and (4.15(f)). It can be noticed that in both the cases the position of DC motor is controlled for all given reference inputs. In order to examine the effect of fractional time delay compensation the same results are magnified in Figures (4.15(g)) and (4.15(h)). It can be noticed that the effect of fractional delay from sensor to controller is nullified as the output follows the reference signal at t = 9msec. The same effect of time delay compensation is observed in sliding variable (4.16(a)) and (4.16(b)) as well as control signal (4.16(e)) and (4.16(f))). Observing the magnified results [(4.16(c)), (4.16(d)), (4.16(g)) and (4.16(h))] it can be noticed that in both the cases the sliding variable and control signal are computed from first sampling instants. Thus the effect of fractional part of delay from sensor to controller is compensated at sliding variable and control signal. Figures (4.17(a)) to (4.18(d)) shows the results of position control of brushless DC motor for total networked delay of  $\tau = 28msec$  with  $\tau_{sc} = 14msec$  and  $\tau_{ca} = 14msec$ . Considering the sampling interval of h = 30msec the fractional part of delay is computed as  $\tau' = 0.933$ ,  $\tau'_{sc} = 0.466$  and  $\tau'_{ca} = 0.46$  respectively. Figures (4.17(a)) and (4.17(b)) shows the simulated and experimental tracking results of the system for the specified networked delay. It can be observed that in both the cases, output tracks the reference trajectory without chattering. In order to examine the actual effect of time delay compensation the results are magnified as shown in Figures (4.17(c)) and (4.17(d)). Observing the magnified results it can be concluded that the effect of fractional part of delay from sensor to controller is compensated as output tracks the reference signal at t = 14msec. Figures (4.17(e)), (4.17(f)), (4.18(a)) and (4.18(b)) shows the simulated and experimental results of compensated sliding variable and control signal for specified networked delay. It can be noticed from the results that the time delay compensation algorithm works efficiently for large value of  $\tau$ . The magnified results of the same are shown in Figures (4.17(g)), (4.17(h)), (4.18(c)) and (4.18(d)) which clearly justifies that in both the cases the effect of fractional part of delay from sensor to controller is compensated in the sliding variable as well as control signal.

Thus from above results, it can be concluded that the non-switching controller designed using proposed algorithm works efficiently with the network delay range of 10msec to 28msec in simulated as well as experimental environment. The proposed controller



Time (sec) (a) Position control of DC motor using proposed al- (b) Po gorithm for  $\tau$ =12.8msec ba



(c) Position control of DC motor using conventional SMC algorithm for  $\tau$ =12.8msec



(e) Compensated sliding surface of switching based SMC algorithm for  $\tau$ =12.8msec



(g) Control signal response of proposed algorithm for  $\tau$ =12.8msec



(b) Position control of DC motor using switching based SMC algorithm for  $\tau$ =12.8msec



(d) Compensated sliding variable of proposed algorithm algorithm for  $\tau$ =12.8msec



(f) Compensated sliding variable of conventional SMC algorithm for  $\tau$ =12.8msec



(h) Control signal response of switching based SMC algorithm for  $\tau$ =12.8msec

Figure 4.19: Comparison of proposed algorithm, switching based sliding mode control and conventional sliding mode control for  $\tau$ =12.8msec

compensates the effect of fractional time delay without chattering for  $\psi = 1500$  satisfying (4.4) and shows the stable response satisfying (4.7) in the presence of matched uncertainty.

#### Comparison of proposed algorithm with conventional sliding mode control

In this section, the experimental results of proposed algorithm are compared with switching sliding mode control using time delay approximation and conventional sliding mode control. The results are compared in terms of tracking response, sliding variable and



Figure 4.20: Control signal response of coventional SMC algorithm for  $\tau$ =12.8msec

Algorithm	Comparison Results				
	au	$T_s$	Chattering	Response	
Coventional SMC	12.8msec	Undefined	high	unstable	
Switching SMC	12.8msec	1sec	within QSMB	stable	
Proposed method	12.8msec	0.3sec	negligible within QSMB	stable	

Table 4.1: Comparison of Proposed Algorithm, Switching based SMC and Conventional SMC

control signal for total networked delay of  $\tau = 10msec$ . From the comparative results [Figures (4.19(a) to (4.20)], it can be observed that the conventional sliding mode control becomes unstable for a small delay of  $\tau_{sc} = 5msec$  while the sliding mode controller designed using switching algorithm generates the chattering behaviour at the output signal compared to proposed algorithm. Thus, Thiran Approximation proved to be more efficient technique for non-switching based discrete-time sliding mode control. The comparison of proposed algorithm with switching sliding mode control using time delay approximation and conventional sliding mode control are summarized in Table-4.1.

### 4.4.3 Simulation Results With Real-Time Networks

In previous section the efficacy of the proposed control law is examined in the presence of Brushless DC motor connected through networked medium. It can be observed from simulation results that, the control law proposed using non-switching reaching law provides faster convergence without increasing the amplitude of control signal. The chattering is also negligible compared to switching-type control law. Thus, in this section the efficacy of the proposed non-switching control law is further tested in the presence of real time networks and matched uncertainty. The real time networks are simulated using True Time software which provides wide range of simulated networks such as CAN, Switched Ethernet, Profibus, Profinet, CSMA/CD, Round Robbin etc.. In this work, the simulations are carried out under CAN and Switched Ethernet communication medium as network delays are assumed to be deterministic in nature. Further, the performance of the system is also studied in the presence of packet loss situation.

The following network specifications and parameters are considered for simulations:

Networked Medium : CAN and Switched Ethernet Data Rate (bits/s)=80000 Minimum frame size (bits)=512 (CAN) and 1024 (Switched Ethernet) Loss Probability = 0 to 0.5 sampling interval h= 0.030 second.  $C_s = [24.5156 \quad 31.6288]$  $|s(k)| \leq +0.2$  to -0.2 with user defined constant  $\psi = 1500$ .

Consider the continuous-time LTI system as,

$$\dot{x}(t) = Ax(t) + Bu(t - \tau) + Dd(t),$$
(4.23)

$$y(t) = Cx(t),$$
 (4.24)

where,  $A = \begin{bmatrix} -0.7 & 2 \\ 0 & -1.5 \end{bmatrix}, B = \begin{bmatrix} -0.03 \\ -1 \end{bmatrix},$   $C = \begin{bmatrix} 1 & 0 \end{bmatrix}, D = \begin{bmatrix} 1 \\ 1 \end{bmatrix}.$ 

Discretizing the above system parameters at sampling interval of h = 30msec we get,

$$x(k+1) = Fx(k) + Gu(k-\tau') + d(k), \qquad (4.25)$$

$$y(k) = Cx(k), (4.26)$$

where,

$$F = \begin{bmatrix} 0.9792 & 0.05805\\ 0 & 0.956 \end{bmatrix}, G = \begin{bmatrix} -0.001771\\ -0.02934 \end{bmatrix},$$

$$C = \begin{bmatrix} 1 & 1 \end{bmatrix}.$$

#### CAN as a Network Medium











Figure 4.23: Tracking response of system with CAN network under ideal condition



Figure 4.24: Tracking response of system with CAN network under traffic conditions

In this section, the nature of the system with CAN as a networked medium is studied in Figures (4.21) to (4.36). The robustness of the proposed controller is checked



Figure 4.25: Magnified tracking response with CAN network under ideal conditions



Figure 4.26: Magnified tracking response with CAN network under traffic conditions



Figure 4.27: Compensated sliding variable s(k)



Figure 4.28: Magnified compensated sliding variable s(k)



Figure 4.29: Control signal u(k)

by applying slowly time varying disturbance at the input side of the system as shown in Figure (4.1). In CAN, it is assumed that the minimum frame size is 512 bits and



Figure 4.30: Magnified control signal u(k)



Figure 4.31: Tracking response of system when packet loss is is 10%



Figure 4.32: Tracking response of system when packet loss is 30%



Figure 4.33: Tracking response of system when packet loss is 50%



Figure 4.34: Scheduling policy of sensor to controller when packet loss is 10%



Figure 4.35: Scheduling policy of sensor to controller when packet loss is 30%



Figure 4.36: Scheduling policy of sensor to controller when packet loss is 50%

data transfer rate is 80000 bits/sec. So, the delay generated in the CAN network to transfer the data packets from sensor to controller is  $\tau_{sc} = 6.4msec$  and from controller to actuator is  $\tau_{ca} = 6.4 msec$ . The processing and the computational delays at sensor, controller and actuator are considered as 0.9msec, 0.5msec and 0.5msec respectively. Thus the total networked delay generated within the closed loop system is  $\tau = 14.7 msec.$  The fractional part of total network delay is obtained as  $\tau' = 0.49$ ,  $au_{sc}' = 0.213$  and  $au_{ca}' = 0.213$  for sampling interval of h = 30 msec. The scheduling policies of sensor to controller and controller to actuator with network under ideal condition and bandwidth sharing condition are shown in Figures (4.21) and (4.22) respectively. It can be observed that blue samples are indicated as the traffic while yellow and red samples indicate the scheduling policy for sensor to controller and controller to actuator. The trajectory response of the system for the network under ideal condition and traffic condition are shown in Figures (4.23) and (4.24). It can be noticed that under both situations the output tracks the reference trajectory for the specified networked delay. In order to show the precise effect of time delay compensation in CAN network at the output results are magnified as shown in Figures (4.25) and (4.26). It can be noticed that the effect of fractional delay from sensor to controller is compensated as the output tracks the reference input at t = 8.3 msec. The similar effect of time delay compensation for the network under traffic condition can be observed in sliding variable Figures (4.27) and (4.28) as well as control signal Figures (4.29) and (4.30). Observing the

magnified results in Figures (4.28) and (4.30), it can be noticed that the sliding variable and control signal are computed exactly after an interval of t = 1.4msec even in the presence of sensor to controller delay. Apart from time delay compensation, the proposed algorithm was examined under packet loss condition. Figures (4.31), (4.32) and (4.33) shows the results of tracking response under packet loss condition while Figures (4.34), (4.35) and (4.36) shows the instances of packet drop. It can be observed from results that when the packet loss is 50% the system goes to unstable condition. Thus it can be concluded that the system shows the satisfactory response under 30% of packet loss for specified network delay with CAN as a communication medium.

#### Switched Ethernet as a Network Medium



Figure 4.37: Scheduling policies of sensor to actuator of Switched Ethernet network under ideal condition



Figure 4.38: Scheduling policies of sensor to actuator of Switched Ethernet network under traffic condition



Figure 4.39: Tracking response of the system with Switched Ethernet as networked medium under idle condition.



Figure 4.40: Tracking response of the system with Switched Ethernet as networked medium with traffic condition



Figure 4.41: Magnified tracking response of the system with network under ideal condition



Figure 4.42: Magnified tracking response of the system with network under traffic load



Figure 4.43: Compensated sliding variable s(k)

In this segment, the nature of the system is studied in Figures (4.37) to (4.52) for Switched Ethernet type of communication medium considering the delays are deterministic in nature. In Switched Ethernet, the minimum frame size is 1024 bits and data transfer rate is 80000 bits/sec. So, the delay generated within the network to transfer the data packets from sensor to controller is  $\tau_{sc} = 12.8msec$  and from controller to actuator


Figure 4.44: Magnified compensated sliding variable s(k)



Figure 4.45: Control signal u(k)



Figure 4.46: Magnified control signal u(k)



Figure 4.47: Tracking response of system when packet loss is 10%



Figure 4.48: Tracking response of system when packet loss is 30%



Figure 4.49: Tracking response of system when packet loss is 50%



Figure 4.50: Scheduling policy of sensor to controller when packet loss is 10%



Figure 4.51: Scheduling policy of sensor to controller when packet loss is 30%



Figure 4.52: Scheduling policy of sensor to controller when packet loss is 50%

is  $\tau_{ca} = 12.8msec$ . The processing and the computational delays at sensor, controller and actuator are considered as 0.9msec, 0.5msec and 0.5msec respectively. Thus the total networked delay within the closed loop system is computed as  $\tau = 27.5msec$ . The fractional part of total network delay is obtained as  $\tau' = 0.91$ ,  $\tau'_{sc} = 0.426$  and  $\tau'_{ca} = 0.426$  for sampling interval of h = 30msec. The scheduling policies of sensor to controller and controller to actuator with network under ideal condition and bandwidth sharing condition are shown in Figures (4.37) and (4.38) respectively. The trajectory

Algorithm	Comparison Results			
	$ au_{CAN}$	$ au_{Ether}$	$T_s$	Response
Coventional SMC	14.7msec	25.7msec	Undefined	unstable
Proposed method	14.7msec	25.7msec	1sec	stable

Table 4.2: Comparison of Proposed Algorithm with Conventional SMC in True Time

response of the system for the network under ideal condition and traffic condition are shown in Figures (4.39) and (4.40). It can be noticed that under both the situations the output tracks the reference trajectory for the specified networked delay. In order to show the exact effect of time delay compensation in Switched Ethernet network at the output, results are magnified as shown in Figures (4.41) and (4.42). It can be noticed that the effect of fractional delay from sensor to controller is compensated as the output tracks the reference input at t = 14.7 msec. The similar effect of time delay compensation for the network under traffic condition can be observed in sliding variable Figures (4.43) and (4.44) as well as control signal Figures (4.45) and (4.46). Observing the magnified results in Figures (4.44) and (4.46), it can be noticed that the sliding variable and control signal are computed exactly after an interval of t = 1.4msec even in the presence of sensor to controller delay. Apart from time delay compensation, the proposed algorithm was examined under packet loss condition. Figures (4.47), (4.48) and (4.49) shows the results of tracking response under packet loss condition while Figures (4.50), (4.51) and (4.52) shows the instances of packet drop. It can be observed from results that when the packet loss is 50% the system goes to unstable condition. Thus, it can be concluded that the system shows the satisfactory response under 30% of packet loss for specified network delay with Switched Ethernet as a communication medium.

### Comparison of proposed algorithm with conventional sliding mode control under CAN and Switched Ethernet as a network medium



Figure 4.53: Time delay compensation scheme with CAN as a networked medium.



Figure 4.54: Tracking response of conventional SMC with CAN as a communication medium.



Figure 4.55: Time delay compensation using Switched Ethernet as a communication medium.



Figure 4.56: Tracking response of conventional SMC with Switched Ethernet as a networked medium.

In this section, the results of proposed algorithm were compared with conventional SMC for CAN and Switch Ethernet communication medium. Figures (4.53), (4.54), (4.55) and (4.56) shows the comparative results of proposed algorithm and conventional SMC for discrete time sliding mode control. It can be observed from comparison that conventional SMC shows unstable response for the specified networked delay. Thus, Thiran approximation proved to be efficient technique in discrete time sliding mode control under packet loss condition and matched uncertainty. The comparison of proposed algorithm and conventional SMC for CAN and Switched Ethernet communication medium are summarized in Table-4.2.

### 4.5 Conclusion

In this chapter, we explored Thiran's approximation technique for fractional delay compensation in discrete-time domain. Using this approach, non-switching type discrete time sliding mode controller is designed which computes the control actions in the presence of network delay and matched uncertainty. The stability of the closed loop NCS is assured by using Lyapunov approach such that system states remain within the specified band. The effectiveness of the derived algorithms is tested using Illustrative example and brushless DC motor setup with deterministic networked delay and matched uncertainty. The experimental results are compared with non-switching SMC, switching SMC as well as conventional algorithm. The comparative results proved that the fractional delay approximated using Thiran Approximation is most efficient technique as it compensates the fractional network delays in discrete-time domain. Moreover, the simulation results carried out using illustrative example and experimental results carried out in the presence of DC servo motor plant proved that the control algorithm designed using non-switching reaching law is robust and efficient algorithm as it provides the faster convergence with less chattering in discrete-time domain. Further, the efficiency of non-switching controller designed using Thiran Approximation was examined under simulated CAN and Switched Ethernet networked medium using True Time. The results shows that the proposed control algorithm designed using Thiran Approximation compensates the networked delay in the presence of network non-idealities (time delay, packet loss and matched uncertainty).

### **CHAPTER 5**

# MULTIRATE OUTPUT FEEDBACK DISCRETE-TIME SLIDING MODE CONTROLLER FOR FRACTIONAL DELAY COMPENSATION

#### 5.1 Introduction

In NCS, if states information is available for measurement then state feedback is the simplest way for designing the SMC controller. In reality for any network based control system most of the states are observable but they are immeasurable. So it is essential to design the SMC controller based on output information which is always measurable. This chapter summarizes the designing of multirate output feedback based discrete-time sliding mode controller in which the control input is computed based on the system outputs and past control signals by taking full advantage of network transmission. A Thiran Approximation technique is used to compensate the networked delays. The sensor to controller delay is compensated at the sliding surface, while sliding mode controller (SMC) is utilized to compute the control sequences. The stability of the closed loop NCSs is derived using Lyapunov approach. Simulations results are given to demonstrate the effectiveness of the proposed approach.

#### 5.2 **Problem Formulation**

Figure (5.1) represents the schematic diagram of NCS with time delay compensator and multirate output feedback approach. As shown in Figure (5.1), the state information and control information are exchanged between plant and controller through the networks suffering from sensor to controller delay ( $\tau_{sc}$ ) and controller to actuator delay ( $\tau_{ca}$ ). The state information are computed based on multirate output feedback approach. In this approach, the system states are estimated through the output information and error between the computed state and actual state of the system goes to zero once the multirate output sample is available. While, in coventional output feedback the error between the actual states and estimated states decreases asymtotically and approches to zero at infinite time. In multirate output feedback technique, the control inputs and plant output signals are sampled at different sampling intervals. It is assumed that the sensor output measurements are faster than control input signals. The mathematical expression of



Figure 5.1: Block diagram of NCS with multirate output feedback approach

control input signal sampled at  $\phi$  sampling interval is given by:

$$x((k+1)\phi) = F_{\phi}x(k\phi) + G_{\phi}(u(k\phi) + d(k\phi)),$$
(5.1)

$$y(k\phi) = Cx(k\phi). \tag{5.2}$$

Let the output samples be sampled at  $\zeta$  sampling interval given by,

$$\zeta = \frac{\phi}{\Lambda},\tag{5.3}$$

where,  $\Lambda$  is the positive interger  $\geq$  observability index of the system.

Similarly, the mathematical expression of the sensor output data sampled at  $\zeta$  sampling interval is given as:

$$x((k+1)\zeta) = F_{\zeta}x(k\zeta) + G_{\zeta}(u(k\zeta) + d(k\zeta)), \qquad (5.4)$$

$$y(k\zeta) = Cx(k\zeta). \tag{5.5}$$

**Remark** – 6: The stability of closed loop system is guaranteed, if pair  $(F_{\phi}, G_{\phi})$  is controllable and pair  $(F_{\zeta}, C)$  is observable.

The output of multirate output feedback is expressed in mathematical form as:

$$\hat{x}(k) = H_y y_k + H_u u(k-1) + H_d d(k-1),$$
(5.6)

where,  $y_k$  represents the output stack, u(k-1) represents the past control input signal, d(k-1) represents the past disturbance signal which are available from measurements.

$$H_y = A_{\phi}(Z_0^T Z_0)^{-1} Z_0^T, H_u = B_{\phi} - H_y X_0, H_d = d_{\phi} - H_y C_d$$

Thus, observing Eqn. (5.6) it can be noticed that the output of multirate depends on past output samples, past control signal and past disturbance signal. Thus, the output of sensor computed based on multirate ouput feedback approach will be applied to the controller through network.

Remark – 7: In this chapter nature of the system, total fractional network delays  $\tau'$ , sensor to controller fractional delay  $\tau'_{sc}$  and controller to actuator fractional delay  $\tau'_{ca}$  would remain same as mentioned in section (3.3).

**Problem Statement**: The main objective, is to design multirate output feedback based discrete-time sliding mode control law for system (3.3,3.4) in the presence of fractional delay  $\tau'_{sc}$  and  $\tau'_{ca}$  satisfying (3.5) and matched uncertainty satisfying (3.7).

The next section describes the mathematical derivation of sliding surface that compensates the effect of fractional delay ( $\tau'$ ) in the presence of matched uncertainty d(k)using multirate output feedback approach mentioned in Eqn. (5.6).

## 5.3 Design of Sliding Surface Using Multirate Output Feedback

For designing DT-SMC, we propose the sliding surface using Thiran Approximation rule and multirate output feedback approach in the form of Lemma -2 as under.

Lemma – 2: The sliding surface for the given system (3.3, 3.4) with sensor to controller fractional network delay satisfying assumption (1) and (2) specified in section (3.3) is given as:

$$s(k) = C_s \hat{x}(k) - \alpha C_s \hat{x}(k-1),$$
(5.7)

where,

 $\alpha = \frac{\hat{\tau}_{sc}}{1 + \hat{\tau}_{sc}},$   $C_s$  represents the sliding gain.

**Proof**: Let the sliding surface with network fractional delay from sensor to controller  $(\tau'_{sc})$  is given by:

$$s(k) = C_s \hat{x}(k - \tau'_{sc}), \qquad (5.8)$$

where,  $C_s$  indicates the sliding gain,  $\hat{x}(k - \hat{\tau}_{sc})$  indicates the delayed state vector. The value of sliding gain  $C_s$  is computed using discrete LQR method with proper selection of Q and R matrices.

Applying z-transform to Eqn. (5.8) we get:

$$s(z) = C_s \hat{x}(z) z^{-\tau'_{sc}},$$
 (5.9)

where,  $\tau'_{sc} = \nu = \tau_{sc}/h$ .

Considering  $\tau'_{sc} \prec 1$  and using Eqn. (3.9)  $z^{-\tau'_{sc}}$  can be approximated as,

$$z^{-\tau'_{sc}} = \sum_{k=0}^{1} (-1)^k \binom{l}{k} \Pi_{i=0}^1 \frac{2\tau'_{sc} + i}{2\tau'_{sc} + k + i} z^{-k}.$$
(5.10)

The above Eqn. (5.10) can be further expanded as:

$$z^{-\tau'_{sc}} = \left[ (-1)^0 \begin{pmatrix} 1\\0 \end{pmatrix} \left\{ \frac{2\tau'_{sc}}{2\tau'_{sc}} * \frac{2\tau'_{sc} + 1}{2\tau'_{sc} + 1} \right\} z^0 + (-1)^1 \begin{pmatrix} 1\\1 \end{pmatrix}$$

$$\left\{ \frac{2\tau'_{sc}}{2\tau'_{sc} + 1} * \frac{2\tau'_{sc} + 1}{2\tau'_{sc} + 2} \right\} z^{-1} \right].$$
(5.11)

On simplifying we get,

$$z^{-\tau'_{sc}} = 1 - \alpha z^{-1}.$$
 (5.12)

where,  $\alpha = \frac{\tau'_{sc}}{\tau'_{sc}+1}$ .

Substituting Eqn. (5.12) into (5.9),

$$s(z) = [1 - \alpha z^{-1}]\hat{x}(z)C_s.$$
(5.13)

Further solving,

$$s(z) = \hat{x}(z)C_s - \alpha z^{-1}\hat{x}(z)C_s.$$
(5.14)

Appling inverse *z*-transform we have,

$$s(k) = C_s \hat{x}(k) - \alpha C_s \hat{x}(k-1).$$
(5.15)

This completes the proof.

Observing Eqn. (5.15) it can be noticed that the sliding surface s(k) depends past output samples, past control signal, past disturbance signal and parameter ' $\alpha$ ' approximated through thiran approximation rule. Thus, the effect of sensor to controller fractional delay  $\tau'_{sc}$  in the sliding surface at each sampling instants h is compensated through proposed technique.

Now, we are ready to propose multirate output feedback based discrete-time sliding mode control law using the sliding surface (5.7).

## 5.4 Design of Discrete-Time Networked Sliding Mode Control (DNSMC) Using Multirate Output Feedback

In this section, discrete-time sliding mode control law is derived for deterministic network fractional delays ( $\tau'_{sc}$  and  $\tau'_{ca}$ ) along with its stability using sliding surface (5.7) represented in form of Theorem – 3.

Theorem -3: The discrete-time sliding surface (5.7) is reached in a finite time in the presence of networked delays satisfying (3.5) and matched uncertainty (3.7) provided the control law is designed as:

$$u(k) = -(C_s G)^{-1} [H_{dy} y_k + H_{ud} u(k-1) + H_{dd} d(k-1) - Js(k) - d_s(k)$$
(5.16)  
+ $d_1 ] - d(k).$ 

where,

$$H = (C_s F), I = \alpha C_s, J = [1 - q(s_c(k))], (H - I)H_y = H_{dy} (H - I)H_u = H_{ud}, (H - I)H_d = H_{dd}.$$

**Proof** : The reaching law proposed in [Bartoszewicz and Lesniewski (2014)], is used to derive the control law since it provides faster convergence. The reaching law in the presence of sensor to controller fractional delay ( $\tau'_{sc}$ ) is given by:

$$s[(k+1)] = \{1 - q[s(k)]\}s(k) - d(k) + d_1,$$
(5.17)

where,

 $\{q[s(k)]\} = \frac{\psi}{\psi+|s(k)|}, d(k)$  represents the disturbance,  $d_1 = \frac{d_u+d_l}{2}$ , mean value of  $d(k), d_2 = \frac{d_u-d_l}{2}$ , deviated value of d(k) and  $\psi$  is the designer's constant satisfying,

$$\psi \ge d_2. \tag{5.18}$$

The reaching law in Eqn. (5.17) contains the disturbance term which is applied through the network. Thus, the compensated disturbance  $d_s(k)$  is given as:

$$d_s(k) = d(k) - \alpha d(k-1).$$
(5.19)

Using Eqn. (5.7), Eqn. (5.17) can be rewritten as

$$\hat{x}(k+1)C_s - \alpha C_s \hat{x}(k) = [1 - q(s(k))]s(k) - d_s(k) + d_1.$$
(5.20)

Substituting the value of  $\hat{x}(k+1)$  in Eqn. (5.20),

$$C_s[F\hat{x}(k) + G(u(k) + d(k))] - \alpha C_s\hat{x}(k) = [1 - q(s(k))]s(k) - d_s(k) + d_1.$$
(5.21)

Further simplification gives,

$$C_s F\hat{x}(k) + C_s G(u(k) + d(k)) - \alpha C_s \hat{x}(k) = [1 - q(s(k))]s(k) - d_s(k) + d_1.$$
(5.22)

The above Eqn. (5.22) further can be expressed in the terms of control law as,

$$u(k) = -(C_s G)^{-1} [H_{dy} y_k + H_{ud} u(k-1) + H_{dd} d(k-1) - Js(k) - d_s(k) \quad (5.23)$$
$$+ d_1] - d(k).$$

This completes the proof.

Observing the Eqn. (5.23) it can be noticed that the control law is computed based on the information available in the output stack  $y_k$  and not the state information. Thus, using sliding surface (5.7) and control law (5.16) the stability of the closed loop NCS is derived such that the system states will remain within the band in the presence of network fractional delay ( $\tau'$ ) and matched uncertainty d(k).

#### 5.4.1 Stability

The system states in (3.3, 3.4) will slides on the sliding surface (5.7) and maintain on it with the controller designed in (5.23) under network fractional delay ( $\tau'$ ) satisfying

(3.5) and matched uncertainty (3.7) such that for any  $\psi \ge d_2$  and  $\gamma \prec 0$  the following condition should hold true:

$$0 \leq \kappa \prec s^{T}(k)s(k). \tag{5.24}$$

**Proof** : Consider the sliding surface (5.7) as,

$$s(k+1) = C_s \hat{x}(k+1) - \alpha C_s \hat{x}(k).$$
(5.25)

Let the Lyapunov function be given by,

$$V_s(k) = s^T(k)s(k).$$
 (5.26)

Taking forward difference of the above Eqn. (5.26),

$$\Delta V_s(k) = s^T(k+1)s(k+1) - s^T(k)s(k).$$
(5.27)

Substituting the value of s(k + 1) from Eq. (5.7) we get,

$$\Delta V_s(k) = [C_s \hat{x}(k+1) - \alpha C_s \hat{x}(k)]^T [C_s \hat{x}(k+1) - \alpha C_s \hat{x}(k)]$$
(5.28)  
$$-s^T(k)s(k).$$

Substituting the value of  $\hat{x}(k+1)$  we get,

$$\Delta V_s(k) = [C_s[F\hat{x}(k) + G(u(k) + d(k))] - \alpha C_s \hat{x}(k)]^T [C_s[F\hat{x}(k) + G(u(k) \ (5.29) + d(k))] - \alpha C_s \hat{x}(k)] - s^T(k)s(k).$$

Substituting the value of u(k) and further solving it we have,

$$\Delta V_s(k) = [[1 - q(s(k))]s(k) - d_c(k) + d_1]^T * [[1 - q(s(k))]s(k) - d_c(k) \quad (5.30)$$
$$+ d_1] - s^T(k)s(k).$$

It can be seen that the term  $[[1 - q(s(k))]s(k) - d_c(k) + d_1]$  contains the disturbance term in  $\Delta V_s(k)$ . Let  $\kappa = [[1 - q(s(k))]s(k) - d_c(k) + d_1]^T * [[1 - q(s(k))]s(k) - d_c(k) + d_1]$ 

Then we have,

$$\Delta V_s(k) = \kappa - s^T(k)s(k). \tag{5.31}$$

If  $\kappa$  is closed tuned to zero by appropriately selecting the parameter  $\psi$ , then  $s^T(k)s(k)$ will be larger than  $\kappa$ . Therefore, for any constant parameter  $\gamma$ , we have  $\Delta V_s(k) \prec -\gamma s^T(k)s(k)$  which guarantees the convergence of  $\Delta V_s(k)$ . This completes the proof.

### 5.5 Results and Discussion

In this section the efficacy of the designed control algorithm based on MROF approach is validated in the presence of fractional network delay and matched uncertainty applied at the input channel of the system. The simulation results are carried out using illustrative example [Wu and Chen (2007)] in MATLAB enviornment considering different fractional delays.

#### 5.5.1 Simulation Results

Consider the continuous-time LTI system as,

$$\dot{x}(t) = Ax(t) + Bu(t - \tau) + Dd(t),$$
 (5.32)

$$y(t) = Cx(t), \tag{5.33}$$

where,

$$A = \begin{bmatrix} -0.7 & 2\\ 0 & -1.5 \end{bmatrix}, B = \begin{bmatrix} -0.03\\ -1 \end{bmatrix},$$
$$C = \begin{bmatrix} 1 & 0 \end{bmatrix}, D = \begin{bmatrix} 1\\ 1 \end{bmatrix}$$

Discretizing the above system with sampling interval of h = 30msec we get,

$$x(k+1) = Fx(k) + Gu(k - \tau') + d(k),$$
(5.34)

$$y(k) = Cx(k), \tag{5.35}$$



Figure 5.2: Actual state  $x_1$  and estimated state  $\hat{x}_1$  with initial condition  $x_1=0.5$  for  $\tau = 12.8msec$ 



Figure 5.3: Actual state  $x_2$  and estimated state  $\hat{x}_2$  with initial condition  $x_2=0.5$  for  $\tau = 12.8msec$ 



Figure 5.4: Magnified result of actual state  $x_1$  and estimated state  $\hat{x}_1$  with initial condition  $x_1=0.5$  for  $\tau = 12.8msec$ 

Figures (5.2) to (5.18) shows the nature of system using multirate output feedback approach under networked environment. The robustness of the system is validated by applying slow time varying disturbance applied at the input side of channel. The networked delay is considered as the time required for the data packets to travel from sensor to controller and controller to actuator. The sliding gain parameter is calculated using discrete LQR method which comes out to be  $C_s = [-1.77 -2.766]$  for



Figure 5.5: Magnified result of actual state  $x_2$  and estimated state  $\hat{x}_2$  with initial condition  $x_2$ =0.5 for  $\tau = 12.8msec$ 



Figure 5.6: Sliding surface s(k) for  $\tau = 12.8msec$ 



Figure 5.7: Magnified sliding surface s(k) for  $\tau = 12.8msec$ 



Figure 5.8: Control signal u(k) for  $\tau = 12.8msec$ 



Figure 5.9: Magnified control signal u(k) for  $\tau = 12.8msec$ 

Q = diag(1000, 1000) and R = 1. The sliding band |s(k)| remains same as that of previous cases. The initial condition of the system in both the cases are considered



Figure 5.10: Actual state  $x_1$  and estimated state  $\hat{x}_1$  with initial condition  $x_1=0.5$  for  $\tau = 25.6msec$ 



Figure 5.11: Actual state  $x_2$  and estimated state  $\hat{x}_2$  with initial condition  $x_2$ =0.5 for  $\tau = 25.6msec$ 



Figure 5.12: Magnified result of actual state  $x_1$  and estimated state  $\hat{x}_1$  with initial condition  $x_1=0.5$  for  $\tau = 25.6msec$ 



Figure 5.13: Magnified result of actual state  $x_2$  and estimated state  $\hat{x}_2$  with initial condition  $x_2=0.5$  for  $\tau = 25.6msec$ 

as  $[x_1 \ x_2] = [0.5 \ 0.5]^T$ . The sensor output signal is sampled at  $\zeta = 2msec$  and the control input signal is sampled at  $\phi = 6msec$  considering  $\Lambda = 3$ . The efficacy of the proposed algorithm is checked under networked delay of  $\tau = 12.8msec$  and  $\tau = 25.6msec$  respectively.

Figures (5.2) to (5.9) show the regulatory response of state variables, sliding variable and control signal for  $\tau = 12.8msec$  with  $\tau_{sc} = 6.4msec$  and  $\tau_{ca} = 6.4msec$ . Since



Figure 5.14: Sliding surface s(k) for  $\tau = 25.6msec$ 



Figure 5.15: Magnified sliding surface s(k) for  $\tau = 25.6msec$ 



Figure 5.16: Control Signal u(k) for  $\tau = 25.6msec$ 



Figure 5.17: Magnified control signal u(k) for  $\tau = 25.6msec$ 



Figure 5.18: Output stack  $y_k$ 

the system is discretized at h = 30 msec, the fractional part of delay is computed to be  $\tau' = 0.426$ ,  $\tau'_{sc} = 0.213$  and  $\tau'_{ca} = 0.213$  respectively. Figures (5.2) and (5.3) show the regulatory response of the system states for the specified networked delay. It can be

observed that the actual states  $(x_1, x_2)$  and estimated states  $(\hat{x}_1, \hat{x}_2)$  both converges to zero from their given initial condition. In order to prove the effect of multirate output feedback and time delay compensation the results are magnified as shown in Figures (5.4) and (5.5) respectively. It can be noticed that in both cases the estimated state variables follows the actual state variable at  $\zeta = 2msec$  which means the error becomes zero once the multirate output sample is available. Apart from these, both the states are computed from first sampling instants. Thus, the effect of fractional time delay from sensor to controller and controller to actuator is compensated. The similar effect of compensation is observed in sliding variable and control signal results shown in Figures (5.6) to (5.9). Observing the magnified results shown in Figures (5.7) and (5.9) it can be noticed that the sliding variable and control signal both are computed from first sampling instants even in the presence of sensor to controller fractional delay.

Figures (5.10) to (5.17) show the response of the system in terms of state variables, sliding variable and control signal for  $\tau = 25.6msec$  with  $\tau_{sc} = 12.8msec$  and  $\tau_{ca} = 12.8msec$ . The fractional part of delay is computed to be  $\tau' = 0.8533$ ,  $\tau'_{sc} = 0.426$  and  $\tau'_{ca} = 0.426$  for h = 30msec. Figures (5.10) and (5.11) show the regulatory response of the system state variables for specified networked delay. It can be noticed that actual states  $(x_1, x_2)$  and estimated states  $(\hat{x}_1, \hat{x}_2)$  both converges to zero from their given initial condition. The magnified regulatory response of the same is shown in Figures (5.12) and (5.13). It can be observed that both states variables start converging to zero at first sampling instant and the estimated states follows the actual state variables at  $\zeta = 2msec$ . Thus, the effect of network delay from sensor to controller and controller to actuator is nullified at the output. The same effect is observed in sliding variable Figure (5.14) as well as control signal Figure (5.16). Observing the magnified results shown in Figures (5.17) and (5.17), it can be concluded that the effect of fractional delay from sensor to controller is compensated through Thiran Approximation. Figure (5.18) shows the result of output stack for both the cases.

Thus, from above results it can be noticed that the proposed control algorithm works efficiently for the networked delay of 12.8*msec* to 25.6*msec*. The sliding surface and control algorithm derived using multirate output feedback approach with Thiran Approximation takes care of sensor to controller fractional delay and controller to actuator fractional delay in the presence of matched uncertainty.

### 5.6 Conclusion

In this chapter, a new idea of compensating the fractional delay in forward as well as feedback channel is introduced using the concept of multirate output feedback approach. The sensor to controller fractional delay is compensated using Thiran Approximation at the sliding surface while controller to actuator fractional delay is compensated at actuator end. Using this novel approach a multirate output feedback discrete-time networked sliding mode control law is derived that compute the control sequences in the presence of network fractional delay and matched uncertainty. The main advantage of using multirate output feedback approach is that the system states are computed based on the output information available and the error between computed as well as estimated state variables becomes zero exactly after one sampling instant. Stability of the closed loop NCS is ensured using Lyapunov approach such that system states would remain within the band under network non-idealities. The simulation results carried out using proposed technique shows the enhanced response and compensates the effect of fractional delay accurately in discrete-time domain. The proposed algorithm is validated for deterministic type of network fractional delay.

### **CHAPTER 6**

# DISCRETE-TIME SLIDING MODE CONTROLLER FOR RANDOM FRACTIONAL DELAYS AND PACKET LOSS

#### 6.1 Introduction

In Networked Control System the behaviour of network delays generally depends on the characteristics of communication medium as well as occupancy of channel by different elements. When large number of sensors, controllers and actuators share their information through the common comunication medium then the network delays and packet losses are random in nature. In this chapter, a novel approach is presented for designing discrete-time sliding mode controller by treating random fractional delay and packet loss seperately. The fractional delay that occurs within sampling period while transmitted from sensor to controller and controller to actuator channel are modelled using Poisson's distribution function and are approximated using Thiran's delay approximation technique for designing the discrete-time sliding mode controller. The packet loss that occur in communication channel between sensor to controller and controller to actuator are treated with Bernoulli's distribution function and compensated at controller end as well as actuator end. Further, Lyapunov approach is used to determine the stability of closed loop NCSs with proposed discrete-time SMC controller. The feasibility and efficiency of the proposed control methodology is outlined through simulation and experimental results which shows a significant response even in the presence of random fractional delay, packets loss and matched uncertainities.

### 6.2 **Problem Formulation**

The block diagram of NCS with communication networked medium and packet loss situation is shown in Figure (6.1). The state information as well as control signal are

transmitted to the controller and actuator through the network medium. During transmission, the state information and control signal experience time delay from sensor to controller channel and controller to actuator channel respectively. These delays are broadly defined as the time required for the data packets to travel within the network. While transmission, if data packets takes longer duration of time to travel within the network than sampling interval it is called as packet loss. This situation mainly arises due to network congestion, node competition or heavy traffic in the network.



Figure 6.1: Block diagram of NCS with fractional delay compensation and packet loss

Consider the linear time invariant SISO system with random network delay in continuoustime domain as:

$$\dot{x}(t) = Ax(t) + Bu(t - \tau_r) + Dd(t),$$
(6.1)

$$y(t) = Cx(t), (6.2)$$

where  $x \in \mathbb{R}^n$  is system state vector,  $u \in \mathbb{R}^m$  is control input,  $y \in \mathbb{R}^p$  is system output,  $A \in \mathbb{R}^{n \times n}$ ,  $B \in \mathbb{R}^{n \times m}$ ,  $C \in \mathbb{R}^{p \times n}$ ,  $D \in \mathbb{R}^{n \times m}$  are the matrices of appropriate dimensions, d(t) is the matched bounded disturbance with  $|d(t)| \leq d_{max}$  and  $\tau_r$  is the total random network delay in continuous-time domain. The discrete form of system (6.1) and (6.2) is:

$$x(k+1) = Fx(k) + Gu(k - \hat{\tau}) + d(k), \tag{6.3}$$

$$y(k) = Cx(k), \tag{6.4}$$

where  $F = e^{Ah}$ ,  $G = \int_0^h e^{At} B dt$ ,  $d(k) = \int_0^h e^{At} D d((k+1)h - t) dt \in O(h)$ . Since  $|d(t)| \leq d_{max}$ , it can be inferred that d(k) is also bounded and O(h) [Mehta and Bandy-opadhyay (2016)]. For simplicity, it is assumed that d(k) is slowly varying and remain constant over the interval  $kh \leq t \leq (k+1)h$  [Mehta and Bandyopadhyay (2016)]. The total random fractional delay  $(\hat{\tau})$  occurring within the network denoted as,

$$\hat{\tau} = \frac{\tau_r}{h}$$

where h is the sampling interval.

**Remark**-8: In this work, the non-integer form of random fractional delay  $\hat{\tau}$  is considered instead of integer form in discrete-time domain in order to compensate the precise effect of random network delay occuring at each sampling instants.

Assumption -3: Network induced fractional delays are random in nature and therefore satisfying

$$\hat{\tau}_l \le \hat{\tau} \le \hat{\tau}_u, \tag{6.5}$$

where  $\hat{\tau}_l$  and  $\hat{\tau}_u$  indicates the lower and upper bound of total random fractional network delays.

The total random network induced fractional delay is the combination of sensor to controller fractional delay ( $\hat{\tau}_{sc}$ ) and controller to actuator fractional delay ( $\hat{\tau}_{ca}$ ) having random behaviour which is given as,

$$\hat{\tau} = \hat{\tau}_{sc} + \hat{\tau}_{ca},\tag{6.6}$$

where,  $\hat{\tau}_{sc} = \frac{\tau_{rsc}}{h}$  and  $\hat{\tau}_{ca} = \frac{\tau_{rca}}{h}$ .

Assumption -4: In this work, it is assumed that only single packet loss occurs. The assumption is justifying as the packet time delay more than sampling period is considered as dropped or lost packet. **Problem Statement:** To design robust non-switching discrete-time sliding mode controller for the system (6.3,6.4) in the presence of random fractional delays  $\hat{\tau}_{sc}$  and  $\hat{\tau}_{ca}$ , matched uncertainty and packet loss situation satisfying condition (6.5), (3.7) and assumption (4).

The sliding mode controller involves the sliding surface design that steers the system states towards the surface and control law that computes the control sequences.

The next section proposes the design of sliding surface that compensates the effect of random fractional delay occuring between sensor to controller in NCSs. It also discusses about the modelling of random fractional delay and packet loss.

## 6.3 Sliding Surface with Random Fractional Delay and Packet Loss

The effect of random fractional delay in discrete-time domain due to presence of networked medium is compensated in sliding surface presented in Lemma -3.

Lemma – 3: The sliding surface s(k) for a random fractional delay ( $\hat{\tau}_{sc}$ ), satisfying conditions (6.5) and (3.7) with packet loss for the system (6.3,6.4) is given as:

$$s(k) = (1 - \bar{\alpha})x'_{c}(k) + \bar{\alpha}x'_{c}(k - 1),$$
(6.7)

where,

 $x'_{c}(k) = C_{s}x(k) - \varsigma C_{s}x(k-1), x'_{c}(k-1) = C_{s}x(k-1) - \varsigma C_{s}x(k-2), \varsigma = \frac{\hat{\tau}_{sc}}{\hat{\tau}_{sc}+1}, C_{s}$  is the sliding gain and  $\bar{\alpha}$  is the probability of the data packet lost.

**Proof**: In discrete-time domain there are various algorithms for modelling the random variables such as Bernoulli's distribution, Geometric distribution, Poisson's distribution, Probability distribution, Binomial distribution and Pascal distribution. Among these algorithms Bernoulli's distribution and Probability distribution are two widely used algorithms for mathematical modelling of integer type of network delays in discretetime domain. Since, the sensor to controller delay  $(\hat{\tau}_{sc})$  is random and fractional in nature, Poisson's distribution is the most suitable approach in discrete-time domain. Poisson's distribution is used to model the random variables of smaller values on the basis of number of events occurred over the specified interval of time. The occurrence of the events is based on the number of trials required to generate the event. So, using this approach the random fractional delay is modelled by assuming that at each sampling instants an event takes place such that the networked delay might be lesser or higher than sampling interval.

Thus, the communicated state variable over the network having random fractional delay from sensor to controller is given as:

$$x_c(k) = x(k - \hat{\tau}_{sc}), \tag{6.8}$$

where,  $\hat{\tau}_{sc}$  is the fractional form of sensor to controller delay in discrete-time domain that takes the values in a finite set, that is,  $\{\hat{\tau}_{sc}\} \in \{d_1, d_2, \dots, d_q\}$ . Thus, the sensor to controller fractional delay  $\{\hat{\tau}_{sc}\}$  can be modelled using Poisson's distribution with probabilities given by,

$$Pr\{\hat{\tau}_{sc} = d_v\} = E\{d_v\} = \beta_v, v = 1, 2, ..., q.$$
(6.9)

where,  $\beta_v$  is the positive scalar and  $\sum_{v=0}^q \beta_v = 1$ ,  $E\{d_v\}$  is the expectation of the stochastic variable  $d_v$ . The mathematical representation of  $\beta_v$  with poisson's distribution is given by:

$$\beta_v = \frac{\lambda^w e^{-\lambda}}{w!}; w = 0, 1, 2, 3, \dots$$
(6.10)

where, w indicates the number of trials,  $\lambda$  denotes average number of events per interval and e denotes the Euler's number.

Applying z-Transform to Eqn. (6.8) we get,

$$x_c(z) = x(z)z^{-\hat{\tau}_{sc}},$$
 (6.11)

where,  $\hat{\tau}_{sc} = \nu = \frac{\tau_{rsc}}{h}$ .

Using Eqn. (3.9),  $z^{-\hat{\tau}_{sc}}$  can be approximated as,

$$z^{-\hat{\tau}_{sc}} = \Sigma_{k=0}^{1} (-1)^{k} \binom{l}{k} \Pi_{i=0}^{1} \frac{2\hat{\tau}_{sc} + i}{2\hat{\tau}_{sc} + k + i} z^{-k},$$
(6.12)

The above Eqn. (6.12) can be further expanded as:

$$z^{-\hat{\tau}_{sc}} = \left[ (-1)^0 \begin{pmatrix} 1\\0 \end{pmatrix} \left\{ \frac{2\hat{\tau}_{sc}}{2\hat{\tau}_{sc}} * \frac{2\hat{\tau}_{sc} + 1}{2\hat{\tau}_{sc} + 1} \right\} z^0 + (-1)^1 \begin{pmatrix} 1\\1 \end{pmatrix}$$

$$\left\{ \frac{2\hat{\tau}_{sc}}{2\hat{\tau}_{sc} + 1} * \frac{2\hat{\tau}_{sc} + 1}{2\hat{\tau}_{sc} + 2} \right\} z^{-1} \right],$$
(6.13)

On simplifying we get,

$$z^{-\hat{\tau}_{sc}} = 1 - \varsigma z^{-1}, \tag{6.14}$$

where,  $\varsigma = \frac{\hat{\tau}_{sc}}{\hat{\tau}_{sc}+1}$  and  $\hat{\tau}_{sc}$  is the random fractional delay defined in Eqns. (6.9) and (6.10) respectively.

Substituting Eqn. (6.14) into (6.11) we have,

$$x_c(z) = x(z)[1 - \varsigma z^{-1}], \tag{6.15}$$

Further expanding it we obtain,

$$x_c(z) = x(z) - \varsigma z^{-1} x(z), \tag{6.16}$$

Applying inverse *z*-Transform, we get,

$$x_c(k) = x(k) - \varsigma x(k-1).$$
(6.17)

The sensor output signal x(k) generated at each sampling instant h is sent to the controller via communication channel. From Eqn. (6.17), it can be noticed that at each sampling instant the effect of fractional delay ( $\hat{\tau}_{sc}$ ) is compensated in the communicated state variable  $x_c(k)$  using the immediate past state, the current state and parameter  $\varsigma$ . The same communicated state variable  $x_c(k)$  is further used to compute the sliding surface.

The mathematical representation of compensated random packet loss within the net-

work is given as,

$$x_p(k) = (1 - \alpha(k))x_c(k) + \alpha(k)x_c(k-1),$$
(6.18)

where  $x_c(k)$  is the compensated communicated state variable available at the controller and  $\alpha(k)\epsilon R$  is the stochastic variable which is represented as Bernoulli's distributed sequence with,

$$Pr\{\alpha(k) = 1\} = E\{\alpha(k)\} = \bar{\alpha}, \tag{6.19}$$

$$Pr\{\alpha(k) = 0\} = 1 - E\{\alpha(k)\} = 1 - \bar{\alpha}, \tag{6.20}$$

where,  $0 \leq \bar{\alpha} < 1$  implies the probability that the data packet is lost and  $E\{\alpha(k)\}$  is the expectation of the stochastic variable  $\alpha(k)$ .

Thus,  $x_p(k)$  can be written as:

$$x_p(k) = (1 - \bar{\alpha})x_c(k) + \bar{\alpha}x_c(k - 1), \tag{6.21}$$

where,  $\bar{\alpha}$  is the probability of the data packet lost defined in Eqns. (6.19) and (6.20). Let the sliding variable that compensate the effect of random fractional delay and packet loss is given by:

$$s(k) = C_s x_p(k), \tag{6.22}$$

where,  $C_s$  is the sliding gain which is calculated using discrete LQR method through proper selection of Q and R matrices.

Substituting the value of  $x_p(k)$  in Eqn. (6.22) we have:

$$s(k) = (1 - \bar{\alpha})x'_{c}(k) + \bar{\alpha}x'_{c}(k - 1), \qquad (6.23)$$

where,

$$x'_{c}(k) = C_{s}x(k) - \varsigma C_{s}x(k-1), \, x'_{c}(k-1) = C_{s}x(k-1) - \varsigma C_{s}x(k-2)$$

This completes the proof.

It is assumed that the state packet delay is smaller than sampling interval if the network is free from congestion. So the data packet  $x_p(k)$  with delay is used without loss to compute the sliding surface. However, if the network is overloaded due to traffic or congestion, the state packet delay will be larger than sampling interval. At that instance,  $x_p(k-1)$  will be used to compute the sliding surface. So from Eqn. (6.23), it can be easily noticed that at each sampling instant the effect of random fractional delay and packet loss in the actual system states are compensated at the sliding surface when it takes the value  $x'_c(k-1)$  with probability  $\bar{\alpha}$  and  $x'_c(k)$  with probability  $(1-\bar{\alpha})$ .

Once the sliding surface is designed, the next step is to design the discrete-time sliding mode control law which is presented in next section.

# 6.4 Discrete-Time Networked Sliding Mode Control for NCSs With Random Fractional Delays and Packet Loss

This section presents the derivation of non-switching discrete-time sliding mode control law for NCS using the sliding surface (6.23) as Theorem -4 below.

**Theorem** – 4: The non-switching discrete-time sliding mode controller for system (6.3, 6.4) in the presence of random fractional network delays satisfying (6.5) with packet loss and matched uncertainty d(k) is given as,

$$u(k) = -(C_s G)^{-1} (1 - \bar{\alpha})^{-1} [Hx(k) - Ix(k) + Kx(k) - Lx(k - 1) - (6.24)]$$
$$J(s(k)) + d_s(k) - d_1] - d(k).$$

where,

$$H = (1 - \bar{\alpha})(C_s F), I = \varsigma(1 - \bar{\alpha})C_s, K = \bar{\alpha}C_s, L = \varsigma\bar{\alpha}C_s \text{ and } J = \{1 - q[s(k)]\}$$

**Proof**: Let us define reaching law given by Bartoszewicz and Lesniewski (2014) in the presence of random fractional delay as:

$$s[(k+1)h] = \{1 - q[s(k)]\} - d_s(k) + d_1,$$
(6.25)

where  $q[s(k)] = \frac{\psi}{\psi + |s(k)|}$  with  $\psi$  as user defined constant satisfying  $\psi \ge d_2$ ,  $d_1$  and  $d_2$  are mean and deviation of d(k).

**Remark** – 6: The disturbace d(k) appearing in the reaching law is applied through the network. So, the compensated disturbance  $d_s(k)$  using Thiran's approximation is given as:

$$d_s(k) = d(k) - \varsigma d(k-1),$$
(6.26)

The reaching law in Eqn. (6.25) indicates that the system states always move towards the specified sliding band given as:

$$|s(k)| \le \frac{\psi d_2}{\psi - d_2},$$
 (6.27)

Substituting the value of s(k + 1) in Eqn. (6.25) we get,

$$(1 - \bar{\alpha})x'_c(k+1) + \bar{\alpha}x'_c(k) = \{1 - q[s(k)]\} - d_s(k) + d_1,$$

Substituting the value of  $x'_c(k+1)$ ,

$$(1 - \bar{\alpha})[C_s x(k+1) - \varsigma C_s x(k)] + \bar{\alpha}[C_s x(k) - \varsigma C_s x(k-1)] =$$

$$\{1 - q[s(k)]\} - d_s(k) + d_1,$$
(6.28)

Further, substituting x(k+1) gives

$$(1 - \bar{\alpha})[C_s[Fx(k) + G(u(k) + d(k))] - \varsigma C_s x(k)] + \bar{\alpha}[C_s x(k) - (6.29)]$$
$$\varsigma C_s x(k-1)] = \{1 - q[s(k)]\} - d_s(k) + d_1,$$

Simplifying,

$$(1 - \bar{\alpha})C_sFx(k) + (1 - \bar{\alpha})C_sG(u(k) + d(k)) - \varsigma(1 - \bar{\alpha})C_sx(k) + (6.30)$$
$$\bar{\alpha}C_sx(k) - \varsigma\bar{\alpha}C_sx(k-1) = \{1 - q[s(k)]\} - d_s(k) + d_1,$$

Further solving above Eqn. (6.30), the control law can be expressed as:

$$u(k) = -(C_s G)^{-1} (1 - \bar{\alpha})^{-1} [Hx(k) - Ix(k) + Kx(k) - Lx(k - 1) - (6.31)]$$
$$J(s(k)) + d_s(k) - d_1] - d(k).$$

where,

$$H = (1 - \bar{\alpha})(C_s F), I = \varsigma(1 - \bar{\alpha})C_s, K = \bar{\alpha}C_s, L = \varsigma\bar{\alpha}C_s \text{ and } J = \{1 - q[s(k)]\}$$

This completes the proof.

Similarly, the effect of random fractional delay at controller to actuator and packet loss can be compensated at the actuator end. The compensated control signal applied to the plant is given by:

$$u_a(k) = (1 - \bar{\beta})(u_c(k)) + \bar{\beta}(u_c(k-1)), \tag{6.32}$$

where,

 $u_c(k) = u(k) - \gamma' u(k-1)$  and  $u_c(k-1) = u(k-1) - \gamma' u(k-2)$  and  $\gamma' = \frac{\hat{\tau}_{ca}}{1+\hat{\tau}_{ca}}$  and  $\hat{\tau}_{ca}$  is the random fractional delay from controller to actuator.

From Eqn. (6.32) it can be easily inferred that at each sampling instant the effect of random fractional delay from controller to actuator is compensated in  $u_c(k)$  using past control signal, present control signal and parameter ' $\gamma$ '' while packet loss is compensated which takes the value  $u_c(k-1)$  with probability  $\overline{\beta}$  and  $u_c(k)$  with probability  $(1-\overline{\beta})$ .

The next section discusses about the stability condition for the closed loop system such that the system states remain within specified band (6.27) using control law (6.31).

#### 6.4.1 Stability Analysis

The trajectories of the closed loop system given in Eqn. (6.3,6.4) drive towards the sliding surface as mentioned in Eqn. (6.23) for a given controller in Eqn. (6.31) in the presence of total networked delay  $\hat{\tau}$  satisfying Eqn. (6.5), packet loss condition

satisfying  $0 \leq \bar{\alpha} < 1$  and matched uncertainty satisfying (3.7) such that the following condition in Eqn. (6.33) must exists:

$$\rho s^T(k)s(k) \succ 0. \tag{6.33}$$

Proof: Consider the compensated sliding surface (6.23),

$$s(k) = (1 - \bar{\alpha})x'_{c}(k) + \bar{\alpha}x'_{c}(k - 1).$$
(6.34)

Let us defined Lyapunov function as,

$$V_s(k) = s^T(k)s(k).$$
 (6.35)

Taking the forward difference we have,

$$\Delta V_s(k) = s^T(k+1)s(k+1) - s^T(k)s(k).$$
(6.36)

Using Eqn. (6.34) we get,

$$\Delta V_s(k) = [(1 - \bar{\alpha})x'_c(k+1) + \bar{\alpha}x'_c(k)]^T [(1 - \bar{\alpha})x'_c(k+1) + \bar{\alpha}x'_c(k)] - s^T(k)s(k).$$
(6.37)

Substituting the value of  $x'_c(k+1)$  we get,

$$\Delta V_s(k) = [(1 - \bar{\alpha})[C_s x(k+1) - \alpha' C_s x(k)] + \bar{\alpha}[C_s x(k) - \alpha' C_s x(k-1)]]^T (6.38)$$
$$[(1 - \bar{\alpha})[C_s x(k+1) - \alpha' C_s x(k)] + \bar{\alpha}[C_s x(k) - \alpha' C_s x(k-1)]] - s^T(k)s(k).$$

Further substituting the value of x(k+1),

$$\Delta V_s(k) = [(1 - \bar{\alpha})[C_s[Fx(k) + G(u(k) + d(k))] - \alpha' C_s x(k)] + \bar{\alpha}[C_s x(k) - (6.39)] \alpha' C_s x(k-1)]^T [(1 - \bar{\alpha})[C_s[Fx(k) + G(u(k) + d(k))] - \alpha' C_s x(k)] + \bar{\alpha}[C_s x(k) - \alpha' C_s x(k-1)]] - s^T(k)s(k).$$

Substituting u(k) and rewritting above Eqn. (6.39),

$$\Delta V_s(k) = \Gamma - s^T(k)s(k). \tag{6.40}$$

where,  $\Gamma = [(1 - \bar{\alpha})^{-1}[1 - q(s(k))]s(k) - d_s(k) + d_1]^T[(1 - \bar{\alpha})^{-1}[1 - q(s(k))]s(k) - d_s(k) + d_1]$ . The term  $\Gamma$  can be tuned closed to zero by appropriately selecting the parameter  $\psi$  and  $\bar{\alpha}$ . If  $\Gamma$  is close to zero then  $s^T(k)s(k)$  will be larger than  $\Gamma$ . Thus for any small parameter  $\rho$  we have,

$$\Gamma - s^T(k)s(k) \prec \rho s^T(k)s(k). \tag{6.41}$$

Tuning of parameter  $\Gamma$ , leads to,  $\Delta V_s(k) \prec \rho s^T(k) s(k)$  which guarantees the convergence of  $\Delta V_s(k)$  and implies that any trajectory of the system (6.3,6.4) will be driven onto the sliding surface and maintain on it.

This completes the proof.

#### 6.5 **Results and Discussions**

In this section the effect of proposed control algorithm is validated through simulation and experimental results carried out in the presence of random fractional delay, packet loss and matched uncertainty. An illustrative example given by Wu and Chen (2007) is considered for simulation while Quanser DC motor is used for experimental purpose.

#### 6.5.1 Simulation Results

Consider the continuous-time LTI system as mentioned in (4.4.1),

$$\dot{x}(t) = Ax(t) + Bu(t - \tau_r) + Dd(t),$$
 (6.42)

$$y(t) = Cx(t), \tag{6.43}$$

where,  $A = \begin{bmatrix} -0.7 & 2 \\ 0 & -1.5 \end{bmatrix}, B = \begin{bmatrix} -0.03 \\ -1 \end{bmatrix},$   $C = \begin{bmatrix} 1 & 0 \end{bmatrix}, D = \begin{bmatrix} 1 \\ 1 \end{bmatrix}.$  Discretizing the above system with sampling interval of h = 30msec we get,

$$x(k+1) = Fx(k) + Gu(k - \hat{\tau}) + d(k), \qquad (6.44)$$

$$y(k) = Cx(k), \tag{6.45}$$

where,  

$$F = \begin{bmatrix} 0.9792 & 0.05805 \\ 0 & 0.956 \end{bmatrix}, G = \begin{bmatrix} -0.001771 \\ -0.02934 \end{bmatrix},$$

$$C = \begin{bmatrix} 1 & 0 \end{bmatrix}.$$



Figure 6.2: Total networked fractional delay  $\tau$ 



Figure 6.3: Magnified plot for sensor to controller fractional delay



Figure 6.4: Magnified plot for controller to actuator fractional delay



Figure 6.5: State variable  $x_1(k)$ 



Figure 6.6: Magnified state variable  $x_1(k)$ 



Figure 6.7: State variable  $x_2(k)$ 



Figure 6.8: Magnified state variable  $x_2(k)$ 

Figures (6.2) to (6.22) show the nature of the system under networked environment with random fractional delays, packet loss situation and matched uncertainty. In order to prove the robustness of the proposed algorithm in the presence of variable networked delay and packet loss the slow time varying disturbance is applied to the system as shown in Figure (4.1). The random nature of total networked induced fractional



Figure 6.9: Compensated sliding variable s(k)



Figure 6.10: Magnified compensated sliding variable s(k)



Figure 6.11: Control Signal u(k)



Figure 6.12: Magnified control signal u(k)

delay is modelled using Poisson's distribution. It is assumed that at every sampling instant one event is generated considering total networked induced delay lesser than sampling interval with zero trial. So, according to Poisson's distribution under these assumptions the probability of networked delays lesser than sampling interval will be p = 0.63 while the probability of the networked delays greater than sampling interval



Figure 6.13: Compensated control signal  $u_a(k)$ 



Figure 6.14: Magnified compensated control signal  $u_a(k)$ 



Figure 6.15: Compensated control signal  $u_a(k)$  with 10% packet loss



Figure 6.16: Magnified compensated control signal  $u_a(k)$  with 10% packet loss

is 1 - p = 0.37. Thus, the total network induced fractional delay generated within the network is  $0.003sec \le \tau' \le 0.055sec$  respectively. Figure (6.2) depict the random nature of total networked induced fractional delays modelled using Poisson's distribution. Figures (6.3) and (6.4) show the magnified results of sensor to controller fractional delay and controller to actuator fractional delay. As discussed in previous chapter, the


Figure 6.17: Compensated control signal  $u_a(k)$  with 30% packet loss



Figure 6.18: Magnified compensated control signal  $u_a(k)$  with 30% packet loss



Figure 6.19: Compensated control signal  $u_a(k)$  with 50% packet loss



Figure 6.20: Magnified compensated control signal  $u_a(k)$  with 50% packet loss

processing delays and computational delays ocuuring at sensor, actuator and controller are neglected due to its negligible effect.

The sliding gain  $C_s$  is calculated using discrete LQR method with Q = diag(1500, 1000)and R = 1. The computed values of sliding gain comes out to be  $C_s = [-1.577 -2.456]$ . The Quasi-sliding mode band comes out to be  $|s(k)| \leq +0.1$  to -0.1 with proper se-



Figure 6.21: Compensated control signal  $u_a(k)$  with fractional delay greater than sampling interval



Figure 6.22: Magnified compensated control signal  $u_a(k)$  with fractional delay greater than sampling interval

lection of user defined constant  $\psi = 10$ .

The simulation results are discussed in three parts: (i) Figures (6.5) to (6.14) discuss the effect of proposed control algorithm for specified networked delay range under single packet transmission (ii) Figures (6.15) to (6.20) describes the effect of fractional delays on compensated control signal for different packet loss situations and (iii) Figures (6.21) and (6.22) show the results of compensated control signal computed at actuator side when fractional delays are greater than the sampling interval.

Figures (6.5) and (6.7) show the results of system state variables with initial condition  $x(k) = \begin{bmatrix} 5 & -5 \end{bmatrix}$ . It can be noticed that both the state variables slide towards the origin from given initial condition in the presence of random fractional delays. In order to show the accurate effect of random fractional delay compensation at the output, results are magnified as shown in Figures (6.6) and (6.8) respectively. It can be observed that in both the cases the state variables are computed from initial sampling. Thus, the effect of random fractional delay from sensor to controller and controller to actuator is compensated using the proposed algorithm. The same effect of compensation is observed in compensated sliding variable (Figure (6.9)), control signal (Figure (6.11)) and compensated control signal (Figure (6.13)) respectively. On observation of the magnified results in Figures (6.10), (6.12) and (6.14) it can be noticed that all the three parameters are computed from first sampling instant even in the presence of sensor to controller

delay and controller to actuator delay. The amount of delays generated at both sides of network at that sampling instant are indicated in Figures (6.3) and (6.4) respectively. Thus the effects of random fractional delay from sensor to controller are compensated in the sliding surface and remains within the specified band (6.27) while controller to actuator random fractional delay is compensated at the actuator end. The efficacy of the proposed algorithm was further tested under packet loss situation for different probabilities. Figures (6.15), (6.17) and (6.19) show the results of compensated control signal for different values of  $\bar{\alpha}=0.10, 0.30$  and 0.50 respectively. It can be noticed that when the packet loss probability within the network is 50% the system shows unacceptable response with high frequency oscillations. Thus it can be noticed that when the packet loss is generated within the network the robust terms will generate more action to stabilize the system which in turn makes the compensated control signal oscillatory in nature. Figures (6.16), (6.18) and (6.20) show the magnified results of the compensated control signal for a given set of packet loss probabilities. It can be observed that when the packet loss probability increases to 50% within the network the proposed algorithm cannot compensate the effect of fractional delay as compared to the cases of 10% and 30% packet losses. Figures (6.21) and (6.22) show the nature of compensated control signal when the probabilities of the random fractional delays are greater than the sampling interval. It can be observed that when the probabilities of network fractional delays are reversed than previous case that is  $p\,=\,0.37$  and  $1-p\,=\,0.63$ the proposed technique cannot compensate the effect of random fractional delays and the system response becomes unstable with high frequency oscillations generated at the output. Thus, from above results it can be concluded that the proposed technique works efficiently with random networked delay of  $0.003 sec \leq \hat{\tau} \leq 0.055 sec$  in simulated environment. The proposed controller compensates the effect of random networked delay for  $\psi = 10$  and  $\bar{\alpha} = 0.3$  satisfying Eqn. (6.5) and shows the stable response satisfying condition mention in Eqn. (6.33) in the presence of random fractional delays, packet loss and matched uncertainty.

#### 6.5.2 Experimental Results

The state space form of DC Motor plant mentioned in (3.6.1) using Eqn. (3.34) is given as,

$$\dot{x}(t) = Ax(t) + Bu(t - \tau_r) + Dd(t),$$
(6.46)

$$y(t) = Cx(t),$$
 (6.47)

where,

$$A = \begin{bmatrix} -201 & 0 \\ 1 & 0 \end{bmatrix}, B = \begin{bmatrix} 1 \\ 0 \end{bmatrix},$$
$$C = \begin{bmatrix} 0 & 1 \end{bmatrix}, D = \begin{bmatrix} 1 \\ 1 \end{bmatrix}.$$

Discretizing the above system with sampling interval of h = 30msec we get,

$$x(k+1) = Fx(k) + Gu(k - \hat{\tau}) + d(k), \tag{6.48}$$

$$y(k) = Cx(k), (6.49)$$

where,

$$F = \begin{bmatrix} 0.001836 & 0 \\ 0.004573 & 1 \end{bmatrix}, G = \begin{bmatrix} -0.004753 \\ -0.0001242 \end{bmatrix},$$
$$C = \begin{bmatrix} 0 & 1 \end{bmatrix}.$$



Figure 6.23: Results of position control of DC Motor with 10% packet loss

In this section, experimental results are briefly discussed with DC motor as a plant in the presence of random fractional delays, packet loss and matched uncertainty situations. The position of DC motor is considered as a reference signal. The variable



Figure 6.24: Result of magnified position control of DC Motor with 10% packet loss



Figure 6.25: Result of compensated sliding variable with 10% packet loss



Figure 6.26: Magnified compensated sliding variable with 10% packet loss



Figure 6.27: Control signal with 10% packet loss



Figure 6.28: Magnified control signal with 10% packet loss

fractional delays are computed using Poisson's distribution considering the same assumptions as mentioned in the simulation results section. The values of sliding gain parameter  $C_s$ , sliding band |s(k)| and user defined constant  $\psi$  are same in order to study the effect of proposed control algorithm on the real time system. In simulation section



Figure 6.29: Compensated control signal  $u_a(k)$  with 10% packet loss



Figure 6.30: Magnified compensated control signal  $u_a(k)$  with 10% packet loss



Figure 6.31: Compensated control signal  $u_a(k)$  with 30% packet loss



Figure 6.32: Magnified compensated control signal  $u_a(k)$  with 30% packet loss



Figure 6.33: Tracking Response with network delays greater than sampling interval

the effects of control algorithm are well explained under two different conditions (i) in the absence of packet loss and (ii) in the presence of packet loss. But, when any system is connected to real time networks there are very few chances of secure communication



Figure 6.34: Magnified tracking response with network delays greater than sampling interval

between the plant and controller in the terms of data transfer. Some of the data packets will be lost due to various reasons such as jitter, congestion or traffic problems within the network. So, in real time application it is essential to study the effect of control algorithm in the presence of packet loss and random time delay. Figures (6.23) to (6.34)shows the nature of the DC motor plant in terms of reference tracking, compensated sliding variable, control signal and compensated control signal for specified random fractional delays shown in Figure (6.2), matched uncertainty shown in Figure (4.1) with probability of single packet loss  $\bar{\alpha} = 0.1$ . Figure (6.23) shows the tracking response of the DC motor. It can be observed that the position of DC motor is controlled accurately in the presence of random fractional delays and packet loss situation. In order to show the effect of time delay compensation the magnified tracking result is shown in Figure (6.24). It can be noticed that the effect of total network fractional delay is compensated as the output tracks the reference signal at first sampling instant. The same effect of fractional delay compensation is observed in compensated sliding variable and control signal as shown in Figures (6.25) and (6.27) respectively. Observing the magnified results shown in Figures (6.26) and (6.28), it can be noticed that both the parameters are computed from first sampling instant. Thus the effect of random fractional delay from sensor to controller is compensated at the sliding surface. The magnified result of random fractional delay from sensor to controller is shown in Figure (6.3). Figure (6.29) shows the nature of compensated control signal. It can be noticed that the effect of random fractional delay from controller to actuator is compensated at the actuator side and converges to zero rapidly without increasing the amplitude. The magnified result of the same in Figure (6.30). The magnified result of controller to actuator delay is shown in Figure (6.4). The efficacy of the proposed algorithm was further tested by increasing the packet loss probability by three times that is,  $\bar{\alpha} = 0.3$  for specified network delays and matched uncertainty. Observing the results of compensated control signal as shown in Figures (6.31) and (6.32), it can be noticed that the effect of fractional delay at the actuator side is still compensated with packet loss probability of 30%. Figure (6.33) shows the nature of tracking response when random fractional delay is greater than sampling interval. It can be noticed that the performance of the system goes to unstable condition and does not compensate the effect of random fractional delays. The magnified result of the same is shown in Figure (6.34).

Thus from above implementation results, it can be noticed that the proposed control algorithm proves to be robust and efficient controller as it shows the stable response satisfying condition (6.33) and compensates the effect of specified random fractional delays satisfying (6.5) in presence of packet loss and matched uncertainty.

### 6.6 Conclusion

In this chapter, we proposed the discrete-time SMC (Non-Switching) algorithm in the presence of random fractional delay and packet loss in the presence of uncertainty. The random fractional delay is modelled using Poisson's distribution and packet loss is modelled using Bernoulli's distribution function. The random fractional delay in forward and feedback channels are compensated by Thiran Approximation in the sliding surface and actuator end respectively. A novel sliding surface is designed using Thiran's Approximation. A non-switching type discrete-time sliding mode controller is designed such that system states slide along the proposed compensated surface and maintain within the specified band. The stability condition of closed loop NCSs is derived using Lyapunov approach that ensures finite time convergence of system states in presence of network non-idealities. The effectiveness of the proposed algorithm is examined under different possible conditions through illustrative example as well as real time plant. The results proved that the control law derived using Thiran's Approximation fractional delay precisely even in the presence of 30% packet loss as well as networked delay having values greater than sampling interval.

# **CHAPTER 7**

# DISCRETE-TIME NETWORKED SLIDING MODE CONTROL (DNSMC) WITH MULPTIPLE PACKET TRANSMISSION POLICY

#### 7.1 Introduction

In communication domain, generally the data transfer between two devices takes place in the form of frames or packets. It mainly depends upon the distance of communication. When the distances are shorter the data are transmitted in the form of frames. However, when the distances are longer the same frame is breakdown in the form of small packets in order to have secure and reliable communication. In NCSs the data transfer takes place over the longer distance. So, those data in the form of frames are converted into small packets which are defined as multiple packets. When such packets are lost during transmission it is defined as multiple packet loss. It is necessary to study the effect of such lost packets during transmission. In this chapter, a mathematical model is derived for multiple packet loss using probability function approach. The random time delay and multiple packet loss are treated separately in order to study the outcome of both parameters on the system. The random communication delay and multiple packet loss in the forward and feedback channel are compensated through Thiran approximation and probability distribution respectively. Based on the proposed approach the sliding surface is designed and discrete-time sliding mode control law is derived that computes the control actions in the presence of random network delay and multiple packet loss. The stability of the closed loop NCS is also derived using Lyapunov approach that assures the finite time convergence in the presence of network non-idealities. The efficacy of the proposed algorithm is examined through simulation and experimental results in the presence of random communication fractional delays and multiple packet loss.

## 7.2 **Problem Formulation**

Figure (7.1) depicts the block diagram of networked control system with multiple packets transmission. It can be observed that the state information and control information available through sensor and controller are splitted in the form of small packets and transmitted through the network. These transmitted packets will suffer from sensor to controller random delay and controller to actuator random delay. During multiple packets transmission it is necessary to consider three different situations (i) none of the packets are lost (ii) any one or more than one packet is lost or (iii) all the packets are lost. If second or third situation arises during transmission the frame structure would be incomplete at the controller side and the false control actions will be generated which affects the performance of the system. The multiple packets received at the controller side and actuator side are converted in the form of frames having fixed length and after processing in the form of small packets they are transmitted back to the networks.



Figure 7.1: Block diagram of NCS with multiple packets transmission

In order to study the effect of multiple packet loss under random time delay compensation on the system it is necessary to developed the mathematical model of multiple packet loss in the forward and feedback channel. The next segment describes the mathematical model of multiple packet loss that is generated from sensor to controller and controller to actuator. The concept of probability distribution function is used to modelled multipe packet loss.

#### Multiple Packet Loss Model From Sensor to Controller and Controller to Actuator

As discussed earlier, in mutiple packet loss model the frames are splitted in the form of small packets. Let, the plants states are splitted into r equal parts given as:

$$x(k) = \begin{bmatrix} x_1^T(k) & x_2^T(k) & \dots & x_r^T(k) \end{bmatrix}^T$$
(7.1)

and every part of plant state is lumped into packet.

Similarly, let the control signal is splitted into *s* equal parts given as:

$$u(k) = \begin{bmatrix} u_1^T(k) & u_2^T(k) & \dots & u_s^T(k), \end{bmatrix}^T$$
(7.2)

and every part of control state is lumped into packet. where  $x_i(k)\epsilon R^{r_i}$ ,  $r_i\epsilon Z^+$  and  $\Sigma_{i=1}^r = n$  and  $u_i(k)\epsilon R^{s_i}$ ,  $s_i\epsilon Z^+$  and  $\Sigma_{i=1}^s = p$ .

For, simplicity we consider the case that the plant states and control signals are splitted



Figure 7.2: Schematic Diagram of NCSs with state data frame splitted in two parts

in two parts. Figure (7.2) describes the schematic diagram of networked control system with multiple packet transmission with state data frames splitted in two equal parts. As shown in Figure (7.2) two different cases are consider on each side of the network medium in form of switch position  $S_1$  to  $S_6$ . When the switch position is at  $S_3$  or  $S_6$ none of the packets will be lost or either all the packet will be lost at the controller and actuator side but when the switch is at position  $S_1$  or  $S_2$  and  $S_4$  or  $S_5$  anyone packet is lost on either side of the channel. As shown in Figure (7.2) at sampling instant h, the state data packets available at the input of the controller will be  $Xc_1(k)$  and  $Xc_2(k)$ respectively. Thus the process of state data transmission can be described as follows:

**Case** – I: If the switch is in position  $S_3$  then,

$$Xc_1(k) = \begin{cases} x_1(k-1); & \text{if } r_1 \prec P_{loss1} \\ x_1(k); & \text{if otherwise} \end{cases}$$
$$Xc_2(k) = \begin{cases} x_2(k-1); & \text{if } r_2 \prec P_{loss2} \\ x_2(k); & \text{if otherwise} \end{cases}$$

**Case** – **II**: If the switch is in position  $S_2$  then,  $Xc_1(k) = x_1(k)$ 

$$Xc_2(k) = \begin{cases} x_2(k-1); & \text{if } r_2 \prec P_{loss2} \\ x_2(k); & \text{if otherwise} \end{cases}$$

 $\mathbf{Case} - \mathbf{III}: \text{ If the switch is in position } S_1 \text{ then,}$  $Xc_1(k) = \begin{cases} x_1(k-1); & \text{if } r_1 \prec P_{loss1} \\ x_1(k); & \text{if otherwise} \end{cases}$ 

$$Xc_2(k) = x_2(k)$$

where,  $0 \prec P_{loss1} \prec 1$  and  $0 \prec P_{loss2} \prec 1$  is the probability of multiple state packet loss over the network and  $r_1, r_2$  are the random variables uniformly distributed over the interval [0,1].

It is also assumed that the measured outputs, which are equivalent to states of the system are sent to the controller via communication channel. As a result, the measurement may involve randomly varying communication delays. The communicated state variable over the network having the random delays from sensor to controller is given by:

$$x_c(k) = \begin{bmatrix} Xc_1(k - \hat{\tau}_{sc}) \\ Xc_2(k - \hat{\tau}_{sc}) \end{bmatrix},$$

simplyfying it we get,

$$x_c(k) = Z_1 X c_1(k - \hat{\tau}_{sc}) + Z_2 X c_2(k - \hat{\tau}_{sc}),$$
(7.3)

where,

 $Z_1 = diag(I_{r_1}, 0)$  and  $Z_2 = diag(I_{r_2}, 0)$ .

Thus the generalized equation is given by:

$$x_c(k) = \sum_{i=0}^r Z_i x_i (k - \hat{\tau}_{sc}),$$
(7.4)

where,  $Z_i = diag(0, ..., I_{r_i}, ..., 0)$ ,  $x_c(k)$  is the communicated state variable over the network and  $\hat{\tau}_{sc}$  is sensor to controller random fractional delay.

Similarly, the process of control data transmission can be described as follows:

Case – IV: If the switch is in position  $S_6$  then,

$$U_1(k) = \begin{cases} u_1(k-1); & \text{if } r_3 \prec P_{loss3} \\ u_1(k); & \text{if otherwise} \end{cases}$$
$$U_2(k) = \begin{cases} u_2(k-1); & \text{if } r_4 \prec P_{loss4} \\ u_2(k); & \text{if otherwise} \end{cases}$$

Case – V: If the switch is in position  $S_5$  then,  $U_1(k) = u_1(k)$ 

$$U_2(k) = \begin{cases} u_2(k-1); & \text{if } r_4 \prec P_{loss4} \\ u_2(k); & \text{if otherwise} \end{cases}$$

 $\mathbf{Case} - \mathbf{VI}: \text{ If the switch is in position } S_4 \text{ then,}$  $U_1(k) = \begin{cases} u_1(k-1); & \text{if } r_3 \prec P_{loss3} \\ u_1(k); & \text{if otherwise} \end{cases}$ 

$$U_2(k) = u_2(k)$$

where,  $0 \prec P_{loss3} \prec 1$  and  $0 \prec P_{loss4} \prec 1$  is the probability of the multiple control packet loss over the network and  $r_3$ ,  $r_4$  are the random variable uniformly distributed over the interval [0,1].

The measured outputs, which are equivalent to control signal of the system are sent to the actuator via communication channel. As a result, the measurement may involve randomly varying communication delay. The communicated controlled variable over the network having the random delays from controller to actuator is given by:

$$u_a(k) = \begin{bmatrix} U_1(k - \hat{\tau}_{ca}) \\ U_2(k - \hat{\tau}_{ca}) \end{bmatrix},$$

Simplyfying it we get,

$$u_a(k) = D_1 U_1(k - \hat{\tau}_{ca}) + D_2 U_2(k - \hat{\tau}_{ca}),$$
(7.5)

where,

 $D_1 = diag(I_{s_1}, 0)$  and  $D_2 = diag(I_{s_2}, 0)$ .

Thus the generalized equation is given by:

$$u_a(k) = \sum_{j=0}^{s} D_j u_j (k - \hat{\tau}_{ca}),$$
(7.6)

where,  $D_j = diag(0, ..., I_{s_j}, ..., 0)$ ,  $u_a(k)$  is the control signal available at actuator side and  $\hat{\tau}_{ca}$  is controller to actuator random fractional delay.

**Remark** – 9: In this chapter nature of the system, total random fractional network delays  $\hat{\tau}$ , sensor to controller random fractional delay  $\hat{\tau}_{sc}$  and controller to actuator random fractional delay  $\hat{\tau}_{ca}$  would remain same as described in section (6.2).

Problem Statement: The main objective, is to design the discrete-time sliding mode control law for system (6.3,6.4) in the presence of random fractional delay  $\hat{\tau}_{sc}$  and  $\hat{\tau}_{ca}$  with multiple packet loss situation and matched uncertainty satisfying (6.5).

Once the mathematical model of multiple packet loss is derived the next step is to design the compensated sliding surface and discrete-time control law that compensates the effect of random fractional delays in the presence of multiple packet loss and matched uncertainty.

#### 7.3 Design of Sliding Surface With Multiple Packet Loss

This section describes the designing of compensated sliding surface under multiple packet loss presented in the form of Lemma -4.

Lemma -4: The compensated sliding surface for the given system (6.3, 6.4) with sensor to controller random fractional delay under multiple packet loss and matched

uncertainty satisfying condition (6.5) and (3.7) is given as:

$$s(k) = \sum_{i=0}^{r} [C_s Z_i x_i(k) - \varsigma C_s Z_i x_i(k-1)], \qquad (7.7)$$

where,  $C_s$  is the sliding gain,  $\varsigma$  is the parameter designed using Thiran Approximation. **Proof**: The sliding variable with sensor to controller random fractional delay and mutiple packet loss is given as:

$$s(k) = C_s x_c(k). ag{7.8}$$

Substituting  $x_c(k)$  from Eqn. (7.4) to Eqn. (7.8) we get,

$$s(k) = \sum_{i=0}^{r} C_s Z_i x_i (k - \hat{\tau}_{sc}),$$
(7.9)

where,  $\hat{\tau}_{sc}$  is random fractional delay from sensor to controller. The random fractional delay from sensor to controller  $(\hat{\tau}_{sc})$  is model using Poisson's distribution function defined in Eqns. (6.9) and (6.10) respectively. Applying *z*-transform to above Eqn. (7.9) we get,

$$s(k) = \sum_{i=0}^{r} C_s Z_i [x_i(z) z^{-\hat{\tau}_{sc}}].$$
(7.10)

Using Thiran Approximation and Eqn. (6.14),  $z^{-\hat{\tau}_{sc}}$  is given as

$$s(k) = \sum_{i=0}^{r} C_s Z_i x_i(z) [1 - \varsigma z^{-1}], \qquad (7.11)$$

where,  $\varsigma = \frac{\hat{\tau}_{sc}}{\hat{\tau}_{sc}+1}$ . Applying inverse *z*-transform to above Eqn. (7.11) we get,

$$s(k) = \sum_{i=0}^{r} C_s Z_i [x_i(k) - \varsigma x_i(k-1)].$$
(7.12)

Further simplification,

$$s(k) = \sum_{i=0}^{r} [C_s Z_i x_i(k) - \varsigma C_s Z_i x_i(k-1)].$$
(7.13)

This completes the proof.

The above Eqn. (7.13) represents compensated sliding surface with multiple packet

loss. It is noticed that at each sampling interval the effect of delayed sensor packets are compensated through immediate past state data packets, current state data packets and parameter  $\varsigma$ . While the effect of multiple packet loss is compensated through probability distribution function. If random variable r is greater than packet loss probabilities then the data packet  $x_i(k)$  is received and, if random variable r is lesser than packet loss probabilities then  $x_i(k-1)$  will be received at the controller.

The next section represents the design of discrete-time sliding mode control law along with stability analysis in the presence of multiple packet loss and matched uncertainty.

# 7.4 Discrete-Time Sliding Mode Control Law With Random Fractional Delay and Multiple Packet Loss

This section discusses the derivation of discrete-time sliding mode control law for NCS using sliding surface (7.13) in the form of Theorem -5.

**Theorem** – 5: The discrete-time sliding mode control law for system (6.3,6.4) in the presence of sensor to controller random fractional delay  $(\hat{\tau}_{sc})$ , multiple packet loss and matched uncertainty d(k) satisfying (3.7) is given as,

$$u(k) = -(C_s G')^{-1} [Hx_i(k) - Ix_i(k) - J(s(k)) + d_s(k) - d_1] - d(k).$$
(7.14)

where,

$$G' = \sum_{i=0}^{r} (Z_i G), H = \sum_{i=0}^{r} (C_s Z_i F), I = \sum_{i=0}^{r} \alpha' C_s Z_i, \text{ and } J = \{1 - q[s(k)]\}$$

**Proof**: Let us define non-switching reaching law described in (6.25) considering the same parameters in the presence of network fractional delay as:

$$s[(k+1)h] = \{1 - q[s(k)]\} - d_s(k) + d_1.$$
(7.15)

The above all parameters such as  $\{q[s(k)]\}, d_s(k), d_1$ , condition of user defined constant  $\psi$  and sliding band |s(k)| remain same as defined in Eqns. ((6.25) to (6.27)) respectively.

Substituting the value of s(k + 1) in Eqn. (7.15) we get,

$$\sum_{i=0}^{r} C_s Z_i x_i(k+1) - \alpha' Z_i C_s x_i(k) = \{1 - q[s(k)]\} - d_s(k) + d_1.$$

Substituting the value of  $x_i(k+1)$ ,

$$\Sigma_{i=0}^{r} Z_{i} C_{s} [Fx_{i}(k) + G(u(k) + d(k))] - \alpha' Z_{i} C_{s} x_{i}(k) =$$

$$\{1 - q[s(k)]\} - d_{s}(k) + d_{1}.$$
(7.16)

On simplifying gives,

$$\Sigma_{i=0}^{r} Z_{i} C_{s} F x_{i}(k) + \Sigma_{i=0}^{r} Z_{i} C_{s} G(u(k) + d(k)) - \alpha' Z_{i} C_{s} x_{i}(k) =$$

$$\{1 - q[s(k)]\} - d_{s}(k) + d_{1}.$$
(7.17)

Further solving the above Eqn. (7.17) control law can be expressed as:

$$u(k) = -(C_s G')^{-1} [Hx_i(k) - Ix_i(k) - J(s(k)) + d_s(k) - d_1] - d(k).$$
(7.18)

where,

$$G' = \sum_{i=0}^{r} (Z_i G), H = \sum_{i=0}^{r} (C_s Z_i F), I = \sum_{i=0}^{r} \alpha' C_s Z_i, \text{ and } J = \{1 - q[s(k)]\}.$$

This completes the proof.

The control law defined in Eqn. (7.18) is transmitted to the network in the form of small packets. These packets will experience the controller to actuator fractional delay. Thus the mathematical model of multiple packets with controller to actuator fractional delay is given as:

$$u_a(k) = \sum_{j=0}^{s} D_j u_i(k - \hat{\tau}_{ca}), \tag{7.19}$$

where,  $D_j = diag(0, ..., I_{s_j}, ..., 0)$ ,  $u_a(k)$  is the control signal computed at actuator side and  $\hat{\tau}_{ca}$  is controller to actuator random fractional delay defined using Poisson's distribution.

Using Thiran Approximation the communicated control signal in Eqn. (7.19) can be

transformed to compensated control signal as,

$$u_a(k) = \sum_{j=0}^{s} D_j u_j(k) - \beta' D_j u_j(k-1),$$
(7.20)

 $\beta' = \frac{\hat{\tau}_{ca}}{1 + \hat{\tau}_{ca}}.$ 

From Eqn. (7.20), it can be noticed that at each sampling instant the effects of random packet loss and random network delay from controller to actuator are compensated in control signal through proposed technique.

The next section discusses about the stability of the closed loop system such that the system states remain within the specified band using control law (7.18).

#### 7.4.1 Stability Analysis

The trajectories of the closed loop system (6.3,6.4) with the controller mentioned in Eqn. (7.18) in the presence of random fractional delay  $(\hat{\tau})$  satisfying (6.26), multiple packet loss and matched uncertainty d(k) drive towards the sliding surface (7.13) such that the following condition holds true:

$$\eta s^T(k)s(k) \succ 0. \tag{7.21}$$

**Proof**: The compensated sliding surface in (7.13) is given by:

$$s(k) = \sum_{i=0}^{r} C_i C_s x_i(k) - \alpha' Z_i C_s x_i(k-1).$$
(7.22)

Let us consider Lyapunov function as,

$$V_s(k) = s^T(k)s(k).$$
 (7.23)

Taking forward difference we have,

$$\Delta V_s(k) = s^T(k+1)s(k+1) - s^T(k)s(k).$$
(7.24)

Using Eqn. (7.13) we get,

$$\Delta V_{s}(k) = [\Sigma_{i=0}^{r} Z_{i} C_{s} x_{i}(k+1) - \alpha' Z_{i} C_{s} x_{i}(k)]^{T} [\Sigma_{i=0}^{r} Z_{i}$$

$$C_{s} x_{i}(k) - \alpha' Z_{i} C_{s} x_{i}(k-1)] -$$

$$s^{T}(k) s(k).$$
(7.25)

Substituting the value of  $x_i(k+1)$ ,

$$\Delta V_s(k) = [\Sigma_{i=0}^r Z_i C_s [Fx_i(k) + G(u(k) + d(k))] - \alpha' Z_i C_s x_i(k)]^T [\Sigma_{i=0}^r Z_i C_s (7.26)]$$
$$[Fx_i(k) + G(u(k) + d(k))] - \alpha' Z_i C_s x_i(k-1)] - s^T(k) s(k).$$

Substituting the value of u(k) and further simplifying it we get,

$$\Delta V_s(k) = \kappa - s^T(k)s(k). \tag{7.27}$$

where,

 $\kappa = [[1 - q(s(k))]s(k) - d_s(k) + d_1]^T[[1 - q(s(k))]s(k) - d_s(k) + d_1].$  The term  $\kappa$  can be tuned closed to zero by appropriately selecting the parameter  $\psi$ . If  $\kappa$  is closed to zero then  $s^T(k)s(k)$  will be larger than  $\kappa$ . Thus for any small parameter  $\eta$  we have,

$$\Delta V_s(k) \prec \eta s^T(k) s(k). \tag{7.28}$$

Thus, by tuning the parameter  $\psi$ , we have,  $\Delta V_s(k) \prec \eta s^T(k)s(k)$  which guarantees the convergence of  $\Delta V_s(k)$  and implies that any trajectory of the system (6.3,6.4) will be driven onto the sliding surface and maintain on it.

This completes the proof.

#### 7.5 Results and Discussions

In this section the efficacy of the designed control algorithm is validated in the presence of multiple packet loss and matched uncertainty applied at the input channel of the system. The simulation results are carried out using illustrative example while implementation results are carried out on DC motor plant considering various possibilities of multiple packet loss.

### 7.5.1 Simulation Results

Consider the continuous-time LTI system as,

$$\dot{x}(t) = Ax(t) + Bu(t - \tau_r) + Dd(t),$$
(7.29)

$$y(t) = Cx(t), \tag{7.30}$$

where,

$$A = \begin{bmatrix} -0.7 & 2\\ 0 & -1.5 \end{bmatrix}, B = \begin{bmatrix} -0.03\\ -1 \end{bmatrix},$$
$$C = \begin{bmatrix} 1 & 0 \end{bmatrix}, D = \begin{bmatrix} 1\\ 1 \end{bmatrix}.$$

Discretizing the above system with sampling interval of h = 30msec we get,

$$x(k+1) = Fx(k) + Gu(k - \hat{\tau}) + d(k), \qquad (7.31)$$

$$y(k) = Cx(k), \tag{7.32}$$

where,

$$F = \begin{bmatrix} 0.9792 & 0.05805 \\ 0 & 0.956 \end{bmatrix}, G = \begin{bmatrix} -0.001771 \\ -0.02934 \end{bmatrix},$$
$$C = \begin{bmatrix} 1 & 0 \end{bmatrix}.$$



Figure 7.3: Total networked fractional delay



Figure 7.4: Magnified sensor to controller random fractional delay



Figure 7.5: Magnified controller to actuator random fractional delay





Figure 7.7: Magnified state variable  $x_1(k)$ 

Figures (7.6) to (7.21) shows the nature of the system in the presence of network non-idealities. In order to prove the robustness of the proposed control algorithm slow time varying disturbance is applied to the input of the system. Figure (7.3) shows the response of total network induced delay modelled using Poisson's distribution. It is assumed that the probability of the networked delay lesser than sampling interval is



Figure 7.8: State variable  $x_2(k)$ 



Figure 7.9: Magnified state variable  $x_2(k)$ 



Figure 7.10: Compensated sliding variable s(k)



Figure 7.11: Magnified compensated sliding variable s(k)

p = 0.67 while the probability of networked delay greater than sampling interval is p = 0.33. This assumption indicates that according to Poisson's distribution at every sampling interval at least one event is generated with zero trial. Thus, the specified network delay range computed through Poisson's distribution is  $0.003sec \le \tau \le 0.055sec$  respectively.



Figure 7.12: Control signal u(k)



Figure 7.13: Magnified control signal u(k)



Figure 7.14: Compensated control signal  $u_a(k)$ 



Figure 7.15: Magnified compensated control signal  $u_a(k)$ 

The sliding gain parameter computed using discrete LQR method with Q = diag(1000, 1000)and R = 1 is  $C_s = [-14.234 -20.625]$  having  $\psi = 10$ . The quasi-sliding mode band comes out to be  $|s(k)| \leq +0.1$  to -0.1.

Figures (7.6) to (7.15) show the nature of state variables, sliding surface, control signal computed at controller side and compensated control signal under multiple packet trans-



Figure 7.16: Compensated control signal  $u_a(k)$  with 10% packet loss



Figure 7.17: Magnified compensated control signal  $u_a(k)$  with 10% packet loss



Figure 7.18: Compensated control signal  $u_a(k)$  with 20% packet loss



Figure 7.19: Magnified compensated control signal  $u_a(k)$  with 20% packet loss

mission. Figures (7.6) and (7.8) show the nature of state variables  $[x_1 \ x_2]$  with initial conditions  $[5 \ -5]$  respectively. It is observed that both the state variables converge to zero from their specified initial condition in the presence of specified random fractional delay and multiple packets transmission. In order to determine the exact effect of random fractional delay the magnified results of the same are shown in Figures (7.7)



Figure 7.20: Compensated control signal  $u_a(k)$  with 30% packet loss



Figure 7.21: Magnified compensated control signal  $u_a(k)$  with 30% packet loss

and (7.9) respectively. It is noticed that both the state variables are computed from first sampling instant even in the presence of random fractional delay and multiple packet transmission in forward and feedback channel. The same effect of fractional delay compensation is observed in sliding variable Figure (7.10) and control signal Figure (7.12) computed at controller side. It is observed that both the parameters are computed from first sampling instant even in the presence of sensor to controller random fractional delay as shown in Figure (7.4). The magnified results of sliding variable and control signal computed at controller side are shown in Figures (7.11) and (7.13) respectively. Figure (7.14) shows the result of compensated control signal computed at the actuator end. Observing the magnified result in Figure (7.15), it is noticed that the effect of random fractional delay from controller to actuator is also compensated as it is computed from first sampling instant. Figure (7.5) shows the magnified result of controller to actuator random fractional delay.

The robustness of the proposed algorithm is further extended for multiple packet loss. The multiple packet loss is considered in both the channels simultaneously. It is assumed that more than one packet is lost on either side of the channel during transmission. Figures (7.16), (7.18) and (7.20) show the nature of compensated control signal computed at actuator side under 10%, 20% and 30% multiple packet loss respectively. It can be observed that the effect of random fractional delay from controller to actuator is compensated precisely as the controller signal is computed from first sampling instant

in the case of 10% and 20% packet loss. While in 30% packet loss the effect of random fractional delay is not compensated. The magnified results of the same are shown in Figures (7.17), (7.19) and (7.21) respectively.

Thus, the proposed control algorithm compensates the effect of random fractional delay and shows the stable response satisfying Eqn. (7.21) in the presence of multiple packet loss and matched uncertainty.

#### 7.5.2 Experimetal Results

In this section the proposed control algorithm is validated on Quanser DC Motor as a plant connected in networked medium. The position of the motor is controlled for the given reference input in the presence of mutiple packet transmission and matched uncertainty. The state space form of DC Motor plant of Eqn. (3.34) is given as,

$$\dot{x}(t) = Ax(t) + Bu(t - \tau_r) + Dd(t),$$
(7.33)

$$y(t) = Cx(t), \tag{7.34}$$

where,

$$A = \begin{bmatrix} -201 & 0 \\ 1 & 0 \end{bmatrix}, B = \begin{bmatrix} 1 \\ 0 \end{bmatrix}$$
$$C = \begin{bmatrix} 0 & 1 \end{bmatrix}, D = \begin{bmatrix} 1 \\ 1 \end{bmatrix}.$$

Discretizing the above system with sampling interval of h = 30msec we get,

$$x(k+1) = Fx(k) + Gu(k - \hat{\tau}) + d(k), \qquad (7.35)$$

$$y(k) = Cx(k), (7.36)$$

where,

$$F = \begin{bmatrix} 0.001836 & 0\\ 0.004573 & 1 \end{bmatrix}, G = \begin{bmatrix} -0.004753\\ -0.0001242 \end{bmatrix}$$
$$C = \begin{bmatrix} 0 & 1 \end{bmatrix}.$$



Figure 7.22: Position Control of DC Motor under multiple packet transmission



Figure 7.23: Magnified position control



Figure 7.24: Compensated sliding variable s(k)



Figure 7.25: Magnified compensated sliding variable s(k)



Figure 7.26: Control signal u(k)

Figures (7.22) to (7.35) show the nature of DC motor plant in the presence of random network fractional delay, multiple packet loss and matched uncertainty. The values of sliding gain  $C_s$ , sliding band |s(k)| and user defined constant  $\psi$  are same as



Figure 7.27: Magnified control signal u(k)



Figure 7.28: Compensated control signal  $u_a(k)$ 



Figure 7.29: Magnified compensated control signal  $u_a(k)$ 



Figure 7.30: Compensated control signal  $u_a(k)$  with 10% multiple packet losses



Figure 7.31: Magnified compensated control signal  $u_a(k)$  with 10% multiple packet loss

considered in simulation section. The results of position control, sliding variable, control signal computed at the controller side and compensated control signal computed at actuator side are carried out in the presence of specified network delay range of



Figure 7.32: Compensated control signal  $u_a(k)$  with 20% multiple packet losses



Figure 7.33: Magnified compensated control signal  $u_a(k)$  with 20% multiple packet losses



Figure 7.34: Compensated control signal  $u_a(k)$  with 30% multiple packet losses



Figure 7.35: Magnified compensated control signal  $u_a(k)$  with 30% multiple packet losses

 $0.003sec \leq \tau \leq 0.055sec$ . This delay range is computed through Poisson's distribution under same assumptions mentioned in simulation section. The slow time varying disturbance is applied to the input of the system defined as matched uncertainty.

The implementation results are carried out in two parts: (i) the first part (Figures (7.22) to (7.29)) describes the effect of compensation algorithm on the system under multiple packet transmission, (ii) while second part (Figures (7.30) to (7.35)) specifies the effect of random fractional delay on the control signal computed at the actuator side in the presence of multiple packet loss.

Figure (7.22) shows the nature of reference tracking response of DC motor plant in the presence of network non-idealities. It is noticed that the position of the DC motor is controlled according to change in the reference signal for the specified fractional delay and matched uncertainty under multiple packet transmission. In order to show the exact effect of compensation the magnified result of the same is shown in Figure (7.23). It is observed that the output signal responds the reference input at first sampling instant even in the presence of random fractional delay. The same effect of compensation algorithm is observed in sliding surface (Figure (7.24)), control signal computed at the controller side (Figure (7.26)) and compensated control signal computed at the actuator side (Figure (7.28)). It can be observed that all the three parameters slides towards the origin and remain within the specified band (6.27) over a finite interval of time. The magnified results of the same are shown in Figures (7.25), (7.27) and (7.29) respectively. Observing the magnified results it is noticed that all the three parameters are computed from initial sampling instant even in the presence of sensor to controller fractional delay.

Figures (7.30) to (7.35) show the nature of compensated control signal computed at actuator side under multiple packet loss. It is assumed that multiple packet loss takes place on either side of the network medium. So the actual effect of multiple packet loss can be studied at the actuator side rather than controller side. Figures (7.30), (7.32) and (7.34) shows the nature of compensated control signal at the actuator side with packet loss probabilities of 10%, 20% and 30% respectively. It is noticed that as the multiple packet loss probability increases the robust terms will generate more control actions to stabilize the system which in turn makes the system oscillatory in nature. The control signal shows the unacceptable response when the packet loss probability increases to 30%. Moreover, it is also observed from magnified results ((7.31), (7.33) and (7.35)) that the effect random fractional delay within the network is also not compensated as the packet loss is increased.

Thus, from results it can be noticed that with real time system the proposed algorithm shows the stable response satisfying (7.21) under 20% multiple packet loss for the specified random fractional delay and matched uncertainty.

# 7.6 Conclusion

In this chapter, time delay approximation algorithm is proposed in discrete-time domain that compensates the effect of variable fractional delay in the presence of mutiple packet loss condition. The random fractional delay is modelled using Poisson's distribution while multiple packet loss is modelled using probability distribution function. The multiple packet loss model is derived at the controller side as well as at the actuator side. The effects of random fractional delay as well as multiple packet loss are compensated in forward channel as well as feedback channel. The discrete-time sliding mode controller is designed based on the proposed sliding surface. The stability condition of the closed loop system is derived using Lyapunov approach such that the system states remain within the specified band over a finite interval of time. The efficacy of the proposed control algorithm is checked under different conditions through illustrative example and real-time plant. The simulation and experimental results shows that the discrete-time control law derived using Thiran's Approximation technique compensates the effect of random fractional delay even in the presence of multiple packet loss with probability of 20% and matched uncertainty.

## **CHAPTER 8**

# CONCLUSION, FUTURE SCOPE AND CHALLENGES

### 8.1 Conclusion and Future Scope

In this thesis, a novel idea of compensating the fractional delay in the sliding surface is introduced. The effect of fractional delay generated in NCS due to the presence of the communication medium is compensated using Thiran approximation technique. The sliding surface designed so that it slides along the predetermined surface according to fractional delays. Using this novel approach, a switching-type discrete-time networked sliding mode controller is designed which computes the control sequences in the presence of deterministic fractional delay and matched uncertainty. The stability of the closed loop system is assured by using Lyapunov approach. The efficacy of the proposed algorithm is tested on DC Servo motor setup with different network delays and external disturbances. The results were also compared with the conventional SMC. It is concluded that the fractional delay approximated using Thiran approximation is more efficient technique as it compensates the fractional networked delays then conventional discrete-time sliding mode control.

The major drawback of switching type discrete-time sliding mode controller is that it generates more chattering due to which the system generates oscillatory type of behaviour. So, to overcome this issue a non-switching type discrete-time sliding mode controller is designed such that system states slide along the proposed compensated surface and maintain within the specified band. The stability of the closed loop system is assured using Lyapunov approach through proposed control law. The efficacy of the proposed algorithm is tested through illustrative example as well as DC servo motor setup with different deterministic fractional delays and matched uncertainty. The results were also compared with switching type SMC as well as conventional SMC. It is concluded that non-switching type SMC provides faster convergence without increasing the amplitude of control signal and offers better fractional delay compensation

than switching-type SMC and conventional SMC. The efficacy of the proposed control algorithm is also tested under real-time networks using True-Time simulator. The performance of the control algorithm is checked using CAN and Switched Ethernet as a network medium in the presence of packet loss condition. It can be noticed from results that the control law derived using Thiran's Approximation compensates the effect of fractional network delay very precisely even in the presence of real-time networks and packet loss situation.

Later on, the concept of Thiran Approximation is used in Multi-rate Output feedback approach in which the sliding surface and control law are computed based on availability of the output information. The main advantage of using multirate output feedback approach is that the system states are computed based on the output information available and the error between computed as well as estimated state variables becomes zero exactly after one sampling instant. Using this novel approach a multirate output feedback discrete-time networked sliding mode control law is derived that compute the control sequences in the presence of deterministic fractional delay and matched uncertainty. The stability condition of closed loop NCSs is derived using Lyapunov approach that ensures finite time convergence of system states in presence of network non-idealities. The effectiveness of the proposed algorithm is examined under different possible conditions through illustrative example.

Further, the concept of Thiran Approximation is examined for random fractional delays with single packet loss situation. The random fractional delay is modelled using Poisson's distribution function and packet loss is modelled using Probability distribution function. The discrete-time sliding mode controller is designed using compensated sliding surface in the presence of random fractional delays, packet loss and matched uncertainty. The stability of the closed loop system is derived that ensures finite time convergence in the presence of random fractional delay, packet loss and matched uncertainty. The effectiveness of the proposed control algorithm is examined through DC servo motor setup under random fractional delay and packet loss. The results proved that the control law derived using Thiran's Approximation compensates random fractional delay accurately even in the presence of single packet loss with probability of 30% as well as networked delays having values greater than sampling interval.

Lastly, the same concept of Thiran Approximation is carried out for random fractional delay with multiple packet loss situations. The multiple packet loss situation is mod-

elled using Probability distribution function while random fractional delay is modelled using Poisson's distribution. The discrete-time sliding mode control law is derived that compensates the effect of random fractional delay with multiple packet loss using compensated sliding surface. The stability of the closed loop NCS is assured through Lyapunov approach under multiple packet transmission. The efficiency of proposed control algorithm is verified through DC servo motor plant under random fractional delay, multiple packet loss and matched uncertainty. The results proved that the proposed control law works efficiently and compensates the effect of random fractional delay even in the presence of multiple packet loss with probability of 30% as well as networked delays greater than sampling interval.

In future, the proposed control algorithm can be extended for Wireless Networked Control System (WNCS) as it possesses random time delay and packet loss. Morover the efficacy of the proposed algorithm is checked for direct structure NCS. The same can be extended for hierarchical structure and shared network structure NCS. The results proved that proposed control algorithm operates properly for the state feedback based controller in all different situations such as deterministic delay, random delay, single packet loss and multiple packet loss. In future, the same work can be extended for multi-rate output feedback controllers considering the same network non-idealities.

#### 8.2 Challenges

In Networked Control System although much work has been carried out, but still there are various challenges that has to be taken care while designing any control algorithm. NCS deals with various open research problems due to the presence of communication medium. Some of the challenges are:

- In some of the applications it might be possible that the delays are greater than the sampling interval. So, in such cases the control algorithm designed for time delay lesser than sampling interval will not able to work efficiently. Thus, in such cases there is a need for developing some robust technique which takes care of the networked delay having value greater than sampling interval.
- There are no standard algorithm available in the literatures that discusses about the variable packet loss model. However, in the existing models, network-induced delays and packet dropouts are lumped together. As a result, it is hard to distin-

guish their respective effects on NCSs. Thus, there is need for developing some model that takes care of only packet loss.

- In event-triggered control and filtering, it is often to assume that packet dropouts and packet disorders do not occur. This assumption is not practical when packets are transmitted through a communication network. How to deal with packet dropouts and packet disorders in event triggered control and filtering is challenging. Up to date, taking packet dropouts and packet disorders into account, no efficient approaches are proposed to design event triggered controllers and filters.
- Distributed networked control and distributed network based filtering are still attractive and challenging. Although some results on these issues are reported in the literature, they are usually based on some strong assumptions when parts of network constraints are taken into account. In fact, many practical factors need to be considered for distributed control and filtering in network environments. To mention a few, network-induced delays in different channels between an agent and its neighbours are different and time-varying, packet dropouts occur randomly in different communication channels, the topologies of the agents in a distributed system are not always the same at any time, and so on. These factors indeed make the analysis and synthesis of distributed networked control and distributed network-based filtering more complicated, especially for distributed systems with a large number of agents. (Zhang *et al.*, 2016)
- For industrial control and applications of NCSs, it is interesting to investigate possible positive effects of network-induced delays and packet dropouts on NCSs. There is no such algorithm designed that considers the positive effect of these parameters for industrial applications.

# REFERENCES

- 1. Argha, A., L. Li, W. Steven, and H. Nguyen (2015). Discrete-time sliding mode control for networked systems with random communication delays. *IEEE Proceedings of American Control Conference*.
- 2. Asif, S. and P. Webb (2015). Networked control system-an overview. *International Journal of Computer Applications*, **115**(6), 26–30.
- 3. Astrom, K., J. Apkarian, P. Karam, M. Levis, and J. Falcon, *Student Workbook: QNET DC Motor Control Trainer for NI ELVIS.* Quanser, 2015.
- 4. Bartolini, G., A. Ferrara, and V. Utkin (1995). Adaptive sliding mode control in discrete-time systems. *Automatica*, **32**(5), 773–796.
- 5. Bartoszewicz, A. (1996). Remarks on discrete-time variable structure control systems. *IEEE Transactions on Industrial Electronics*, **43**, 235–238.
- 6. **Bartoszewicz, A.** (1998). Discrete-time quasi-sliding-mode control strategies. *IEEE Transactions on Industrial Electronics*, **45**, 633–637.
- 7. Bartoszewicz, A. and P. Lesniewski (2014). Reaching law approach to the sliding mode control of periodic review inventory systems. *IEEE Transactions on Automatic Science and Engineering*, **11**(3), 810–817.
- 8. Cac, N., X. Hung, and V. Khang (2014). Can-based networked control systems: a compensation for communication time delays. *American Journal of Embedded Systems and Applications*, **2**(3), 13–20.
- Cervin, A., D. Henricksson, B. Lincoln, J. Eker, and K. Arzen (2003). Analysis and simulation of timing using jitterbug and true time. *IEEE on Control Systems Magazines*, 138(12), 1–12.
- 10. Chan, H. (1995). Closed loop control of a system over a communication network with queues. *International Journal of Control*, **62**(3), 493–501.
- 11. Chan, H. and U. Ozguner (1994). Closed-loop control of systems over a communication network with queues. *IEEE Proceeding of American Control Conference*, 811–815.
- 12. Cuellar, B., M. Villa, G. Anaya, O. Ramirez, and J. Ramirez (2007). Observerbased prediction scheme for time-lag processes. *IEEE Proceedings of American Control Conference*, 639–644.
- 13. **Dong, J.** and **W. Kim** (2012). Markov-chain-based output feedback control for stabilization of networked control systems with random time delays and packet losses. *International Journal of Control, Automation, and Systems*, **10**(5), 1013–1022.
- 14. **Drakunov, C., D. Izosimov, A. Loukianov, V. A. Utkin**, and **V. I. Utkin** (1990). Block control principle, i and ii. *Automation and Remote Control*, **51**(5), 601–609.
- 15. Edwards, C. and S. K. Spurgeon (1996). Robust output tracking using a sliding mode controller/observer scheme. *International Journal of Control*, **64**(5), 967–983.
- 16. Edwards, C. and S. K. Spurgeon, *Sliding Mode Control, Theory and Applications*. Systems and Control Book Series, Taylor and Francis Ltd, 1998.
- 17. Emelyanov, S. and S. Korovin (1981). Applying the principle of control by deviation to extend the set of possible feedback types. *Soviet Physics, Doklady*, **26**(6), 562–564.
- 18. Fallaha, C., M. Saad, H. Kanaan, and K. Haddad (2011). Sliding mode robot control with exponential reaching law. *IEEE Transactions on Industrial Electronics*, **58**(2), 600–610.
- 19. Furuta, K. (1990). Sliding model control of a discrete system. *Systems and Control Letters*, 14, 144–152.
- 20. Gao, H., T. Chen, and J. Lam (2007). A new delay system approach to network-based control. *Automatica*, 4(2008), 39–52.
- 21. Gao, W., Y. Wang, and A. Homaifa (1995). Discrete-time variable structure control systems. *IEEE Transactions on Industrial Electronics*, **42**(2), 117–122.
- 22. Gao, Z. (2013). Integral sliding mode controller for an uncertain network control system with delay. *Computer Engineering and Networking*, 277, 31–38.
- Ge, Y., Q. Chen, M. Jiang, and Y. Huang (2013). Modeling of random delays in networked control systems. *Journal of Control Science and Engineering*, 2013(2013), 1–9.
- 24. Godoy, E. and A. J. V. Porto (2010). Using simulation tools in the development of a networked control systems research platform. *ABCM Symposium Series in Mechatronics*, **4**(1), 384–393.
- 25. Gou, X. (2009). Controller design based on variable-period sampling approach for networked control systems with random delays. *IEEE Proceedings of International Conference on Networking, Sensing and Control*, 147–151.
- 26. Goyal, V., V. Deolia, and T. Sharma (2015). Robust sliding mode control for nonlinear discrete-time delayed systems based on neural network. *Intelligent Control and Automation*, **6**, 75–83.
- 27. Guo, P., J. Zhang, H. Karimi, Y. Liu, M. Lyu, and Y. Bo (2014). State estimation for wireless network control system with stochastic uncertainty and time delay based on sliding mode observer. *Abstract and Applied Analysis*, **2014**(2014), 1–8.
- 28. Gupta, R. and M. Chow (2010). Networked control system: overview and research trends. *IEEE Transactions on Industrial Electronics*, **57**(7), 2527–2535.
- 29. Gupta, V., B. Hassibi, and R. Murray (2007). Optimal lqg control across packetdropping links. *Systems and Control Letters, Elsevier*, **56**(2007), 439–446.
- 30. **Hespanha, J.** and **P. Naghshtabrizi** (2005). Designing an observer-based controller for a network control system. *IEEE Proceedings of Control and Decision Conference*, 848–853.

- 31. Hespanha, J., P. Naghshtabrizi, and Y. Xu (2007). A survey of recent results in networked control systems. *Proceedings of the IEEE*, **95**(1), 138–162.
- 32. Hikichi, Y., K. Sasaki, R. Tanaka, H. Shibasaki, K. Kawaguchi, and Y. Ishida (2013). A discrete pid control system using predictors and an observer for the influence of a time delay. *International Journal of Modelling and Optimization*, **3**(1), 1–4.
- 33. Hong, Z., J. Gao, and N. Wang (2014). Output-feedback controller design of a wireless networked control system with packet loss and time delay. *Mathematical Problems in Engineering*, 2014, 1–7.
- 34. Hu, J., J. Liang, H. Karimi, and J. Cao (2013). Sliding intermittent control for bam neural networks with delays. *Abstract and Applied Analysis*, **2013**(2013), 1–15.
- 35. Jacovitti, G. and G. Scarano (1993). Discrete time techniques for time delay estimation. *IEEE Transactions on Signal Processing*, **41**(2), 525–533.
- 36. Janardhanan, S. and B. Bandyopadhyay (2006). Output feedback sliding mode control for uncertain system using fast output sampling technique. *IEEE Transactions on Industrial Electronics*, **53**(1), 1677–1682.
- Janardhanan, S. and V. Kariwala (2008). Multirate-output feedback-based lq-optimal discrete-time sliding mode control. *IEEE Transactions on Automatic Control*, 53(1), 367–373.
- 38. **Ji, K.** and **W. Kim** (2005). Real-time control of networked control systems via ethernet. *International Journal of Control, Automation and Systems*, **3**(4), 591–600.
- 39. Jiang, X., Q. Han, and X. Yu (2005). Stability criteria for linear discrete-time systems with interval-like time-varying delay. *IEEE Proceeding of American Control Conference*, 2817–2822.
- 40. Jungers, M., E. Castelan, V. Moraes, and U. Moreno (2013). A dynamic output feedback controller for a network controlled system based on delay estimates. *Automatica, Elsevier*, **49**(3), 788–792.
- 41. Khanesar, M., O. Kaynak, S. Yin, and H. Gao (2015). Adaptive indirect fuzzy sliding mode controller for networked control systems subject to time varying network induced time delay. *IEEE Transactions on Fuzzy Systems*, **23**(1), 205–214.
- 42. Kim, D., J. Ko, and P. Park (2002). Stabilization of the asymmetric network control system using a deterministic switching system approach. *Proceedings of the 41st IEEE Conference on Decision and Control (CDC 02)*, 1638–1642.
- 43. Kokil, P., H. Kar, and V. Krishna (2011). Stability analysis of linear discrete-time systems with interval delay: A delay-partitioning approach. *International Scholarly Research Network Applied Mathematics*, **2011**, 1–10.
- 44. Li, H., H. Yang, F. Sun, and Y. Xia (2014). Sliding-mode predictive control of networked control systems under a multiple-packet transmission policy. *IEEE Transactions on Industrial Electronics*, **61**(11), 201–221.
- 45. Li, Y., S. Yang, Z. Zhang, and Q. Wang (2010). Network load minimisation design for dual-rate internet-based control systems. *IET Control Theory Application*, **4**(2), 197–205.

- 46. Ling, W., X. Ding-yu, and E. Da-zhi (2007). Some basic issues in networked control systems. *Second IEEE Conference on Industrial Electronics and Applications*, 2098–2112.
- 47. Liu, G., Y. Xia, J. Chen, D. Rees, and W. Hu (2007). Networked predictive control of systems with random networked delays in both forward and feedback channels. *IEEE Transactions on Industrial Electronics*, **54**(3), 1282–1296.
- 48. Luck, R. and A. Ray (1990*a*). Delay compensation in integrated communication and control systems: conceptual development and analysis. *Proceedings of American Control Conference (ACC90)*, 2045–2050.
- 49. Luck, R. and A. Ray (1990b). Delay compensation in integrated communication and control systems: implementation and verification. *Proceedings of American Control Conference (ACC90)*, 2051–2055.
- 50. Luck, R. and A. Ray (1990*c*). An observer-based compensator for distributed delays. *Automatica*, **26**(5), 903–908.
- 51. Luck, R. and A. Ray (1994). Experimental verification of a delay compensation algorithm for integrated communication and control systems. *International Journal of Control*, **59**(6), 1357–1372.
- 52. **Ma, D.** and **J. Zhao** (2006). Stabilization of networked control systems via switching controllers: an average dwell time approach. *IEEE Proceedings of World Congress on Intelligent Control*, 4619–4622.
- 53. **Mehta, A.** and **B. Bandyopadhyay** (2009). Frequency-shaped sliding mode control using output sampled measurements. *IEEE Transactions on Industrial Electronics*, **56**(1), 28–35.
- 54. Mehta, A. J. and B. Bandyopadhyay, *Frequency-Shaped and Observer-Based Discrete- Time sliding Mode Control*. SpringerBriefs in Applied Sciences and Technology, 2015.
- Mehta, A. J. and B. Bandyopadhyay (2016). Multirate output feedback based stochastic sliding mode control. *Journal of Dynamic Systems, Measurement, and Control*, 138(12), 124503(1–6).
- 56. **Milosavljevic, C.** (1985). General conditions for the existence of a quasisliding mode on the switching hyperplane in discrete variable structure systems. *Automatic Remote Control*, **46**, 307–314.
- 57. **Monsees, G.** (2002). *Discrete-Time Sliding Mode Control*. Ph.D. thesis, Technische Universiteit Delft, –.
- 58. Montestruque, L. A. and P. J. Antsaklis (2003). On the model-based control of networked systems. *Automatica*, **39**(10), 1837–1843.
- 59. Moudgalya, K. M., Digital Control. John Wiley and Sons Ltd, 2007.
- 60. **Mu, S., T. Chu, L. Wang**, and **W. Yu** (2004). Output feedback networked control system. *International Journal of Automation and Computing*, **1**(1), 26–34.

- 61. Negi, R., S. Purwar, and H. Kar (2012). Delay-dependent stability analysis of discrete time delay systems with actuator saturation. *Intelligent Control and Automation*, 2012(3), 34–43.
- 62. Nilsson, J., B. Bemhardson, and B. Wittenmark (1998). Stochastic analysis and control of real-time systems with random time delays. *Automatica*, **34**(1), 57–64.
- 63. Niu, Y. and D. Ho (2010). Design of sliding mode control subject to packet losses. *IEEE Transactions on Automatic Control*, **55**(11), 2623–2628.
- 64. **Ono, M., N. Ban, K. Sasaki, K. Matsumoto**, and **Y. Ishida** (2010). Discrete modified smith predictor for an unstable plant with dead time using a plant predictor. *International Journal of Computer Science and Network Security*, **10**(3), 80–85.
- 65. **Peng, C.** and **D. Yue** (2006). State feedback controller design of networked control systems with parameter uncertainty and state-delay. *Asian Journal of Control*, **8**(4), 385–392.
- 66. Qiu, J., G. Feng, and H. Gao (2013). Asynchronous output-feedback control of networked nonlinear systems with multiple packet dropouts: TŰs fuzzy affine model-based approach. *IEEE Transactions on Fuzzy Systems*, **19**(6), 1030–1040.
- 67. R.C.Dorf and R.H.Bishop, *Modern Control Systems*. Pearson Prentice Hall, 2008.
- 68. **Ridwan, W.** and **B. Trilaksono** (2011). Networked control synthesis using time delay approach:state feedback case. *International Journal on Electrical Engineering and Informatics*, **3**(4), 441–452.
- 69. Sabanovic, A., L. Fridman, and S. Spurgeon, *Variable Structure Systems from principles to implementation*. IET control engineering, 2004.
- Saravanakumar, R., M. S. Ali, C. Ahn, H. Karimi, and P. Shi (2016). Stability of markovian jump generalized neural networks with interval time-varying delays. *IEEE Transaction on Neural Network Learning System*, 99, 1–11.
- 71. Sarpturk, S., Y. Istefanopulos, and O. Kaynak (1987). On the stability of discretetime sliding mode control systems. *IEEE Transactions on Automatic Control*, **32**(10), 930–932.
- 72. Shi, Y. and B. Yu (2009). Output feedback stabilization of networked control systems with random delays modeled by markov chains. *IEEE Transactions on Automatic Control*, **54**(7), 1668–1674.
- 73. Shousong, H. and Z. Qixin (2003). Stochastic optimal control and analysis of stability of networked control systems with long delay. *Automatica*, **39**(2003), 1877–1884.
- 74. Song, H., S. Chen, and Y. Yam (2016). Sliding mode control for discrete-time systems with markovian packet dropouts. *IEEE Transactions on Cybernetics*, **99**, 1–11.
- 75. Su, W., S. V. Drakunov, and U. Ozguner (2000). An  $o(t^2)$  boundary layer in sliding mode for sampled-data systems. *IEEE Transactions on Automatic Control*, **45**(3), 482–485.

- 76. **Sun, Y.** and **J. Xu** (2009). Stability analysis and controller design for networked control systems with random time delay. *The Ninth International Conference on Electronic Measurement and Instruments*, 136–141.
- 77. **Tang, X.** and **J. Yu** (2010). Stability analysis of discrete-time systems with additive time-varying delays. *International Journal of Automation and Computing*, **7**(2), 219–223.
- 78. **Tascikaraoglu, F., I. Kucukdemiral**, and **J. Imura** (2014). Robust moving horizon h-inf control of discrete time- delayed systems with interval time-varying delays. *Mathematical Problems in Engineering*, **2014**, 1–13.
- 79. **Thiran, J.** (1971). Recursive digital filters with maximally flat group delay. *IEEE Transactions on Circuit Theory*, **18**(6), 659–664.
- 80. **Tipsuwan, Y.** and **M.-Y. Chow** (2004). Gain scheduler middleware: A methodology to enable existing controllers for networked control and teleoperation-part i: Networked control. *IEEE Transactions on Industrial Electronics*, **51**(6), 1218–1227.
- 81. Urban, M., M-Blaho, M. Murgas, and M. Foltin (2011). Simulation of networked control systems via truetime. *Technical Computing Prague*, **1**(1), 1–7.
- 82. Utkin, V., J. Guldner, and J. Shi, *Sliding Mode Control in Electromechanical Systems*. Taylor and Francis Ltd, 1999.
- 83. Utkin, V. I. (1993). Sliding mode control design principles and applications to electric drives. *IEEE Transactions on Industrial Electronics*, **40**(1), 23–36.
- 84. Vallabhan, M., S. Srinivasan, S. Ashok, S. Ramaswamy, and R. Ayyagari (2012). An analytical framework for analysis and design of networked control systems with random delays and packet losses. *IEEE Proceedings of Canadian Conference on Electrical and Computer Engineering*, 1–6.
- 85. Vardhan, S. and R. Kumar (2011). Simulations for time delay compensation in networked control systems. *Journal of Selected Areas in Telecommunications*, 1(1), 38–43.
- 86. Walsh, G., H. Ye, and L. Bushnell (2002). Stability analysis of networked control systems. *IEEE Transactions on Control Systems Technology*, **10**(3), 438–446.
- 87. Wang, B. (2008). *On Discretization of Sliding Mode Control Systems*. Ph.D. thesis, School of Electrical and Computer Engineering, RMIT University.
- 88. Wang, Y., Q. Han, and W. Liu (2013). Modelling and dynamic output feedback controller design for networked control systems. *IEEE Proceedings of American Control Conference*, 1615–1620.
- 89. Wang, Y., Z. Xia, Z. Jiang, and G. Xie (2011). A quasisliding mode variable structure control algorithm for discrete-time and time delay systems. *IEEE Proceedings of Control Decision Conference*, 107–110.
- Wen, D. and Y. Gao (2013). H-inifinity controller design for networked control systems in multiple packet transmission with random delays. *Applied Mechanics and Materials*, 278.

- 91. Wu, J. and T. Chen (2007). Design of networked control systems with packet dropouts. *IEEE Transactions on Automatic Control*, **52**(7), 1314–1319.
- Xiaojuan, Y. and L. Jinglin (2010). Bldcm fuzzy sliding mode control based on network control system. *International Conference on Advanced Computer Control*, 332– 335.
- 93. Xiong, J. and J. Lam (2009). Satbilization of networked control systems with a logic zoh. *IEEE Transactions on Automatic Control*, **54**(2), 358–363.
- Yang, F., Z. Wang, Y. Hung, and M. Gani (2006). H-infinity control for networked systems with random communication delays. *IEEE Transactions on Automatic Control*, 51(3), 511–518.
- 95. Yang, T. C. (2006). Network control system: A brief survey. *IEE Proceedings on Control Theory Applications*, **153**(4), 403–412.
- 96. Yang, W., L. Fan, and J. Luo (2010). Design of discrete time sliding mode observer in networked control system. *IEEE Proceedings of Chinese Control and Decision Con-ference*, 1884–1887.
- 97. Yao, D., H. Karimi, Y. Sun, and Q. Lu (2014). Robust model predictive control of networked control systems under input constraints and packet dropouts. *Abstract and Applied Analysis*, 2014(2014), 1–11.
- 98. Yi, H., H. Kim, and J. Choi (2014). Design of networked control system with discretetime state predictor over wsn. *Journal of Advance Computing Networks*, 2(2), 106–109.
- 99. Yin, Y., L. Xia, L. Song, and M. Qian (2011). Adaptive sliding mode control of networked control systems with variable time delay. *International Conference on Electric and Electronics*, 131–138.
- 100. Yu, H. and P. J. Antsaklis (2011). Event-triggered output feedback control for networked control systems using passivity: Time-varying network induced delays. *IEEE Conference on Decision and Control and European Control Conference*.
- 101. Yu, J., M. Yu, H. Shi, B. Liu, and Y. Gao (2014). Design of multiple dynamic output feedback controllers for networked control systems. *IEEE Proceedings of Chinese Control Conference*.
- 102. Yu, Z., H. Chen, and Y.Wang (2000). Control of network system with random communication delay and noise disturbance. *Control and Decision*, **15**(5), 518–526.
- 103. Yue, D., Q. Han, and J. Lam (2005). Network-based robust h-infinity control of systems with uncertainty. *Automatica*, **41**(1), 999–1007.
- 104. Yue, D., Q. Han, and C. Peng (2004). State feedback controller design of networked control systems. *IEEE Transactions on Circuits and Systems-II*, **51**(11), 640–644.
- 105. **Yuhong, H.** and **S. Yeguo** (2010). State feedback controller design of networked control system with time delay in the state derivatives. *International Conference on Computer Application and System Modeling*, 319–322.
- 106. Zhang, J., J. Lam, and Y. Xia (2013*a*). Output feedback sliding mode control under networked environment. *International Journal of Systems Science*, **44**(4), 750–759.

- 107. **Zhang, J.** and **Y. Xia** (2011). Networked predictive output feedback control of networked control systems. *IEEE Conference on Chinese Control Conference*.
- Zhang, L., H. Gao, and O. Kaynak (2013b). Network-induced constraints in networked control systems: A survey. *IEEE Transactions on Industrial Informatics*, 9(1), 403–416.
- Zhang, L., Y. Shi, T. Chen, and B. Huang (2005). A new method for stabilization of networked control systems with random delays. *IEEE Transactions on Automatic Control*, 50(8), 1171–1181.
- 110. **Zhang, W., M. Branicky**, and **S. Phillips** (2001). Stability of networked control systems. *IEEE Control System Magazine*, **1**(1), 84–99.
- 111. Zhang, X., Q. Han, and X. Yu (2016). Survey on recent advances in networked control systems. *IEEE Transactions on Industrial Informatics*, **12**(5), 1740–1752.
- 112. **Zhao, Y., G. Liu**, and **D. Rees** (2008). Integrated predictive control and scheduling co-design for networked control systems. *IET Control Theory Applications*, **2**(1), 7–15.
- Zhao, Y., X. Sun, J. Zhang, and P. Shi (2015). Networked control systems: The communication basics and control methodologies. *Mathematical Problems in Engineering*, 2015, 1–9.
- 114. **Zhivoglyadov, P.** and **R. Middleton** (2003). Networked control design for linear systems. *Automatica*, **39**(2003), 743–750.
- 115. **Zhu, X.** and **G. Yang** (2009). State feedback controller design of networked control systems with multiple-packet transmission. *International Journal of Control*, **82**(1), 86–94.

# PATENT, PUBLICATIONS AND ACKNOWLEDGEMENTS

#### • Indian Patent

1. A J Mehta, D H Shah, H A Mehta and R D Shah, Discrete-Time Sliding Mode Controller for Networked Control System with Random Fractional Delay, Application No. 201721015486, May, 2017.

#### • Book

1. D H Shah and A J Mehta, *Discrete-Time Sliding Mode Control for Networked Control System*, Springer, ISBN- 978-981-10-7535-3, http://www.springer.com/in /book/9789811075353, Feb. 2018.

## • Papers in Referred Journals

- 1. D H Shah and A J Mehta, *Fractional delay compensated discrete-time SMC* for networked control system, Digital Communication and Networks, Elsevier, Vol. 3, No. 2, May 2017, pp. 112-115.
- D H Shah and A J Mehta, Discrete-Time Sliding Mode Controller Subject to Real-Time Fractional Delays and Packet Losses for Networked Control System, International Journal of Control, Automation and Systems, Springer, Vol. 15, No. 6, Dec.-2017, pp. 2690-2703.
- 3. D H Shah and A J Mehta, *Discrete-Time Sliding Mode Control for Networked Control System with Random Fractional Delay and Packet Loss*, International Journal of Systems, Communication and Control, Inderscience, Under Review.
- 4. D H Shah and A J Mehta, *Discrete-Time Sliding Mode Control with Disturbance Estimator for Networked Control System*, International Journal of Systems, Communication and Control, Inderscience, Under Review.

### • Papers in International Conferences

- D H Shah and A J Mehta, *Robust Controller Design for Networked Control* System, IEEE International Conference on Computer, Communication and Control Technology (I4CT-2014), LangKawi, Malaysia. (ISBN- 978-1-4799-4555-9), Vol. 2, pp. 385-390, Sept. 2014.
- D H Shah and A J Mehta, *Output Feedback Discrete-Time Networked Sliding Mode Control*, IEEE International Workshop on Recent Advances in Sliding Modes (RASM-2015), Istanbul, Turkey. (ISBN-978-1-4799-8948-5), Vol. 1, pp. 143-150, April 2015.
- D H Shah and A J Mehta, Multirate Output Feedback Based Discrete-Time Sliding Mode Control for Fractional Delay Compensation in NCSs, IEEE International Conference on Industrial Technology, Toronto, Canada (ICIT- 2017), Torronto, Canada. (ISBN-978-1-5090-5319-3), pp. 848-853, March 2017.
- D H Shah and A J Mehta, Discrete-Time Sliding Mode Control Using Thiran's Delay Approximation for Networked Control System, Submitted to 43rd Annual Conference of the IEEE Industrial Electronics Society, (IECON- 2017), Beijing, China, (ISBN-978-1-5386-1126-5), pp. 3025-3031, Nov. 2017.

# • Acknowledgements

- 1. Received International Travel Grant from Science Education and Research Board (SERB), New Delhi of 1 Lac to present the research paper in IEEE International Conference on Computer, Communication and Control Technology (I4CT-2014), Langkawi, Malaysia, 2014.
- 2. Received Partial Grant from Gujarat Technological University in the form of registration fees to present the research paper in IEEE International Conference on Industrial Technology (ICIT-2017), Toronto, Canada, 2017.